INVERTIBLE PHASES FOR MIXED SPATIAL SYMMETRIES AND THE FERMIONIC CRYSTALLINE EQUIVALENCE PRINCIPLE

ARUN DEBRAY

ABSTRACT. Freed-Hopkins [FH19a] give a mathematical ansatz for classifying gapped invertible phases of matter with a spatial symmetry in terms of Borel-equivariant generalized homology. We propose a slight generalization of this ansatz to account for cases where the symmetry type mixes nontrivially with the spatial symmetry, such as crystalline phases with spin-1/2 fermions. From this ansatz, we prove as a theorem a "fermionic crystalline equivalence principle," as predicted in the physics literature. Using this and the Adams spectral sequence, we compute classifications of some classes of phases with a point group symmetry; in cases where these phases have been studied by other methods, our results agree with the literature.

CONTENTS

0.	Introduction	1
1.	Phases on a G -space: the general principle	6
2.	The fermionic crystalline equivalence principle	12
3.	Computations in examples: summary of results and some generalities	16
4.	Examples: rotations and reflections	23
5.	Examples: tetrahedral, octahedral, and icosahedral symmetries	48
6.	Glide symmetry protected phases	78
7.	Conclusion and outlook	79
References		

0. INTRODUCTION

The classification of topological phases of matter has been the subject of intensive research in condensedmatter physics and nearby areas of mathematics for the last decade, but difficult problems still remain: for example, there is not yet an accepted mathematical definition of a topological phase of matter, so researchers must study these systems using ansatzes or heuristic definitions of phases. Restricting to invertible phases, also known as symmetry-protected topological (SPT) phases, simplifies the classification question, but defining these phases precisely is also still an open problem. Freed-Hopkins [FH16a] make an ansatz modeling SPT phases using reflection-positive invertible field theories (IFTs), then classify these IFTs using homotopy theory. This approach has been successfully employed in several cases to study examples of SPTs, as in [FH16a, Cam17, WWW18, FHHT20, GOP⁺20, PW20].

Condensed-matter physicists are also interested in invertible phases in more general settings, including invertible phases on a particular space Y, as in [Ran10], or invertible phases symmetric for a group G acting on space, such as phases on the plane which have a rotation symmetry and the examples in [SMJZ13]. These spatial symmetries are often present in real-world examples of topological phases of matter (see [WACB16, $MYL^{+}17$ for one example), and can be modeled by lattice Hamiltonian systems in which the symmetry group also acts on the lattice, though again providing precise definitions is still open. In the case where G is a crystallographic group acting on $Y = \mathbb{R}^d$, these systems are called *crystalline SPT phases*. Freed-Hopkins' field-theoretic approach does not directly generalize to this setting, but there is a general ansatz of Kitaev [Kit13, Kit15] that groups of phases on Y for a fixed symmetry type should define a generalized homology theory. Freed-Hopkins [FH19a] apply this to propose a classification of invertible phases in the presence of a G-action on space using equivariant generalized homology.

Date: March 10, 2021.

Researchers interested in computing groups of crystalline SPT phases provide crystalline equivalence principles, including the first such proposal of Thorngren-Else [TE18] and subsequent work in [JR17, CW18, FH19a, ZWY⁺20, ZYQG20]. Crystalline equivalence principles are arguments that groups of crystalline SPT phases are isomorphic to groups of ordinary SPT phases, where the symmetry type is modified. The theory is well-understood for symmetry types such as O_n and SO_n , corresponding to the physicists' notion of bosonic SPT, but for fermionic SPTs, corresponding to symmetry types such as $Spin_n$, $Spin_n^c$, Pin_n^{\pm} , etc., the story is more complicated. Cheng-Wang [CW18], Zhang-Wang-Yang-Qi-Gu [ZWY⁺20], and Zhang-Yang-Qi-Gu [ZYQG20] study examples of fermionic crystalline SPTs, and show cases of a fermionic crystalline equivalence principle. Crucially, their work implies any fermionic crystalline equivalence principle must address fermionic phases in which the spatial symmetry mixes with fermion parity, which goes beyond the scope of Freed-Hopkins' ansatz.

The purpose of this paper is to formulate and prove such a fermionic crystalline equivalence principle (FCEP). To do so, we provide an ansatz expressing groups of invertible phases on a *G*-space *Y* in which the symmetry type can be merely locally constant over space and can mix with *G*, including as a special case spatial symmetries mixing with fermion parity. Given data \mathcal{L} expressing this mixing and variance of the symmetry type, we define *phase homology* groups of *Y*, denoted $Ph_*^G(Y, \mathcal{L})$, and our ansatz predicts that the group of such invertible phases is isomorphic to $Ph_0^G(Y, \mathcal{L})$. Providing this ansatz is an additional goal of this paper, and is necessary input for our FCEP: the ansatz reexpresses the FCEP as an isomorphism between certain phase homology groups and groups of IFTs, as we state and prove in Theorem 2.8. This is the first homotopy-theoretic account of an FCEP, and to the best of our knowledge is the first fully general version of the FCEP in the literature.

As a corollary of the FCEP, the computation of phase homology groups that represent groups of pointgroup-equivariant fermionic phases reduces to computations of bordism groups; this paper's third goal is to make these computations in several examples, both for the purpose of testing our ansatz by comparing it to established predictions in physics, and for making additional predictions of groups of crystalline SPT phases in as yet unstudied settings. For symmetry types that have been studied before by other methods, our computations agree with the literature, bolstering our ansatz.

Now we go into a little more detail about these ansatzes and theorems. Freed-Hopkins [FH19a] formulate an ansatz for invertible phases of matter on a topological space Y equipped with an action of a compact Lie group G. First, specify the symmetry type of the theory as a map $\rho: H \to O$, where $O := \varinjlim_n O_n$ is the infinite orthogonal group and H is a topological group. From this data we can form a Madsen-Tillmann spectrum MTH, whose homotopy groups compute the bordism groups of manifolds with an H-structure on the tangent bundle. Let $I_{\mathbb{Z}}$ denote the Anderson dual of the sphere spectrum and $E := \operatorname{Map}(MTH, \Sigma^2 I_{\mathbb{Z}})$.

Ansatz 0.1 (Freed-Hopkins [FH19a, Ansatz 3.3]). The abelian group of isomorphism classes of phases on Y equivariant for a G-symmetry that does not mix with the symmetry type H is the Borel-equivariant Borel-Moore homology group $E_{0,BM}^{hG}(Y)$.

We will define equivariant Borel-Moore homology in the generality we need in Definition 1.17.

When G is trivial and $Y = \mathbb{R}^n$, the group of phases in Ansatz 0.1 is naturally isomorphic to $[MTH, \Sigma^{d+2}I_{\mathbb{Z}}]$, which Freed-Hopkins [FH16a] show is the classification of invertible field theories with symmetry type H.¹ When $Y = \mathbb{R}^d$ and G is a crystallographic group, this group of phases is expected to model the classification of crystalline SPT phases with this symmetry type, and indeed, Freed-Hopkins [FH19a, Example 3.5] prove a version of the bosonic crystalline equivalence principle of Thorngren-Else [TE18] as a consequence of their ansatz, matching physicists' predictions.

For fermionic phases, Ansatz 0.1 is not the full answer, and providing the full answer is a major goal of this paper. Physicists distinguish between phases with "spinless fermions" and "spin-1/2 fermions", asking how the spatial symmetry group G mixes with fermion parity. For example, one could consider phases on the plane equivariant for a C_4 rotation symmetry, and either ask that fermions' spin is unaffected by the spatial rotations, or that a full spatial rotation flips the spin on the fermion. This is reminiscent of the better-understood dichotomy of fermionic phases with a time-reversal symmetry T: one may have $T^2 = 1$ or

¹This result is conditioned on a conjecture about non-topological invertible theories; at present, we have as a theorem only that the invertible TFTs are classified by the torsion subgroup of this group. This is discussed by Freed-Hopkins [FH16a, §5.4] and Freed [Fre19, Lecture 9].

 T^2 equal to the fermion parity operator. These two classes of phases are modeled with different symmetry types, and similarly we use different data to model crystalline phases with spinless vs. spin-1/2 fermions.

To accommodate this mixing between the internal symmetry type H and the spatial symmetry group G, we generalize Freed-Hopkins' setup slightly using parametrized homotopy theory, considering local systems fof symmetry types over the base Y. These give rise to local systems of Thom spectra; if Y has a G-action we obtain from f a local system \mathcal{L}' of Borel-equivariant Thom spectra, modeled as a functor from Y, thought of as an ∞ -groupoid, to the ∞ -category Sp^G of Borel-equivariant spectra. Let $\mathcal{L} := \operatorname{Map}(-, \Sigma^2 I_{\mathbb{Z}}) \circ \mathcal{L}'$ as maps $Y \to Sp^G$, where $I_{\mathbb{Z}}$ has trivial G-action. We define the *equivariant phase homology* $Ph^G_*(Y; f)$ to be the equivariant Borel-Moore homology of the local system $\mathcal{L}: Y \to Sp^G$.

Ansatz 1.22. The group of *G*-equivariant invertible phases on *Y* for this data is isomorphic to the equivariant phase homology group $Ph_0^G(Y; f)$.

When f is trivializable, this reduces to Ansatz 0.1; in general, though, it allows the symmetry type to mix with the spatial symmetry, or to be merely locally constant on Y.

Now we specialize to the cases of spinless and spin-1/2 fermions. For spinless fermions, G and H do not mix, so we use the data of a constant local system of symmetry types and recover Freed-Hopkins' original ansatz. For spin-1/2 fermions, we specify data of an extension of G by H

$$(0.2) 0 \longrightarrow H \longrightarrow \widetilde{H} \longrightarrow G \longrightarrow 0,$$

together with a representation $\lambda: G \to O_d$ dictating how G acts on space.² In the cases we consider in this paper, $H = \text{Spin} \text{ or } H = \text{Spin}^c$, and we specify \widetilde{H} by way of the central extension

$$(0.3) 0 \longrightarrow \mu_2 \longrightarrow \widetilde{G} \longrightarrow G \longrightarrow 0$$

whose isomorphism class is picked out by $w_2(V_{\lambda}) + w_1(V_{\lambda})^2 \in H^2(BG; \mu_2)$, where $V_{\lambda} \to BG$ is the associated vector bundle to the representation λ and μ_2 is the group of square roots of unity. Then, $\tilde{H} := H \times_{\mu_2} \tilde{G}$. Using this data, we build an equivariant local system f of symmetry types, obtaining a phase homology group $Ph_0^G(\mathbb{R}^d, f)$ that we predict is isomorphic to the group of invertible phases for this data.

The FCEP, previously studied in special cases by [CW18, TE18, ZWY⁺20, ZYQG20], identifies groups of crystalline SPT phases with groups of fermionic SPT phases with an internal *G*-symmetry — but exchanging symmetry types: spinless crystalline phases correspond to spin-1/2 internal phases, and vice versa. Freed-Hopkins [FH16a] model groups of SPT phases with an internal *G*-symmetry using IFTs, and following Freed-Hopkins [FH16a] and the excellent overview by Beaudry-Campbell [BC18], these groups of TFTs can be expressed in terms of bordism groups of certain Thom spectra. Standard techniques in algebraic topology, notably the Adams spectral sequence over $\mathcal{A}(1)$, can be used to compute these bordism groups, so one application of a general version of the FCEP is to provide access to tractable tools for computing groups of crystalline SPT phases.

One of the major aims of this paper is to state and prove as a theorem a version of the FCEP, identifying phase homology groups with groups of IFTs; then Ansatz 1.22 translates this into a statement about crystalline SPTs and ordinary SPTs. In Definitions 2.3 and 2.4, we define the symmetry types for spinless and spin-1/2 fermions for a purely internal *G*-symmetry. In general these definitions are a little technical, but when the spatial representation λ factors through SO_d \subset O_d, the spinless internal symmetry type is $H \times G \to O$ and the spin-1/2 symmetry type is $H \times_{\mu_2} \tilde{G} \to O$, with the maps induced by the projection onto the first factor.

Theorem 2.8 (Fermionic crystalline equivalence principle). Fixing data of G, H, λ , etc. as above, let $f_0, f_{1/2}$ denote the local systems of symmetry types for the case of spinless, resp. spin-1/2 fermions. Then $Ph_0^G(\mathbb{R}^d; f_0)$ is isomorphic to the group of deformation classes of d-dimensional IFTs for the spin-1/2 internal symmetry type, and $Ph_0^G(\mathbb{R}^d, f_{1/2})$ is isomorphic to the group of deformation classes of d-dimensional IFTs for the spinless internal symmetry type.

The proof has two key steps.

(1) Phase homology groups are defined using equivariant parametrized homotopy theory. Proposition 1.32 reexpresses them using ordinary homotopy theory, as homotopy groups of a Thom spectrum

 $^{^{2}}$ We also specify some additional data; see Data 2.1 in §2 for the full details.

built from a virtual vector bundle over $B\tilde{H}$. The proof uses the Ando-Blumberg-Gepner-Hopkins-Rezk [ABG⁺14a, ABG⁺14b] approach to Thom spectra.

(2) Then, in Theorems 2.11 and 2.24, we "shear" this Thom spectrum, writing down a map $H_n \to H_{n+d} \times G$ and showing that it induces a homotopy equivalence on Thom spectra, implying that phase homology groups are determined by *H*-bordism groups of a Thom spectrum over *BG*. Our proof is modeled on a fairly general shearing theorem in Freed-Hopkins [FH16a, §10].

After these two steps, the proof of Theorem 2.8 amounts to looking at the Thom spectra for the internal symmetry types and noticing that we end up with equivalent Thom spectra over BG in the cases we want to equate.

With this tool in hand, we can compute phase homology groups for point groups acting on \mathbb{R}^d , which are our model for groups of fermionic phases equivariant for point group symmetries. We do these computations for many 2d and 3d point groups, for both spinless and spin-1/2 fermions, and in Altland-Zirnbauer classes D and A (corresponding to H = Spin, resp. Spin^c). Our computations use two avatars of the Adams spectral sequence. It is well-known that low-dimensional spin bordism can be computed using connective *ko*-homology and the Adams spectral sequence over $\mathcal{A}(1)$, and there is an excellent introduction to this technique by Beaudry-Campbell [BC18], but we also use a variant, computing spin^c bordism via *ku*-homology and the Adams spectral sequence over $\mathcal{E}(1)$, e.g. in §4.4.3. This is hardly a new idea, but there appear to be no examples of this specific kind of computation in the literature before now. We hope that our computations serve as useful examples of how to use this version of the Adams spectral sequence for spin^c bordism; this could be of independent interest.

For 2d point groups, these phases have been studied in the physics literature using very different methods. We compare our results with those of other researchers in §4.1.4, §4.2.4, §4.3.4, and §4.4.5, and find agreement, providing evidence in favor of Freed-Hopkins' ansatz and our generalization. However, there is not yet work on fermionic crystalline SPT phases for most 3d point groups, so our computations are predictions. We do many computations and make many predictions, and in §3.1 we collect a few that we think are relatively interesting or accessible. For example:

Theorem. Let A_4 act on \mathbb{R}^3 as the orientation-preserving symmetries of a tetrahedron. Then $Ph_0^{A_4}(\mathbb{R}^3; f)$ vanishes, where f is the local system of symmetry types for either spinless or spin-1/2 fermions in both Altland-Zirnbauer classes D and A.

This is a combination of Theorems 5.4, 5.6, and 5.8. Therefore, assuming Ansatz 1.22, there are no nontrivial spinless nor spin-1/2 fermionic SPT phases equivariant for a chiral tetrahedral symmetry in Altland-Zirnbauer classes D or A. It would be interesting to see this prediction studied using lattice methods for fermionic crystalline phases.

In §6, we leave behind the FCEP and consider a different class of examples, SPTs equivariant for a glide reflection symmetry, providing a test for Freed-Hopkins' ansatz for a crystallographic group that is not a point group. Lu-Shi-Lu [LSL17] conjecture a general classification of these SPTs: that if $TP_d(H)$ denotes the group of d-dimensional SPT phases with symmetry type H, then the group of d-dimensional glide SPTs is isomorphic to $TP_{d-1}(H) \otimes \mathbb{Z}/2$. Xiong-Alexandradinata [XA18] derive this classification using physics-based arguments. We use Freed-Hopkins' ansatzes [FH16a, FH19a] to translate Lu-Shi-Lu's conjecture into a statement about phase homology groups and prove it.

Recall $E := \operatorname{Map}(MTH, \Sigma^2 I_{\mathbb{Z}})$ and let $\widehat{Ph}^{\mathbb{Z}}_*(\mathbb{R}^d, \underline{E})$ denote the kernel of the forgetful map from \mathbb{Z} -equivariant phase homology to nonequivariant phase homology, where \mathbb{Z} acts on \mathbb{R}^d by glide translations, and $\underline{E} \to \mathbb{R}^d$ is the constant local system. This kernel models Lu-Shi-Lu's group of glide SPTs, as they require glide SPTs to be trivial in the absence of the glide symmetry.

Theorem 6.4. There is a natural isomorphism $\widehat{Ph}_0^{\mathbb{Z}}(\mathbb{R}^d;\underline{E}) \cong E_{-(d-1)} \otimes \mathbb{Z}/2.$

This provides additional evidence in favor of the ansatz.

We want to mention that there are other homotopy-theoretic approaches to the study of phases of matter with a spatial symmetry, including those of Antolín Camarena, Sheinbaum, and collaborators [AACSS16, SC20] and Cornfeld-Carmeli [CC21]. These authors deal with free fermion phases, which are out of scope of this paper, though see §7.1.

0.1. Reader's guide to the different sections. Overview:

- In \$ 2 we discuss general aspects of our model for phases on a G-space Y and prove the FCEP. These sections involve the most homotopy theory.
- In §§3–5 we make phase homology calculations which according to Ansatz 1.22 calculate groups of fermionic crystalline SPT phases for which the symmetry group is a point group. We collect the results of these computations in Tables 1, 2, 3, 4, 5, and 6, and summarize the methods of computation in §3.2.
- In §6 we consider phases on \mathbb{R}^d with a glide symmetry, and prove a theorem computing the corresponding phase homology classification.

Now a little more detail. In §1, we use Borel-equivariant parametrized homotopy theory to state a mild generalization of Freed-Hopkins' ansatz on invertible phases with spatial symmetry. In §1.1, we consider phases on a space Y without a group action, using local systems of symmetry types (Definition 1.3). We define phase homology and in Ansatz 1.10 express the group of invertible phases for such a local system in terms of phase homology. This is a slight generalization of [FH19a, Ansatz 2.1]. In §1.2, we allow group actions, defining equivariant local systems of symmetry types and equivariant Borel-Moore homology for a local system for the purpose of formulating Ansatz 1.22 expressing groups of invertible phases for a spatial symmetry in terms of equivariant phase homology. This is a minor generalization of Freed-Hopkins' ansatz [FH19a, Ansatz 3.3] to the parametrized setting. Then, in §1.3, we specialize to the case relevant to the FCEP, defining the local systems of symmetry types for spatial symmetries that mix with fermion parity. We prove Proposition 1.32 expressing the phase homology groups for this data in terms of nonequivariant, nonparametrized homotopy theory, and do not need equivariant or parametrized homotopy theory in the rest of the paper.

Next, §2, whose goal is to state and prove the FCEP. We begin in Definitions 2.2, 2.3, and 2.4 by defining the spinless and spin-1/2 local systems of symmetry types for both equivariant (i.e. G acting on space) and internal (G not acting on space) symmetries, and use these definitions to state our FCEP theorem in Theorem 2.8, identifying phase homology groups for these local systems in terms of groups of IFTs. As mentioned, the nontrivial part of the proof runs a shearing argument to simplify a Thom spectrum over $B\tilde{H}$ into a smash product of MTSpin and a Thom spectrum over BG. In §2.1, we prove Theorem 2.11 accomplishing this in class D, for which H = Spin. Then, in §2.2, we prove Theorem 2.24, which is the analogous theorem in class A, i.e. for $H = \text{Spin}^c$, via a similar proof. Finally, in §2.3, we combine these arguments to prove Theorem 2.8.

In §3, we address a few generalities related to the FCEP before studying it in examples. First, in §3.1, we provide a summary of some phases or phenomena newly predicted by our computations which might be interesting to investigate further. In §3.2, we introduce and review the tools from algebraic topology we need to make these computations: the Adams and Atiyah-Hirzebruch spectral sequences. In §3.3, we discuss how to use the Adams filtration to detect when an invertible TFT of \tilde{H} -manifolds only depends on the underlying SO × G-structure, which is believed to correspond to detecting which fermionic phases are really bosonic phases that are fermionic in a trivial way. Finally, in §3.4, we state and prove several lemmas needed in the computations in the next sections.

Then, in §§4–5, we implement this in examples, computing phase homology groups of \mathbb{R}^d equivariant for two- and three-dimensional point-group symmetries, which in Ansatz 1.22 are interpreted as groups of point group equivariant fermionic phases on \mathbb{R}^d . In all cases we consider Altland-Zirnbauer classes D and A (corresponding to symmetry types spin and spin^c, respectively), and consider phases with spinless fermions and spin-1/2 fermions. These computations amount to computing spin and spin^c bordism groups of Thom spectra of vector bundles over BG, where G is the point group of interest; we use the Adams and Atiyah-Hirzebruch spectral sequences to determine these bordism groups.

In §4, we consider $\mathbb{Z}/2$ acting by a reflection (§4.1) and by an inversion (§4.2), as well as C_n acting by rotations (§4.3) and D_{2n} acting by rotations and reflections on \mathbb{R}^2 (§4.4) or purely by rotations on \mathbb{R}^3 (§4.5). The results of these computations can be found in Tables 1, 2, 3, 4, and 5. Most of these symmetry types have been studied in the physics literature, and we compare our results with other researchers'.

In §5, we study many 3d point groups, including chiral tetrahedral symmetry (§5.1), pyritohedral symmetry (§5.2), full tetrahedral symmetry (§5.3), chiral octahedral symmetry (§5.4), full octahedral symmetry (§5.5), chiral icosahedral symmetry (§5.6), and full icosahedral symmetry (§5.7). In all cases, we study phases with spinless and spin-1/2 fermions in Altland-Zirnbauer types D and A. Our predictions in this section are new as far as we can determine. See Table 6 for the results of the computations.

In §6, we discuss phases equivariant for a glide reflection symmetry. Lu-Shi-Lu [LSL17] conjecture a general classification of such phases, and we translate their conjecture into a statement on phase homology groups using Freed-Hopkins' ansatz, then prove that statement. Finally, in §7, we suggest some directions for further research.

Acknowledgments. I gratefully thank my advisor, Dan Freed, for his constant help, guidance, and patience. In addition, this paper benefited from conversations with Dexter Chua, Miguel Montero, Riccardo Pedrotti, Oscar Randal-Williams, Daniel Sheinbaum, Luuk Stehouwer, Ryan Thorngren, Juven Wang, and Qing-Rui Wang. Thank you to all.

A portion of this work was supported by the National Science Foundation under Grant No. 1440140 while the author was in residence at the Mathematical Sciences Research Institute in Berkeley, California, during January–March 2020.

1. Phases on a G-space: the general principle

We reprise the ansatz of Freed-Hopkins [FH19a, Ansatzes 2.1, 3.3] on invertible phases on a G-space, though we need to generalize it: physicists often consider crystalline phases in which the symmetry acting on spacetime mixes with the internal symmetry (e.g. a reflection squaring to $(-1)^F$), leading us to generalize from homology to twisted homology.

What we do not do is define a phase of matter. Precisely defining topological phases of matter, even in the absence of spatial symmetries, is a difficult open question. Our ansatz is a heuristic that these objects can be classified with what we call *phase homology*, which we do define.

1.1. Invertible phases on a space. Let Y be a locally compact topological space and C an ∞ -category.³ Following Ando-Blumberg-Gepner [ABG10, ABG18], we say a C-valued local system on Y is a functor $\mathcal{L}: \pi_{\leq \infty} Y \to \mathbb{C}$ here $\pi_{\leq \infty} Y$ is the fundamental ∞ -groupoid of Y.⁴ If $\mathcal{L}: Y \to Sp$ is a local system of spectra, the homology of Y valued in \mathcal{L} is $\mathcal{L}_*(Y) \coloneqq \pi_*(\operatorname{hocolim} \mathcal{L})$, and the cohomology of Y valued in \mathcal{L} is $\mathcal{L}^*(Y) \coloneqq \pi_*(\operatorname{holim} \mathcal{L})$; this generalizes (co)homology with local coefficients.

Given a subspace $j: Y' \hookrightarrow Y$, we also define relative homology groups: j induces a map j_* : hocolim_{Y'} $\mathcal{L}|_{Y'} \to$ hocolim_Y \mathcal{L} , and we define $\mathcal{L}(Y, Y') \coloneqq \pi_*(\operatorname{cofib}(j_*))$. Relative cohomology is analogous.

Definition 1.1. Assume that the one-point compactification \overline{Y} of Y is a finite CW complex and \mathcal{L} extends to a local system $\overline{\mathcal{L}}: \overline{Y} \to \mathcal{S}p$. Choose such an extension $\overline{\mathcal{L}}$ over the basepoint *. The *Borel-Moore homology* of Y valued in \mathcal{L} is

(1.2)
$$\mathcal{L}_{\mathrm{BM},*}(Y) := \overline{\mathcal{L}}_*(\overline{Y},*).$$

Definition 1.1 appears to depend on the choice of extension of \mathcal{L} to \overline{Y} , but given two choices of extension, the cofibers of the induced maps hocolim $\overline{\mathcal{L}}|_* \to \operatorname{hocolim} \overline{\mathcal{L}}$ are equivalent, hence compute the same Borel-Moore homology groups.

When \mathcal{L} is constant, this recovers the usual notion of Borel-Moore (generalized) homology [BM60, Mil95]. Recall that a symmetry type is a space B with a map $f: B \to BO$.

Definition 1.3. A local system of symmetry types over the space Y is a local system on Y valued in the ∞ -category of spaces with a map to BO.

This is closely related to Raptis-Steimle's definition of parametrized tangential structures [RS17, §2].

Symmetry types often arise as the stabilizations in n of maps $B\rho_n: BH_n \to BO_n$ induced from representations $\rho_n: H_n \to O_n$; see [FH16a, §2] for a general discussion. Likewise, the local systems of symmetry types we consider arise from BH-bundles over Y.

We repeatedly use the notion of *Thom spectra*; the definition given by Freed-Hopkins [FH16a, §6.1.4] covers the cases we need.⁵

³There are different definitions of ∞ -categories; we work with *quasicategories* as developed by Joyal [Joy02] and Lurie [Lur09], so as to follow [ABG10, ABG18]. However, this paper does not depend on implementation-specific details. See [ABG18, §2] for more information and some useful references.

⁴This is not the only approach to parametrized homotopy theory; see also May-Sigurdsson [MS06] and Braunack-Mayer [BM19].

⁵Thom spectra have been heavily studied in homotopy theory; key references include Thom [Tho54], Atiyah [Ati61], May-Quinn-Ray-Tornehave [May77], and Ando-Blumberg-Gepner-Hopkins-Rezk [ABG⁺14a, ABG⁺14b].

Definition 1.4. Given a representation $\rho_n \colon H_n \to O_n$ or $\rho \colon H \to O$, where $O \coloneqq \varinjlim_n O_n$, we introduce notation for several Thom spectra. Let $V_n \to BO_n$ and $V \to BO$ denote the tautological vector bundle, resp. the tautological stable vector bundle. By convention, $V \to BO$ has rank zero.

- (1) The Thom spectra MH_n , resp. MH, are the Thom spectra of $(B\rho_n)^*V_n \to BH_n$, resp. $(B\rho)^*V \to BH$.
- (2) The Madsen-Tillmann spectra [MT01, MW07] MTH_n , resp. MTH, are the Thom spectra of $(B\rho_n)^*(-V) \rightarrow BH_n$, resp. $(B\rho)^*(-V) \rightarrow BH$.

We will use $H \in \{O, SO, Spin, Spin^c, Pin^{\pm}, Pin^c\}$; in all of these cases, ρ is the usual map $H \to O$ used in, e.g., [FH16a].

Remark 1.5. Some Thom spectra go by many names. The notation \mathbb{RP}_n^{∞} denotes $(BO_1)^{nV_1}$, and similarly $\mathbb{CP}_n^{\infty} \coloneqq (BSO_2)^{nV_2}$. Thus, for example, $\Sigma^2 MTSO_2$, $\Sigma^2 MTU_1$, and $\Sigma^2 \mathbb{CP}_{-1}^{\infty}$ all refer to $(BSO_2)^{2-V_2}$.

Definition 1.6. The Anderson dual of the sphere spectrum [And69, Yos75] is a spectrum $I_{\mathbb{Z}}$ satisfying the universal property that for any spectrum X, there is a natural short exact sequence

(1.7) $0 \longrightarrow \operatorname{Ext}(\pi_{n-1}(X), \mathbb{Z}) \longrightarrow [X, \Sigma^n I_{\mathbb{Z}}] \longrightarrow \operatorname{Hom}(\pi_n(X), \mathbb{Z}) \longrightarrow 0.$

As all such spectra are equivalent, we refer to "the" Anderson dual of the sphere spectrum to mean any particular choice of $I_{\mathbb{Z}}$.

(1.7) splits, but not naturally, implying a non-natural isomorphism from $[X, \Sigma^n I_{\mathbb{Z}}]$ to the direct sum of the torsion summand of $\pi_{n-1}(X)$ and the free summand of $\pi_n(X)$. We often use this fact implicitly, calculating $\pi_*(X)$ but depending on the reader to rearrange it into $[X, \Sigma^* I_{\mathbb{Z}}]$. For more on $I_{\mathbb{Z}}$ and its appearance in this context, see Freed-Hopkins [FH16a, §5.3, §5.4].

Let Th: $\operatorname{Top}_{BO} \to \operatorname{Sp}$ denote the Thom spectrum functor and $\mathcal{I}: \operatorname{Sp}^{\operatorname{op}} \to \operatorname{Sp}$ denote the functor $\operatorname{Map}(-, \Sigma^2 I_{\mathbb{Z}})$.

Definition 1.8. Let Y be a locally compact space and $f: Y \to \Im op_{BO}$ be a parametrized symmetry type on Y. The *phase homology* of this data, denoted $Ph_*(Y; f)$, is the Borel-Moore homology

(1.9)
$$Ph_*(Y;f) \coloneqq (\mathcal{I} \circ \mathrm{Th} \circ f)_{\mathrm{BM},*}(Y).$$

Ansatz 1.10. With Y and f as in Definition 1.8, the group of invertible topological phases on Y for the local system of symmetry types f is the phase homology group $Ph_0(Y; f)$.

Again, this is not a mathematical definition, but rather a heuristic.

Remark 1.11. When f is constant, Ansatz 1.10 is the original ansatz of Freed-Hopkins [FH19a, Ansatz 2.1]. In that case, the ansatz builds on the idea that invertible phases on Y are related to families of reflection-positive invertible field theories on Y. The generalization to nonconstant f allows one to prescribe how the symmetry type of the family varies along Y. For example, one might want to consider families of phases in which the monodromy around a loop in Y acts by orientation reversal.

1.2. Invertible phases on a G-space. Our model for invertible crystalline phases requires considering the case where a compact Lie group G acts on Y. Again we closely follow Freed-Hopkins [FH19a, §3] but using twisted Borel-Moore homology.

Throughout this section, G is a Lie group; unlike in [FH16a, FH19a], we do not need G to be compact. Indeed, in the study of crystalline phases, G is often an infinite discrete subgroup of $\text{Isom}(\mathbb{E}^n)$, and we will consider one such example in §6. We work with the ∞ -category Sp^G of Borel G-equivariant spectra, whose objects can be modeled by data of a sequence of G-spaces X_n together with G-equivariant maps $\Sigma X_n \to X_{n+1}$.⁶ Notions of homotopy equivalence, etc., are as in [FH16a, 6.1], and do not require their compactness assumption on G.

⁶There are a few different notions of G-spectra in the equivariant homotopy theory literature, and their names can be confusing. Borel G-equivariant spectra can be thought of as "spectra with a G-action" or "spectra living over BG," and are different from *genuine G-spectra*, which have a richer structure. To a geometer, "equivariant (generalized) cohomology" usually means the Borel theory, but to a homotopy theorist, it means the genuine theory. See [Sul20, §2.1] for a detailed introduction into the different names and notions of G-spaces and G-spectra.

Definition 1.12. Suppose G admits a finite-dimensional, real orthogonal representation $\lambda: G \to O_d$. The one-point compactification of \mathbb{R}^d with this G-action is a G-space denoted S^{λ} and called a representation sphere.

The suspension functor $\Sigma^{\lambda} \coloneqq S^{\lambda} \wedge -i$ s not invertible in *G*-spaces, but upon stabilization is invertible in Borel *G*-spectra; we denote its inverse by $\Sigma^{-\lambda}$. Given a virtual *G*-representation $V = \lambda - \mu$ (i.e. a formal difference of two finite-dimensional real orthogonal representations), we define the Borel *G*-spectrum $\mathbb{S}^{V} \coloneqq \Sigma^{-\mu} \Sigma^{\infty} S^{\lambda}$. We will let \mathbb{S} denote the sphere spectrum with trivial *G*-action.

Definition 1.13. Let Y be a G-space and $\mathcal{L}: Y \to Sp^G$ be a local system. The *(Borel-)equivariant homology* of Y with respect to \mathcal{L} is denoted $\mathcal{L}^G_*(Y)$ and defined to be

(1.14)
$$\mathcal{L}^G_*(Y) \coloneqq \pi_*(\operatorname{Map}_{\mathcal{S}p^G}(\mathbb{S}, \operatorname{hocolim} \mathcal{L})^{hG}),$$

where $(-)^{hG} \colon \mathbb{S}p^G \to \mathbb{S}p$ denotes the homotopy fixed-points functor.

If $j: Y' \hookrightarrow Y$ is an inclusion of G-spaces, it induces a map

(1.15)
$$j_* \colon \operatorname{Map}_{\mathbb{S}p^G}(\mathbb{S}, \operatorname{hocolim} \mathcal{L}|_{Y'})^{hG} \longrightarrow \operatorname{Map}_{\mathbb{S}p^G}(\mathbb{S}, \operatorname{hocolim} \mathcal{L})^{hG},$$

and we define the relative (Borel-)equivariant homology

(1.16)
$$\mathcal{L}^G_*(Y, Y') \coloneqq \pi_*(\operatorname{cofib}(j_*))$$

as in the nonequivariant case.

Definition 1.17. Let Y be a G-space and $\mathcal{L}: Y \to Sp^G$ be an Sp^G -valued local system. Assume that the one-point compactification \overline{Y} of Y is a CW complex and \mathcal{L} extends to a local system $\overline{\mathcal{L}}: \overline{Y} \to Sp^G$. Choose such an extension $\overline{\mathcal{L}}$. The *equivariant Borel-Moore homology* of Y valued in \mathcal{L} is

(1.18)
$$\mathcal{L}^G_{\mathrm{BM},*}(Y) \coloneqq \overline{\mathcal{L}}^G_*(\overline{Y},*).$$

Just like Definition 1.1, this does not actually depend on the choice of extension.

Definition 1.19. Let Y be a G-space. A G-equivariant local system of symmetry types is a G-space B and a G-equivariant map $f: B \to Y \times BO$, where BO has a trivial G-action.

Taking the Thom spectrum of the map to BO defines a local system $\text{Th} \circ f \colon Y \to \p^G .

Definition 1.20. Let Y be a G-space whose one-point compactification is a finite CW complex, and let $f: B \to Y \times BO$ be a G-equivariant local system of symmetry types for Y. The G-equivariant phase homology of this data, denoted $Ph_*^G(Y; f)$, is the equivariant Borel-Moore homology

(1.21)
$$Ph_*^G(Y;f) \coloneqq (\mathcal{I} \circ \operatorname{Th} \circ f)^G_{BM,0}(Y).$$

Ansatz 1.22. With Y and f as in Definition 1.20, the group of invertible topological phases on Y for the equivariant local system of symmetry types f is the G-equivariant phase homology group $Ph_0^G(Y; f)$.

Again, this is a heuristic and not a definition. When G is a discrete subgroup of $\text{Isom}(\mathbb{E}^n)$ (e.g. a wallpaper or space group) acting on $Y = \mathbb{E}^n$, these phases are called *crystalline SPT phases* in the physics literature.

1.3. Mixing internal and crystalline symmetries. The fermionic crystalline equivalence principle is about invertible topological phases in which an internal symmetry mixes with the symmetry group acting on space. In this section, we construct the equivariant local systems of symmetry types for these phases. First, we review how mixing of symmetries is handled in the purely internal case in Example 1.23; then we address the case of spatial symmetries in Proposition 1.32, showing how to reduce the computation of the relevant equivariant phase homology groups to a nonparametrized question. We will simplify these computations further in \S^2 when we discuss the FCEP in more detail, then study several examples in $\$\S^{4-5}$.

Example 1.23 (Mixing for internal symmetries). In the study of SPTs, one commonly encounters symmetry types where there are two different symmetries present, such as time reversal and fermion parity, but they mix, meaning the group they generate is not a product of the individual symmetry groups, but rather an extension. For example, we could ask for a generator T of the group of time-reversal symmetries to square to the fermion parity $(-1)^F$, via the extension $0 \to \mathbb{Z}/2 \to \mathbb{Z}/4 \to \mathbb{Z}/2 \to 0$, rather than considering phases where $T^2 = 1$, corresponding to the split extension $0 \to \mathbb{Z}/2 \to \mathbb{Z}/2 \to \mathbb{Z}/2 \to \mathbb{Z}/2 \to 0$.

9

Freed-Hopkins [FH16a] make the ansatz that SPT phases are classified up to equivalence by their low-energy limits, which are invertible field theories. The symmetry type is expressed as an H_n -structure, where H_n is a group with a map to O_n ; mixing manifests as an extension involving the base symmetry type (e.g. Spin_n for fermionic phases) and the additional symmetry. For example, the two cases of time-reversal symmetry squaring to the identity or to fermion parity are represented by the extensions

(1.24a)
$$1 \longrightarrow \operatorname{Spin}_n \longrightarrow \operatorname{Pin}_n^+ \longrightarrow \mathbb{Z}/2 \longrightarrow 1$$

(1.24b)
$$1 \longrightarrow \operatorname{Spin}_n \longrightarrow \operatorname{Pin}_n^- \longrightarrow \mathbb{Z}/2 \longrightarrow 1$$

respectively, together with the standard maps $\mathrm{Pin}_n^\pm \to \mathrm{O}_n.$

When one of the groups we want to mix acts on space, we can specify a mixed symmetry type by the following data:

- a symmetry type $\rho_n \colon H_n \to O_n$, called the base symmetry type,
- the point group symmetry $\lambda: G \to O_d$,
- an extension

$$(1.25) 1 \longrightarrow H_n \longrightarrow \widetilde{H}_n \longrightarrow G \longrightarrow 1$$

specifying how they mix, and

• an extension $\tilde{\rho}_n \colon \tilde{H}_n \to \mathcal{O}_n$ of $\rho_n \colon H_n \to \mathcal{O}_n$.

Freed-Hopkins [FH16a, §9.2] relate Altland-Zirnbauer's symmetry classes of condensed-matter systems [Zir96, AZ97] to ten symmetry types in topology.⁷ Using this, we call the case H = Spin the class D case and H = Spin^c the class A case.

Let Y be a G-space. Then the map

$$(1.26) Y \times E\tilde{H}_n/H_n \longrightarrow Y$$

is a G-equivariant fiber bundle with fiber BH_n , and the total space maps to BO_n as specified by the virtual vector bundle

(1.27)
$$f: -(Y \times (E\widetilde{H}_n \times_{H_n} \mathbb{R}^n)) \longrightarrow Y \times E\widetilde{H}_n/H_n.$$

After stabilizing (i.e. letting $n \to \infty$), this is an equivariant local system of symmetry types over Y, so has equivariant phase homology groups $Ph_*^G(Y; f)$. Under Ansatz 1.22, $Ph_0^G(Y; f)$ models the group of invertible topological phases on Y in which fermion parity mixes with the spatial symmetry as specified by (1.25). The notion of G-equivariant phases for this symmetry type (without a reference space Y) is taken to mean G-equivariant phases on \mathbb{R}^d , where G acts on \mathbb{R}^d through λ .

Remark 1.28 (Change of symmetry type). We would like to be able to move information between instances of this data: for example, there should be forgetful maps from equivariant phases on a space to nonequivariant ones, and we model them with maps between phase homology groups for the two local systems of symmetry types.

Suppose we are given two instances of the data above. That is, we ask for a commutative diagram of Lie groups

$$(1.29) \qquad \begin{array}{c} 1 \longrightarrow H_n \longrightarrow \widetilde{H}_n \longrightarrow G \longrightarrow 1 \\ & \downarrow^{\varphi} & \downarrow^{\widetilde{\varphi}} & \downarrow^{\varphi_G} \\ 1 \longrightarrow H'_n \longrightarrow \widetilde{H'}_n \longrightarrow G' \longrightarrow 1 \end{array}$$

together with maps $\rho_n: H_n \to \mathcal{O}_n$ and $\rho'_n: H_n \to \mathcal{O}_n$, $\lambda: G \to \mathcal{O}_d$ and $\lambda': G' \to \mathcal{O}_d$, and $\tilde{\rho}_n: \tilde{H}_n \to \mathcal{O}_n$ and $\tilde{\rho}'_n: \tilde{H}'_n \to \mathcal{O}_n$ which commute with the vertical maps in (1.29). Fix a G'-space Y; then through (1.27) this defines equivariant local systems of symmetry types f for G, resp. f' for G'. The maps between the data induce a pullback or forgetful map $\varphi^*: Ph^{G'}_*(Y; f') \to Ph^G_*(Y; f)$, where G acts on Y through φ_G . Using

⁷This "tenfold way" is a relativistic version of Dyson's threefold way [Dys62], and appears in many contexts in physics, including [Kit09, RSFL10, FM13, WS14, FH16a, KZ16, GM20, IT20].

Ansatz 1.22, we interpret this pullback map realizing an invertible phase on Y with a G'-symmetry to a phase with a G-symmetry.

The construction of φ^* amounts to checking that diagrams you would expect to commute do in fact commute. The data we gave induces a commutative diagram

(1.30)
$$\begin{array}{c} -(Y \times (E\widetilde{H}_n \times_{H_n} \mathbb{R}^n)) \longrightarrow Y \times E\widetilde{H}_n/H_n \\ \downarrow \\ -(Y \times (E\widetilde{H}'_n \times_{H'_n} \mathbb{R}^n)) \longrightarrow Y \times E\widetilde{H}'_n/H'_n. \end{array}$$

The rows define equivariant local systems symmetry types; then f and f' are the maps to $Y \times BO$. Let $\varphi^{\circ} \colon Sp^{G'} \to Sp^{G}$ be the map in which G acts on Borel G'-spectra through φ ; then, upon applying $\mathcal{I} \circ Th$, we obtain local systems \mathcal{L} , resp. \mathcal{L}' of Borel G-, resp. G'-spectra. To define phase homology, we assumed that an extension $\overline{\mathcal{L}}$ of \mathcal{L} to \overline{Y} exists, so choose such an extension; then $\overline{\mathcal{L}}' := \overline{\mathcal{L}} \circ \varphi^{\circ}$ is an extension of \mathcal{L} . We obtain from the inclusion $* \hookrightarrow \overline{Y}$ a commutative diagram of spectra

Thus, we get a map between the cofibers of the vertical arrows, and π_* of that map is the desired map on phase homology.

For us there are two particularly important examples.

- (1) Let $H'_n = H_n$ and G = 1, which forces $\tilde{\varphi} \colon \tilde{H}_n \to \tilde{H}'_n$ to be the inclusion $H'_n \to \tilde{H}'_n$. The above construction produces a map from *H*-equivariant phase homology to nonequivariant phase homology on *Y*, which we interpret as modeling the forgetful map from phases with a *G*-symmetry to phases without a *G*-symmetry.
- (2) Let G' = G, $H'_n = SO_n$. and H_n be either $Spin_n$ or $Spin_n^c$, with φ the usual map. In this case the pullback map goes from equivariant phase homology where the base symmetry type is SO to equivariant phase homology where the base symmetry type is Spin or Spin^c. We interpret this in physics as modeling the procedure that regards a bosonic phase as a fermionic phase by adding some trivial fermionic degrees of freedom. This is analogous to the procedure which regards an oriented TFT as a spin TFT that does not depend on the spin structure.

Crucially for computations, we can simplify the equivariant phase homology groups for the symmetry types in (1.27) into a description not requiring equivariant or parametrized stable homotopy theory.

Proposition 1.32. There is an isomorphism

(1.33)
$$Ph_0^G(\mathbb{R}^d; f) \xrightarrow{\cong} [(B\widetilde{H})^{d-\lambda-\widetilde{\rho}}, \Sigma^{d+2}I_{\mathbb{Z}}]$$

natural for changing the symmetry type in the sense of Remark 1.28.

Proof. We want to compute the twisted equivariant Borel-Moore homology for this equivariant local system of symmetry types, where $Y = \mathbb{R}^d$ with G acting through λ . This amounts to the following: one-point compactify to a local system over S^{λ} ; take the colimit of the local system and call it E; then compute $[\mathbb{S}, E]^G$ (in the notation of [FH19a]; this means $\pi_0(\operatorname{Map}(\mathbb{S}, E)^{hG})$). Now, the local system $(\mathcal{I} \circ \operatorname{Th} \circ f) \colon S^{\lambda} \to Sp^G$ is nonequivariantly the trivial local system with fiber $\operatorname{Map}(MTH, \Sigma^2 I_{\mathbb{Z}})$, so $E \simeq S^{\lambda} \wedge \operatorname{Map}(MTH, \Sigma^2 I_{\mathbb{Z}})$; in general, G can act nontrivially on both S^{λ} and MTH, but always acts trivially on $\Sigma^2 I_{\mathbb{Z}}$. Therefore we may follow [FH19a, (3.6)] and identify

(1.34)
$$\operatorname{Map}(\mathbb{S}, S^{\lambda} \wedge \operatorname{Map}(MTH, \Sigma^{2}I_{\mathbb{Z}})) \simeq \operatorname{Map}(\mathbb{S}^{d-\lambda} \wedge MTH, \Sigma^{d+2}I_{\mathbb{Z}})$$

though the *G*-action on $\mathbb{S}^{d-\lambda} \wedge MTH$ is not the diagonal action, but rather the induced *G*-action on the Thom spectrum of the *G*-equivariant virtual bundle $(d - \underline{\lambda} - \rho) \rightarrow BH$ (see [FH16a, §6.2.2]).

Since G acts trivially on $\Sigma^{d+2}I_{\mathbb{Z}}$,

(1.35)
$$\operatorname{Map}(\mathbb{S}^{d-\lambda} \wedge MTH, \Sigma^{d+2}I_{\mathbb{Z}})^{hG} \simeq \operatorname{Map}((\mathbb{S}^{d-\lambda} \wedge MTH)_{hG}, \Sigma^{d+2}I_{\mathbb{Z}}).$$

It now suffices to show that

(1.36)
$$(\mathbb{S}^{d-\lambda} \wedge MTH)_{hG} \simeq (B\widetilde{H})^{-\rho-\lambda+d}$$

Ando-Blumberg-Gepner-Hopkins-Rezk [ABG⁺14a, Proposition 1.20] show that the Thom spectrum of a virtual bundle $V \to X$, identified with a map $V: X \to BO$, is the homotopy colimit

(1.37)
$$X^{V} \simeq \operatorname{hocolim}\left(X \xrightarrow{V} BO \xrightarrow{BJ} BGL_{1}(\mathbb{S}) \longrightarrow \mathbb{S}p\right),$$

where the notation means to interpret X as, through its fundamental ∞ -groupoid, providing a diagram in the ∞ -category S_P of spectra. Here $BGL_1(S)$ is the classifying space of stable spherical fibrations [Sta63, May77] and $BJ: BO \rightarrow BGL_1(S)$ is a form of the J-homomorphism [Whi42, May77]. Heuristically, (1.37) says that the virtual vector bundle V defines a local system of \wedge -invertible spectra, with the fiber at a point $x \in X$ given by S^{V_x} , and that the Thom spectrum is obtained from an associated bundle construction. See [ABG⁺14a, ABG⁺14b] for more detail on this approach to Thom spectra.

Homotopy quotients are also homotopy colimits, meaning

(1.38a)
$$(\mathbb{S}^{d-\lambda} \wedge MTH)_{hG} = \underset{\text{pt/}G}{\text{hocolim}} \Big(\text{hocolim} \Big(BH \xrightarrow{d-\underline{\lambda}-\rho} BO \xrightarrow{BJ} BGL_1(\mathbb{S}) \longrightarrow \mathbb{S}p \Big) \Big)$$

where G acts on the spectra in the diagram through its action on λ , as well as on BH, as prescribed by the extension (1.25). This in particular implies the double homotopy colimit above simplifies into a single homotopy colimit over a $B\tilde{H}$ -shaped diagram:

(1.38b)
$$\simeq \operatorname{hocolim}\left(B\widetilde{H} \xrightarrow{d-\lambda-\widetilde{\rho}} BO \xrightarrow{BJ} BGL_1(\mathbb{S}) \longrightarrow \mathbb{S}p\right),$$

which by (1.37) is the Thom spectrum for $d - \lambda - \tilde{\rho} \to B\tilde{H}$, proving (1.36).

Our next step in §2 is to simplify $(B\tilde{H})^{d-\lambda-\tilde{\rho}}$. This allows both for a general formulation of the fermionic crystalline equivalence principle as well as explicit calculations.

The following lemma will be helpful for simplifying Thom spectra.

Theorem 1.39 (Relative Thom isomorphism). Let $\rho: H \to O$ be a symmetry type with the two-out-of-three property, i.e. an H-structure on any two of E, F, or $E \oplus F$ induces one on the third. If $V, W \to X$ are virtual vector bundles such that V has an H-structure, then there is an equivalence

$$(1.40) MTH \wedge X^W \xrightarrow{\simeq} MTH \wedge X^{V \oplus W}$$

Proof. The two-out-of-three property gives MTH an E_{∞} -ring structure, which is needed for some of the constructions we employ from [ABG⁺14a, ABG⁺14b] below.

Up to equivalence, the Thom spectrum of a virtual vector bundle $E \to X$ depends only on the homotopy class of the map $f_E: X \to BO \to B\operatorname{GL}_1(\mathbb{S})$, where the first map is given by E, and the second map is the *J*-homomorphism, as in (1.37). Smashing with *MTH* corresponds to composing f_E with the map $B\operatorname{GL}_1(\mathbb{S}) \to B\operatorname{GL}_1(MTH)$ induced by the Hurewicz map $\mathbb{S} \to MTH$ [ABG⁺14b, §1.4], and in particular, up to equivalence, $MTH \wedge X^E$ only depends on the homotopy type of the map $X \to B\operatorname{GL}_1(MTH)$.

Because MTH is an E_{∞} -ring spectrum, $BGL_1(MTH)$ is a grouplike E_{∞} -space, and the composition $\psi \colon BO \to BGL_1(\mathbb{S}) \to BGL_1(MTH)$ is a map of grouplike E_{∞} -spaces, where BO carries the E_{∞} structure coming from direct sum. This means that $[X, BGL_1(MTH)]$ is naturally an abelian group, and that if we define classes in this group using virtual vector bundles $V, W \to X$ to map to BO then composing with ψ , the class of $E \oplus F$ is the sum of the classes of V and W.

An *H*-structure on *V* trivializes the map $X \to BO \xrightarrow{\psi} BGL_1(MTH)$ defined by *V*, so the class of the map defined by $V \oplus W$ is equal to the class of the map defined by *W* in the abelian group $[X, BGL_1(MTH)]$. \Box

2. The fermionic crystalline equivalence principle

In this section, our goal is to state and prove the FCEP, Theorem 2.8, identifying phase homology groups in classes D and A with groups of deformation classes of invertible field theories. Assuming Ansatz 1.22, this leads to the more familiar version of the FCEP: crystalline equivalence principles are first introduced by Thorngren-Else [TE18]: the idea is to equate the classification of crystalline topological phases of matter for some group G acting on spacetime with a classification of a different kind of topological phases of matter, in which G is part of the internal symmetry group. Then one may use preexisting techniques for phases without a spatial symmetry to classify phases with the specified G-action on space.

The best-understood crystalline equivalence principles are for bosonic SPTs, as first considered by Thorngren-Else [TE18]. "Bosonic" does not have a precise mathematical translation here; these are phases for which the symmetry type is built using SO or O rather than Spin, Spin^c , Pin^{\pm} , and so on. If a group G acts on space by orientation-preserving symmetries and H is SO or O, the classification of crystalline SPTs in dimension n with symmetry type H and this G-action is identified with the classification of SPTs for $H = \text{SO} \times G$. To what extent this is an ansatz or a theorem depends on one's model for crystalline SPTs: Freed-Hopkins [FH19a, Example 3.5] derive it as a corollary of their ansatz.⁸ For other derivations of the bosonic crystalline equivalence principle from different ansatzes, see Jiang-Ran [JR17] and Thorngren-Else [TE18, ET19].

The fermionic analogue of this statement is more complicated because there are more ways for G to mix with the symmetry type. Thorngren-Else [TE18, §VII.B], Cheng-Wang [CW18], Zhang-Wang-Yang-Qi-Gu [ZWY⁺20], and Zhang-Wang-Yang-Gu [ZYQG20, §V] all study examples in which an FCEP holds, and each paper discusses that such a principle would have to account for the different ways in which G mixes with H: crystalline phases for which the spatial G-symmetry does not mix with fermion parity correspond to phases with an internal G-symmetry that does mix with fermion parity, and vice versa. Examples of this twisted correspondence also appear in work of Freed-Hopkins [FH19a, Example 3.5], Guo-Ohmori-Putrov-Wan-Wang [GOP⁺20], and Mao-Wang [MW20], though until now there was no precise general version of the FCEP.

Our version of the FCEP applies in Altland-Zirnbauer classes A and D (i.e. H = Spin or $H = \text{Spin}^c$), for all compact Lie groups G acting on faithfully on space, and all ways in which G may mix with fermion parity. The slogan "mixed crystalline goes to unmixed internal, and vice versa" is a little hard to glean from the result when the G-action includes reflections, but we obtain an equivalence from phase homology groups for certain equivariant local systems of symmetry types, which under Ansatz 1.22 stands in for groups of crystalline SPT phases, to groups of deformation classes of IFTs, which under Freed-Hopkins' ansatz [FH16a] model groups of phases without spatial symmetries.

To precisely state our FCEP, we must fix some data.

Data 2.1.

- Let H denote the base symmetry type, which today is either of the infinite-dimensional topological groups Spin or Spin^c.
- Let G be a compact Lie group, $\lambda: G \to O_d$ be a faithful representation, and $V_{\lambda} := EG \times_G \mathbb{R}^d \to BG$ be the associated vector bundle.
- Let $\xi: G \to O_{d'}$ be another faithful representation and $V_{\xi} \to BG$ be the associated vector bundle. Let $1 \to \mu_2 \to \widetilde{G} \to G \to 1$ be the central extension classified by $w_2(V_{\lambda}) + w_1(V_{\lambda})^2 \in H^2(BG; \mu_2)$. Here μ_2 denotes the group of square roots of unity.
- Let $\widetilde{H} := H \times_{\mu_2} \widetilde{G}$. Let ρ be the composition $\widetilde{H} \to H \to O$ and $V \to B\widetilde{H}$ be the associated tautological vector bundle.

For us, ξ and λ are usually the same, but they differ when $G = \mathbb{Z}/2$ acts on \mathbb{R}^d by inversion in the case of spin-1/2 fermions: here ξ is the sign representation $\sigma: \mathbb{Z}/2 \to O_1$, but $\lambda = d\sigma$. See §4.2 for more detail.

Definition 2.2. The spin-1/2 equivariant local system of symmetry types for the above data is the *G*-equivariant parametrized symmetry type $f_{1/2}: B\tilde{H} \to \mathbb{R}^{d'} \times BO$ which sends $x \mapsto (0, B\rho(x))$, and in which *G* acts on \mathbb{R}^d through λ . The spinless equivariant local system of symmetry types f_0 is defined in the same way, except using $H \times G$ instead of \tilde{H} .

⁸If G acts by reflections, almost as nice of a story is still true, but the internal G-symmetry mixes with H. Thorngren-Else [TE18] and Freed-Hopkins [FH19a, Example 3.5] discuss this case too.

Definition 2.3. Recall that H is either Spin or Spin^c. Let $\dagger \in \{-, c\}$ be - if H = Spin and c otherwise. The *spinless internal symmetry type* is the symmetry type

- $(-V, d V_{\lambda})$: $BH \times BG \to BO$, if λ is pin[†], or
- $(-V, V_{\xi} + \text{Det}(V_{\xi}) V_{\lambda}): BH \times BG \to BO$, if λ is not pin[†].

For shorthand, we denote this symmetry type $\rho(0): BH \times BG \to BO$.

Definition 2.4. The spin-1/2 internal symmetry type is the symmetry type

(2.5)
$$(-V, d - V_{\lambda}) \colon BH \times BG \to BO.$$

For shorthand, we denote this symmetry type $\rho(1/2)$: $BH \times BG \to BO$.

Remark 2.6. The internal symmetry types probably look pretty arbitrary. This is because of the generality of our setup: in some cases of interest, we can rewrite these symmetry types in ways which more closely resembles the proposals of Thorngren-Else [TE18, VII.B], Cheng-Wang [CW18], and Zhang-Wang-Yang-Qi-Gu [ZWY⁺20] for the FCEP in specific cases.

Suppose $\lambda = \xi$ and $\operatorname{Im}(\lambda) \subset \operatorname{SO}_d$ but does not lift across $\operatorname{Spin}_d \twoheadrightarrow \operatorname{SO}_d$. Then, the spinless internal symmetry type simplifies to $BH \times BG \to BO$, where the map is just projection onto the first factor followed by the usual map $BH \to BO$. That is, for representations with image contained in SO_d , the FCEP switches the "unmixed" (i.e. $BH \times BG$) and "mixed" (i.e. $B(H \times_{\mu_2} \widetilde{G})$) symmetry types when passing between crystalline and internal phases. This matches predictions by Thorngren-Else [TE18] and Cheng-Wang [CW18].

Freed-Hopkins [FH16a, Corollary 8.21] show that the group of deformation classes of reflection-positive IFTs with symmetry type $\rho': H' \to O$ in (space) dimension n is naturally isomorphic to⁹

$$[MTH', \Sigma^{d+2}I_{\mathbb{Z}}].$$

Theorem 2.8 (Fermionic crystalline equivalence principle). There are isomorphisms

(2.9a)
$$Ph_k^G(\mathbb{R}^d; f_0) \xrightarrow{\cong} [MT\rho(1/2), \Sigma^{d+k+2}I_{\mathbb{Z}}]$$

(2.9b)
$$Ph_k^G(\mathbb{R}^d; f_{1/2}) \xrightarrow{\cong} [MT\rho(0), \Sigma^{d+k+2}I_{\mathbb{Z}}].$$

Assuming Ansatz 1.22, the physics implication of this theorem is that the abelian group of crystalline SPT phases for the spinless equivariant local system of symmetry types is naturally isomorphic to the abelian group of deformation classes of IFTs for the spin-1/2 internal symmetry type; and the classification of crystalline SPT phases for the spin-1/2 equivariant local system of symmetry types is naturally isomorphic to the abelian group of deformation classes of IFTs of the spinless internal symmetry types.

We break the proof of Theorem 2.8 down into a few steps. First, Proposition 1.32 simplifies the question into one of ordinary stable homotopy theory.¹⁰ We obtain Thom spectra for vector bundles over $B\tilde{H}$, and to finish we must compare these spectra to $MTH \wedge (BG)^E$, where $E \to BG$ is some rank-zero virtual vector bundle. This comparison, in the form of *shearing arguments*, is the core of the proof: we prove Theorem 2.11 (H = Spin) and Theorem 2.24 $(H = \text{Spin}^c)$ establishing the homotopy equivalences we need, and after that proving Theorem 2.8 amounts to verifying that the outputs of Theorems 2.11 and 2.24 simplifying the crystalline symmetry types match the Thom spectra for the internal symmetry types in Definitions 2.3 and 2.4.

The proofs of Theorems 2.11 and 2.24 resemble the proofs of the more standard equivalences

- (2.10a) $MTPin^+ \simeq MTSpin \wedge (B\mathbb{Z}/2)^{1-\sigma}$
- (2.10b) $MTPin^{-} \simeq MTSpin \wedge (B\mathbb{Z}/2)^{\sigma-1}$
- (2.10c) $MTPin^c \simeq MTSpin^c \wedge (B\mathbb{Z}/2)^{\pm(1-\sigma)}$
- (2.10d) $MTSpin^c \simeq MTSpin \wedge (BSO_2)^{\pm (2-V_2)}$

 $^{^{9}}$ Strictly speaking, Freed-Hopkins' theorem classifies only the invertible *topological* field theories, which form the torsion subgroup of (2.7), and they conjecture that the entire group classifies all reflection-positive IFTs.

¹⁰For the spinless equivariant symmetry type, this is just [FH19a, Example 3.5].

where $\sigma \to B\mathbb{Z}/2$ and $V_2 \to BSO_2$ denote the respective tautological line bundles. These decompositions were first proven by Kirby-Taylor [KT90a, Lemma 6] (pin⁺), Peterson [Pet68, §7] (pin⁻), and Bahri-Gilkey [BG87a, BG87b] (spin^c and pin^c). For a unified proof of all of these equivalences, see Freed-Hopkins [FH16a, §10].

2.1. Case H =Spin.

Theorem 2.11 (Shearing, class D). Let $V \to B\widetilde{H}$ be the tautological bundle.

(1) Suppose V_{ξ} admits a pin⁻ structure. Then there is an equivalence

$$(2.12) (B\widetilde{H})^{d-V_{\lambda}-V} \xrightarrow{\simeq} MTSpin \wedge (BG)^{d-V_{\lambda}}$$

(2) If V_{ξ} does not admit a pin⁻ structure, there is an equivalence

(2.13)
$$(B\widetilde{H})^{d-V_{\lambda}-V} \xrightarrow{\simeq} MTSpin \wedge (BG)^{V_{\xi}+\operatorname{Det}(V_{\xi})-V_{\lambda}-d'-1+d}.$$

We will most often consider case (2) with $\lambda = \xi$, in which case we learn $(B\widetilde{H})^{d-\lambda-V} \simeq MTSpin \wedge (BG)^{\text{Det}(V_{\lambda})-1}$.

Proof. Case (1) is by far the easier of the two: V_{ξ} admits a pin⁻ structure iff $w_2(V_{\xi}) + w_1(V_{\xi})^2 = 0$ iff the extension $1 \to \mu_2 \to \tilde{G} \to G \to 1$ splits. Since $\mu_2 \subset \tilde{G}$ is central, a splitting induces isomorphisms $\tilde{G} \cong \mu_2 \times G$ and $\tilde{H}_n \cong \operatorname{Spin}_n \times G$. Passing to classifying spaces, this identifies $d - V_{\lambda} - V \colon B\tilde{H} \to BO$ with $-V \boxplus (d - \lambda) \colon BSpin \times BG \to BO$; then take Thom spectra.

On to case (2). In this case, in $H^2(B\widetilde{H};\mu_2)$, $w_2(V_{\xi})+w_1(V_{\xi})^2 = w_2(V)$, so the map $V+V_{\xi}+\text{Det}(V_{\xi}): B\widetilde{H} \to BSO$ lifts across $BSpin \to BSO$. Choose such a lift.

Proposition 2.14. The induced map

(2.15)
$$(V + V_{\xi} + \operatorname{Det}(V_{\xi}), \xi) \colon BH \longrightarrow B\operatorname{Spin} \times BG$$

is a homotopy equivalence commuting with the maps to BSO.

The proof is due to Freed-Hopkins [FH16a, §10].

Proof. We will show that the commutative square

(2.16a)
$$BH \longrightarrow BSpin \\ \downarrow_{B(\pi_1 \oplus \pi_2)} \\ \downarrow_{BSO \times BG} \xrightarrow{B(\mathrm{id} \oplus \xi)} BSO$$

is homotopy Cartesian. Any two homotopy pullbacks of the same diagram are weakly equivalent, with the weak equivalence intertwining the maps to BSO. Since there is also a homotopy pullback square

then $B\tilde{H} \simeq B$ Spin × BG; this equivalence is realized by (2.15) because that is the only possibility that intertwines the maps in (2.16a) and (2.16b).

To fulfill the promise that (2.16a) is a homotopy pullback square, begin with the commutative diagram of short exact sequences

$$(2.17) \qquad \begin{array}{c} 1 \longrightarrow \mu_{2} \longrightarrow \widetilde{H}_{n} \xrightarrow{(\pi_{1},\pi_{2})} \operatorname{SO}_{n} \times G \longrightarrow 1 \\ \\ \\ \\ \\ \\ \\ 1 \longrightarrow \mu_{2} \longrightarrow \operatorname{Spin}_{n+d} \longrightarrow \operatorname{SO}_{n+d} \longrightarrow 1. \end{array}$$

This induces a map of fiber sequences

e.g. $B\tilde{H}$ is the fiber of $w_2: BSO \times BG \to K(\mu_2, 2)$. The left square in such a pullback is always homotopy Cartesian, and in (2.18) the left square can be identified with (2.16a).

Including the maps down to BSO produces the commutative diagram

(2.19)
$$B\widetilde{H} \xrightarrow{(V+V_{\xi}+\operatorname{Det}(V_{\xi}),\xi)} B\operatorname{Spin} \times BG$$
$$\xrightarrow{-V} V+V_{\xi}+\operatorname{Det}(V_{\xi})$$
BSO.

Taking Thom spectra of the vertical maps, the shearing map induces a homotopy equivalence

(2.20)
$$(B\widetilde{H})^{-V} \xrightarrow{\simeq} MTSpin \wedge (BG)^{V_{\xi} + \operatorname{Det}(V_{\xi}) - d' - 1}$$

To finish, we subtract V_{λ} from the vertical arrows in (2.19), then take Thom spectra again.

2.2. Case $H = \operatorname{Spin}^{c}$. Let $\widetilde{H}_{n} := \operatorname{Spin}_{n}^{c} \times_{\mu_{2}} \widetilde{G}$, and define \widetilde{H} similarly. The shearing argument is scarcely different than for Theorem 2.11, but it will be useful to rephrase \widetilde{H}_{n} using the circle group \mathbb{T} instead of μ_{2} .

The extension of G by μ_2 defines an extension of G by \mathbb{T} by pushing forward along the inclusion $\mu_2 \hookrightarrow \mathbb{T}$:

In cohomology, this construction is classified by the Bockstein map $H^2(BG; \mu_2) \to H^3(BG; \mathbb{Z})$. Let $\widehat{H}_n := \operatorname{Spin}_n^c \times_{\mathbb{T}} \widehat{G}$ and $\widehat{H} := \operatorname{Spin}^c \times_{\mathbb{T}} \widehat{G}$. The map $\widetilde{G} \to \widehat{G}$ induces maps $\varphi_n \colon \widetilde{H}_n \to \widehat{H}_n$ and $\varphi \colon \widetilde{H} \to \widehat{H}; \varphi$ is the colimit of the φ_n s.

Lemma 2.22. The maps $\varphi_n \colon \widetilde{H}_n \to \widehat{H}_n$ are isomorphisms of Lie groups.

Proof. Write down the commutative diagram

and apply the five lemma.

(2.23)

And now we shear. Recall our notation from Data 2.1.

Theorem 2.24 (Shearing, class A).

(1) Suppose V_{ξ} admits a pin^c structure. Then there is an equivalence

$$(2.25) (B\widehat{H})^{d-V_{\lambda}-V} \xrightarrow{\simeq} MTSpin^{c} \wedge (BG)^{d-V_{\lambda}}$$

(2) If V_{ξ} does not admit a pin^c structure, there is an equivalence

(2.26)
$$(B\widehat{H})^{d-V_{\lambda}-V} \xrightarrow{\simeq} MTSpin^{c} \wedge (BG)^{V_{\xi}+\operatorname{Det}(V_{\xi})-V_{\lambda}-d'+1-d}.$$

Again, we most often use case (2) when $\lambda = \xi$, in which case the right-hand side simplifies to $MTSpin^c \wedge (BG)^{\text{Det}(V_{\lambda})-1}$.

Proof. The proof is barely different than that of Theorem 2.11; we indicate only the differences. In that theorem, the engine of the proof when V_{ξ} was not pin⁻ was the map (2.15) from $B(\operatorname{Spin} \times_{\mu_2} \widetilde{G}) \to B\operatorname{Spin} \times BG$. Here, V_{ξ} is not pin^c, so $V_{\xi} \oplus \operatorname{Det}(V_{\xi})$ is oriented but not spin^c. We have that if $\beta \colon H^2(B\widehat{H};\mu_2) \to H^3(B\widehat{H};\mathbb{Z})$ is the Bockstein, $\beta(w_2(V_{\xi}) + w_1(V_{\xi})^2 + w_2(V)) = 0$, so $V + V_{\xi} + \operatorname{Det}(V_{\xi})$, interpreted as a map $B\widehat{H} \to BSO$, lifts to $B\operatorname{Spin}^c$. Our analogue of (2.15) is

(2.27)
$$(V + V_{\xi} + \operatorname{Det}(V_{\xi}), \xi) \colon B\widehat{H} \longrightarrow B\operatorname{Spin}^{c} \times BG.$$

As in Proposition 2.14, this is a homotopy equivalence commuting with the maps down to BSO. The proof is almost the same, though we replace Spin with Spin^c in (2.16a) and (2.16b), μ_2 with \mathbb{T} in (2.17), and $K(\mu_2, 2)$ with $K(\mathbb{Z}, 3)$ in (2.18).

2.3. Putting it together. The hard work of the proof is already done.

Proof of Theorem 2.8. By Proposition 1.32,

where $X := (B\tilde{H})^{d-V_{\lambda}-V}$. Then Theorem 2.11 (H = Spin) and Theorem 2.11 $(H = \text{Spin}^c)$ split this into $MTH \wedge (BG)^E$ for some rank-zero virtual vector bundle E. For f_0 , because $\tilde{H} \cong H \times G$, Proposition 1.32 gets us to $MTH \wedge (BG)^E$ without having to shear. The only thing left to do is compare these Thom spectra to Definitions 2.3 and 2.4, and sure enough, they match.

3. Computations in examples: summary of results and some generalities

In the next two sections, we study the fermionic crystalline equivalence principle in many examples where the symmetry is given by a two- or three-dimensional point group. Here, we summarize the results and some takeaways for researchers interested in crystalline phases; for more detailed results of computations of groups of phases, see Tables 1, 2, 3, 4, 5, and 6.

In §3.1, we indicate some example phases predicted by our phase homology calculations that have not been previously studied to our knowledge, and which might have accessible or interesting lattice realizations. We also summarize which of our calculations correspond to phases already studied in the literature. In §3.2, we briefly review the computational techniques we use to study phase homology groups, namely the Adams and Atiyah-Hirzebruch spectral sequences. In §3.3, we use the Adams filtration to characterize which invertible field theories with \tilde{H} -structure actually only require weaker structure, such as an SO × G-structure; this is believed to model the phenomenon in physics of phases which appear to be fermionic, but are in fact bosonic phases that are not fermionic in an interesting way. Finally, in §3.4, we gather some lemmas we use repeatedly in the coming sections. The reader interested in the computations can read §3.1 and §3.2, returning to the other sections later.

3.1. Some interesting phases to study. In §§4–5, we compute equivariant phase homology groups for many 2- and 3-dimensional point groups. Using Ansatz 1.22, these computations yield predictions of groups of invertible topological phases. This is a lot of data, so we take the opportunity here to highlight which of our predictions would be interesting to study by other means, e.g. by arguing on the lattice.

We first study some cases already present in the literature and find agreement, including reflections in Altland-Zirnbauer classes D and A ($\S4.1$), inversions in classes D and A ($\S4.2$), cyclic groups acting by rotations in classes D and A ($\S4.3$), and dihedral groups acting by rotations and reflections in class D ($\S4.4$). In all cases we consider both spinless and spin-1/2 fermions.

In addition, we study rotations in class A and many three-dimensional point group symmetries in classes D and A: dihedral groups acting by rotations, pyritohedral symmetry, and chiral and full tetrahedral, octahedral, and icosahedral symmetries. We consider symmetry types with both spinless and spin-1/2 fermions. To the best of our knowledge, these symmetry types have not been studied in the literature, so we indicate some of our predictions that might be interesting to study.

- (1) In §4.4.3 and §4.4.4, we compute phase homology groups for the local systems of symmetry types corresponding to class A phases in which the dihedral group D_{2n} acts by rotations and reflections.
 - (a) In dimension d = 2, we predict using Theorems 4.46 and 4.53 a phase generating a $\mathbb{Z}/2n$ for even n with spinless fermions.

- (b) In dimension d = 3, we would be interested in the predicted Z/8 ⊕ Z/2 for n odd, with either spin-1/2 or spinless fermions (based on (4.45), Theorem 4.45), as well as a phase generating a Z/4 for n even with spin-1/2 fermions (based on Theorems 4.57 and 4.60).
- (2) We predict using §5.2 a $\mathbb{Z}/2 \oplus \mathbb{Z}/2$ of 3d class D phases with a pyritohedral symmetry and spinless fermions. In class A, we predict a phase generating a $\mathbb{Z}/4$ subgroup, again with spinless fermions.
- (3) In §5.1, we calculate equivariant phase homology groups on \mathbb{R}^3 for A_4 acting by tetrahedral symmetry and find that for classes A and D and the spinless and spin-1/2 cases, the zeroth phase homology groups all vanish. Under our ansatz, this predicts there are no nontrivial fermionic phases equivariant for a tetrahedral symmetry in these cases. Can this be seen using a lattice argument?
- (4) We predict in §5.3.1 that for 3d class D phases with a full tetrahedral symmetry (i.e. including reflections) and spinless fermions, there is a phase generating a Z/4 subgroup. This phase homology calculation required the most involved mathematical argument, and it would be interesting to see a physical description. A physical interpretation of Proposition 5.46 specifically or an argument averting it would provide some insight into the meaning in physics of the Adams spectral sequence as a tool for studying fermionic phases.

Our computations predict plenty of other phases, but many of them either have Adams filtration zero (see §3.3) and therefore are not predicted to be intrinsically fermionic, or have more complicated symmetry types, such a full octahedral symmetry, that would be harder to study on the lattice.

Remark 3.1. In the computations we make in the next several sections, we generally report more bordism groups than we need to determine the phase homology groups corresponding to groups of invertible phases: to compute the group of *n*-dimensional invertible field theories with symmetry type $H \to O$, we need the torsion subgroup of $\pi_n(MTH)$ and the free summand in $\pi_{n+1}(MTH)$. Bordism has other applications in geometry and physics, so we usually report all bordism groups $\pi_k(MTH)$ that follow from the calculations that we need for crystalline phases. When $k \ge n + 1$, these provide information about higher-dimensional crystalline phases; for $k < \dim(\lambda)$, though, it is not clear what a crystalline phase could mean when there are not enough space dimensions for G to act by λ , and we do not give a physical meaning to these computations. See [GOP⁺20] for some discussion when space*time* is dim (λ) -dimensional.

3.2. Methods of computation. In this section, we summarize the techniques we use to make these computations, and gather a few auxiliary lemmas we need along the way. Most of our computations can be reframed as computing certain twisted ko- and ku-homology groups of finite groups in low degrees; the reader interested in learning how to perform such computations is encouraged to refer to the monographs of Bruner-Greenlees [BG03, BG10] on connective ko- and ku-theory, as well as Beaudry-Campbell's article [BC18] on using the Adams spectral sequence to compute ko-theory.

- **Computing spin bordism:** Let ko denote the connective real K-theory spectrum. Anderson-Brown-Peterson [ABP67] show that the Atiyah-Bott-Shapiro map $MTSpin \to ko$ [ABS64] is 7-connected, meaning that for any space or spectrum X, the induced map $\Omega_k^{\text{Spin}}(X) \to ko_k(X)$ is an isomorphism for $k \leq 7$. We often pass between spin bordism and ko-theory without comment. We compute the free and 2-torsion summands of $ko_*(X)$ using the Adams spectral sequence; see below. The forgetful map $MTSpin \to MTSO$ induces an equivalence on odd-primary torsion, so to compute odd-primary torsion, we typically compute $\Omega_*^{\text{SO}}(X)$ via the Atiyah-Hirzebruch spectral sequence, which we also discuss below.
- **Computing spin**^c **bordism:** Let ku denote connective complex K-theory. Anderson-Brown-Peterson [ABP67] also produce a 7-connected map $MTSpin^c \to ku \vee \Sigma^4 ku$; we will also use the Adams spectral sequence to determine the free and 2-torsion summands of $ku_*(X)$, as described below. The forgetful map $MTSpin^c \to MTSO \wedge (BU_1)_+$ induces an equivalence on odd-primary torsion, so we compute $\Omega_*^{SO}(X \times BU_1)$, typically with the Atiyah-Hirzebruch spectral sequence.

Now we briefly introduce the Adams and Atiyah-Hirzebruch spectral sequences in the ways that we use them.

3.2.1. The Adams spectral sequence. The (2-primary) Adams spectral sequence [Ada58, Theorems 2.1, 2.2] computes the 2-completion of the homotopy groups of a pointed space or spectrum X. Its E_2 -page is

(3.2)
$$E_2^{s,t} = \operatorname{Ext}_{\mathcal{A}}^{s,t}(\widetilde{H}^*(X;\mathbb{Z}/2),\mathbb{Z}/2) \Longrightarrow \pi_{t-s}(X)_2^{\wedge}$$

Here \mathcal{A} is the 2-primary Steenrod algebra.

Remark 3.3. The usual bigrading convention for Adams spectral sequences places $E_r^{s,t}$ at coordinates (t-s,s). We follow this convention. The topological degree of an element at coordinates (t-s,s) in an Adams spectral sequence refers to t-s, and s is called its *filtration*.

There is a general change-of-rings theorem, where if \mathcal{B} is a graded Hopf algebra, $\mathcal{C} \subset \mathcal{B}$ is a graded Hopf subalgebra, and M and N are graded \mathcal{B} -modules, then there is a natural isomorphism

$$(3.4) \qquad \operatorname{Ext}_{\mathcal{B}}^{s,t}(\mathcal{B} \otimes_{\mathcal{C}} M, N) \xrightarrow{\cong} \operatorname{Ext}_{\mathcal{C}}^{s,t}(M, N).$$

When $X = ko \wedge Y$ or $ku \wedge Y$, this greatly simplifies the E_2 -page of (3.2). Inside the mod 2 Steenrod algebra \mathcal{A} , define the subalgebras $\mathcal{A}(1) \coloneqq \langle \operatorname{Sq}^1, \operatorname{Sq}^2 \rangle$ and $\mathcal{E}(1) \coloneqq \langle Q_0, Q_1 \rangle$;¹¹ then, Stong [Sto63] showed $\widetilde{H}^*(ko; \mathbb{Z}/2) \cong \mathcal{A} \otimes_{\mathcal{A}(1)} \mathbb{Z}/2$ and Adams [Ada61] showed $\widetilde{H}^*(ku; \mathbb{Z}/2) \cong \mathcal{A} \otimes_{\mathcal{E}(1)} \mathbb{Z}/2$. Both $\mathcal{A}(1)$ and $\mathcal{E}(1)$ are Hopf subalgebras of \mathcal{A} so (3.4) says we need only consider

(3.5a)
$$E_2^{s,t} = \operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(\widetilde{H}^*(X;\mathbb{Z}/2),\mathbb{Z}/2) \Longrightarrow \widetilde{ko}_{t-s}(X)_2^{\wedge}$$

(3.5b)
$$E_2^{s,t} = \operatorname{Ext}_{\mathcal{E}(1)}^{s,t} (\widetilde{H}^*(X;\mathbb{Z}/2),\mathbb{Z}/2) \Longrightarrow \widetilde{ku}_{t-s}(X)_2^{\wedge}.$$

This line of reasoning, first used by Davis [Dav74], is by now a standard trick in algebraic topology. For further reading, we recommend the paper of Beaudry-Campbell [BC18], who go into detail about how to define and calculate these Ext groups and work several examples over $\mathcal{A}(1)$. There are fewer worked examples of (3.5b) in the literature; see Bruner-Greenlees [BG03], Nguyen [Ngu09], Francis [Fra11, §5] and Al-Boshmki [AB16] for closely related calculations.

Our notation is standard in the $\mathcal{A}(1)$ -case, but since examples for $\mathcal{E}(1)$ are sparser, we record here a few notational conventions for working with $\mathcal{E}(1)$ -modules and this spectral sequence. When we draw $\mathcal{E}(1)$ -modules, we will use solid straight lines to denote Q_0 -actions and dashed curved lines to denote Q_1 -actions. Therefore, for example, $\mathcal{E}(1)$ as a module over itself looks like this.



For any $\mathcal{E}(1)$ -module M, $H^{*,*}(\mathcal{E}(1)) \coloneqq \operatorname{Ext}_{\mathcal{E}(1)}^{*,*}(\mathbb{Z}/2,\mathbb{Z}/2)$ acts on $\operatorname{Ext}_{\mathcal{E}(1)}^{s,t}(M,\mathbb{Z}/2)$, analogously to the case of $\mathcal{A}(1)$ -modules; if $M = \widetilde{H}^*(X;\mathbb{Z}/2)$, then just as over $\mathcal{A}(1)$, tracking this action through the Adams spectral sequence provides information about the action of ku_* on $\widetilde{ku}_*(X)$. Differentials are equivariant for this action, just like for the Adams spectral sequence over $\mathcal{A}(1)$. Since $\mathcal{E}(1)$ is an exterior algebra, Koszul duality provides an isomorphism of bigraded algebras

(3.7)
$$H^{*,*}(\mathcal{E}(1)) \cong \mathbb{Z}/2[h_0, v_1],$$

where $|h_0| = (1, 1)$ and $|v_1| = (1, 3)$ [BC18, Example 4.5.6]. We will denote an h_0 -action by a vertical line, and a v_1 -action by a lighter diagonal line. Like for ko, h_0 lifts to multiplication by 2; v_1 lifts to the action of the Bott element $\beta \in ku_2$ [BG03, §2.1].

We will often write $\operatorname{Ext}_{\mathcal{A}(1)}(M)$ for $\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(M,\mathbb{Z}/2)$, and similarly for $\mathcal{E}(1)$; when it is clear which subalgebra we are working over, we will just write $\operatorname{Ext}(M)$.

By now there is a large body of work using the Adams spectral sequence, especially over $\mathcal{A}(1)$, to compute things related to invertible field theories or invertible phases. This includes [Sto86, Kil88, Hil09, Fra11, FH16a, Cam17, BC18, GPW18, Guo18, FH19b, WW19a, WW19b, WWZ19, DL20a, DL20b, DL20c, GOP⁺20, KPMT20, LOT20, LT20, WW20a, WW20b, WW20c, WWZ20].

3.2.2. The Atiyah-Hirzebruch spectral sequence. The (homological) Atiyah-Hirzebruch spectral sequence [AH61] for oriented bordism has signature

(3.8)
$$E_{p,q}^2 = \widetilde{H}_p(X; \Omega_*^{\rm SO}) \Longrightarrow \Omega_{p+q}^{\rm SO}(X).$$

¹¹These generators are given in two different bases of \mathcal{A} ; the relations between them are $Q_0 = \mathrm{Sq}^1$ and $Q_1 = \mathrm{Sq}^1 \mathrm{Sq}^2 + \mathrm{Sq}^2 \mathrm{Sq}^1$.

In general, using the Atiyah-Hirzebruch spectral sequence can feel different depending on application-specific details, so we point the reference-minded reader to García-Etxebarria-Montero [GEM19, §2.2.2, §3] for an introduction and some examples which may be helpful.

There are many references using the Atiyah-Hirzebruch spectral sequence to compute things related to invertible field theories or invertible phases, such as [Kil88, Edw91, Mon15, Cam17, KT17, Mon17, Hsi18, SdBKP18, SSG18, Ste18, STY18, SXG18, Xio18, ET19, FH19a, GEM19, MM19, OSS19, Shi19, TY19, BLT20, DGL20, DH20, DL20c, ETS20, GOP⁺20, HH20, HKT20, Hor20, HTY20, JF20a, JF20b, KPMT20, LOT20, LT20, SFQ20, Tho20, TW20, WW20b, Yu20, DGG21, KLST21].

We use a few other spectral sequences in our computations, but only for one-off computations, so we address them when we get to them.

3.3. Adams filtration 0 phases are secretly bosonic. In Remark 1.28, we defined a map from phase homology with symmetry type SO to phase homology with symmetry types Spin or Spin^c and interpreted it as regarding bosonic SPT phases as fermionic SPT phases in a trivial way. Physicists studying fermionic SPT phases are often interested in the cokernel of this map, which is thought of as the group of intrinsically fermionic SPT phases. Because bosonic crystalline phases are relatively well-understood, e.g. in the work of Hermele, Huang, Song, and their collaborators [HSHH17, SHFH17, HH18, SHQ⁺19, SFQ20, SXH20] and via the bosonic crystalline equivalence principle of Thorngren-Else [TE18], we are most interested in intrinsically fermionic SPT phases.

The structure of the Adams spectral sequence allows us to identify the image of this bosonic-to-fermionic map on phase homology with little extra work. For more about the Adams spectral sequence, see §3.2; for now, we need only that phase homology groups, reinterpreted through Theorem 2.8 as groups of invertible field theories, are computed as homotopy groups of spectra, and that the homotopy groups of any spectrum M come with a canonical filtration called the $(mod \ 2)$ Adams filtration

(3.9)
$$\pi_n M = F_n^0 \supseteq F_n^1 \supseteq F_n^2 \supseteq \cdots$$

For more information, see [BC18, §4.7]. This has two properties which are important for us.

- (1) The Adams spectral sequence computes the Adams filtration: after 2-completing, the associated graded of (3.9) is the E_{∞} -page of the Adams spectral sequence, in that $E_{\infty}^{s,t} = \operatorname{gr}_s \pi_{t-s} M$.
- (2) If M = MTH is a Thom spectrum whose homotopy groups compute bordism groups, elements of the associated graded in degree 0 correspond to the 2-primary part of the group of deformation classes of invertible TFTs which depend on something weaker than an H-structure, such as a spin IFT which is defined by evaluating an oriented IFT on spin manifolds.

This means we can identify which invertible TFTs really use the *H*-structure, and which do not.

Now a little more detail. We do not need to say much more about (1): we depict Adams spectral sequences on a grid with coordinates (t-s, s), such as in Figure 1, right, so F_n^0/F_n^1 is found in the E_{∞} -page at coordinate (n, 0).

For (2), we make a simplifying assumption: that for the specific degree n we are investigating, $\pi_n MTH$ is 2-torsion. This assumption holds in all cases where we want to study the Adams filtration in this article, but if you want to relax it, see Remark 3.18. The assumption implies that up to extension questions on the E_{∞} -page, the mod 2 Adams spectral sequence fully determines $\pi_n MTH$,¹² and that the natural map

(3.10)
$$(\pi_n(MTH))^{\vee} := \operatorname{Hom}(\pi_n(MTH), \mathbb{C}^{\times}) \longrightarrow [MTH, \Sigma^{n+1}I_{\mathbb{Z}}]$$

is an isomorphism.

To pass from bordism groups to isomorphism class of invertible field theories, we must take character duals $A \mapsto A^{\vee} := \operatorname{Hom}(A, \mathbb{C}^{\times})$. This is a good thing, actually: a degree-0 element of $\operatorname{gr}_{\bullet}\pi_n(MTH)$ does not usually uniquely lift to an element of $\pi_n MTH$: the ambiguity is F_n^1 . But in $(\pi_n(MTH))^{\vee}$, we get a subgroup: the surjection

(3.11a)
$$\pi_n(MTH) \longrightarrow \pi_n(MTH)/F_n^1 \cong \operatorname{gr}_0\pi_n(MTH)$$

¹²Some extension questions can be addressed using the $H^{*,*}(\mathcal{A}(1))$ -action on the E_{∞} -page, but there are also hidden extensions which are harder to address. None of the calculations we make in this article manifest hidden non-split extensions; one example where they do occur is $H = \text{Spin} \times_{\mathbb{Z}/2} \mathbb{Z}/8$ [DDHM].

passes under character duality to an inclusion

(3.11b)
$$(\operatorname{gr}_0\pi_n(MTH))^{\vee} \hookrightarrow (\pi_n(MTH))^{\vee}$$

Therefore, in a mild abuse of notation, we refer to this subgroup of $(\pi_n(MTH))^{\vee}$, identified with a subgroup of the group isomorphism classes of invertible TFTs with *H*-structure, as the group of Adams filtration 0 invertible TFTs with *H*-structure.

It is a theorem [FH19b, §8.4] that this subgroup consists of theories closely related to classical Dijkgraaf-Witten theories [FQ93, §1].¹³ Isomorphism classes of these invertible TFTs are determined by their partition functions [FH16a, §5.3], so we specify these theories by their partition functions, which are bordism invariants $\Omega_n^H \to \mathbb{C}^{\times}$.

For the Adams spectral sequence, $E_2^{0,n} = \text{Ext}_{\mathcal{A}}^{0,n}(\widetilde{H}^*(MTH;\mathbb{Z}/2);\mathbb{Z}/2)$ is canonically identified with

(3.12)
$$\operatorname{Hom}_{\mathcal{A}}(H^*(MTH;\mathbb{Z}/2),\Sigma^n\mathbb{Z}/2),$$

which is a subspace of

(3.13)
$$\operatorname{Hom}_{\mathcal{A}b}(\widetilde{H}^n(MTH;\mathbb{Z}/2),\mathbb{Z}/2) \cong (\widetilde{H}^n(MTH;\mathbb{Z}/2))^{\vee}.$$

The fourth quadrant of the Adams spectral sequence is empty, so $E_{\infty}^{0,n}$ is a subspace of $E_2^{0,n}$. Take the sequence of maps

(3.14a)
$$\operatorname{gr}_0\pi_n(MTH) = E^{0,n}_{\infty} \longrightarrow E^{0,n}_2 \longrightarrow (\widetilde{H}^n(MTH;\mathbb{Z}/2))^{\vee}$$

and apply character duality:

(3.14b)
$$(\operatorname{gr}_0 \pi_n(MTH))^{\vee} \longleftarrow (E_2^{0,n})^{\vee} \longleftarrow \widetilde{H}^n(MTH; \mathbb{Z}/2)$$

Now compose with the Thom isomorphism to obtain

(3.14c)
$$\zeta \colon H^n(BH; \mathbb{Z}/2) \longrightarrow (\operatorname{gr}_0 \pi_n(MTH))^{\vee}.$$

That is, a degree- $n \mod 2$ cohomology class of BH determines an isomorphism class of Adams filtration 0 invertible TFTs, and all Adams filtration 0 invertible TFTs arise in this way. The map need not be injective, e.g. by the Wu formula when H = O.

Tracing this through Thom's collapse map tells us that given a cohomology class $\theta \in H^n(BH; \mathbb{Z}/2)$, the partition function $\zeta(\theta)$ is the bordism invariant which takes a closed *n*-manifold with *H*-structure $(M, f: M \to BH)$ and returns

(3.15)
$$\zeta(\theta)(M,f) = (-1)^{\langle f^*\theta, [M] \rangle}.$$

That is, use the *H*-structure to pull θ back to *M*, then evaluate it on the mod 2 fundamental class. This construction uses some aspects of the *H*-structure on *M*, but in the cases relevant to this paper, it is insensitive to the difference between Spin and O, which is believed to pass to the physicists' distinction between fermionic and bosonic phases.

Lemma 3.16. If $H = \text{Spin} \times_{\mu_2} \widetilde{G}$ or $H = \text{Spin}^c \times_{\mu_2} \widetilde{G}$, where \widetilde{G} is in Data 2.1, and $H' \coloneqq O \times G$, then the map $H \to H'$ of tangential structures induces a surjective map $H^*(BH'; \mathbb{Z}/2) \to H^*(BH; \mathbb{Z}/2)$, and therefore the partition functions (3.15) of the Adams filtration 0 theories only depend on the underlying H'-structure of an H-manifold.

Proof. First, the Spin case. We established a shearing equivalence $MTH \cong MTSpin \wedge X$, where X is a Thom spectrum of a rank-zero virtual vector bundle over BG, and this equivalence fits into a homotopy commutative diagram

¹³These theories are not quite the same thing as classical Dijkgraaf-Witten theories, which are TFTs of oriented manifolds with a principal *G*-bundle, and which use \mathbb{R}/\mathbb{Z} -valued cohomology, rather than $\mathbb{Z}/2$ -valued cohomology. Unoriented generalizations of classical Dijkgraaf-Witten theory are studied in more detail in work of Kim [Kim18, §6], the author [Deb20, §3.1], and You [You20].

Apply mod 2 cohomology and invoke the Thom isomorphism to obtain a commutative diagram

The map $H^*(BO; \mathbb{Z}/2) \to H^*(BSpin; \mathbb{Z}/2)$ is surjective, so the Künneth formula implies ξ is too, so the left-hand arrow $H^*(BH'; \mathbb{Z}/2) \to H^*(BH; \mathbb{Z}/2)$ is as well.

For $H = \text{Spin}^c$, the proof is the same $-B\text{Spin}^c$ has an additional characteristic class $c_1 \in H^2(B\text{Spin}^c; \mathbb{Z})$, but its mod 2 reduction is w_2 , so ξ is still surjective.

Remark 3.18. In all cases that one might reasonably encounter, the bordism group $\pi_n X$ is finitely generated, so we can ask what happens if it contains *p*-torsion for an odd prime *p* or free summands. For a *p*-torsion summand, the story is very similar: one instead uses the mod *p* Adams filtration on $\pi_n M_p^{\wedge}$, which is detected by the \mathbb{Z}/p -Adams spectral sequence. This has almost the same signature as the $\mathbb{Z}/2$ -Adams spectral sequence we use in this paper, except that $\mathbb{Z}/2$ is replaced with \mathbb{Z}/p and the Steenrod algebra is over \mathbb{Z}/p instead of $\mathbb{Z}/2$. Because the mod *p* Thom isomorphism requires an orientation, the story is a little more nuanced for tangential structures which do not induce an orientation.

For free summands in $\pi_n M$, there is no analogous story. The invertible field theories in question are not topological, and at present their classification is still a conjecture [Fre19, Lecture 9]. Assuming this conjecture, though, the Adams filtration does not tell the whole story. For example, consider 3d invertible spin field theories, (conjecturally) classified by

$$[MTSpin, \Sigma^4 I_{\mathbb{Z}}] \xrightarrow{\cong} \operatorname{Hom}(\Omega_4^{\operatorname{Spin}}, \mathbb{Z}) \cong \mathbb{Z},$$

generated by the map φ sending a spin 4-manifold to its signature divided by 16 [Roh52]. As the signature does not depend on the spin structure, 16φ generates $\text{Hom}(\Omega_4^{SO}, \mathbb{Z})$,¹⁴ and therefore the image of the forgetful map $[MTSO, \Sigma^4 I_{\mathbb{Z}}] \rightarrow [MTSpin, \Sigma^4 I_{\mathbb{Z}}]$ is identified with the subgroup 16 \mathbb{Z} . That is, assuming the conjecture on the classification of not-necessarily-topological invertible field theories, a 3d spin invertible field theory only depends on the underlying orientation iff it is q times a generator, where 16 | q. So for free summands in the abelian group of isomorphism classes of invertible field theories, the Adams filtration approach does not work, and one must use other methods.

3.4. A few utility lemmas.

Definition 3.20. Let A be an abelian group, X be a connected space, and $\alpha \in H^1(X; \mathbb{Z}/2)$. Then A_α denotes the local system on X given by the $\mathbb{Z}[\pi_1(X)]$ -module with underlying abelian group A and in which $g \in \pi_1(X)$ acts on A by $(-1)^{\alpha(g)}$, where we interpret α as a map $\pi_1(X) \to \mathbb{Z}/2$ under the identification $H^1(X; \mathbb{Z}/2) \cong \operatorname{Hom}(\pi_1(X), \mathbb{Z}/2)$.

Usually α will be the first Stiefel-Whitney class of a vector bundle, as in the following lemma.

Proposition 3.21. Let $\sigma \to B\mathbb{Z}/2$ denote the tautological line bundle.

- (1) $H^k(B\mathbb{Z}/2;\mathbb{Z}_{w_1(\sigma)})$ is isomorphic to $\mathbb{Z}/2$ in odd degrees and 0 in even degrees.
- (2) If n is odd, $H^k(B\mathbb{Z}/2; (\mathbb{Z}/n)_{w_1(\sigma)}) \cong 0$ for all k.
- (3) If n is even, $H^k(B\mathbb{Z}/2; (\mathbb{Z}/n)_{w_1(\sigma)}) \cong \mathbb{Z}/2$ for all k.

Proof. Use $\mathbb{RP}^{\infty} := \varinjlim_n \mathbb{RP}^n$ as our model for $B\mathbb{Z}/2$. Let A be any abelian group. Given k, choose a very large even m; then, the map $\mathbb{RP}^m \hookrightarrow B\mathbb{Z}/2$ induces an isomorphism $H^k(\mathbb{RP}^m; A_{w_1(\sigma)}) \stackrel{\cong}{\to} H^k(B\mathbb{Z}/2; A_{w_1(\sigma)})$. Since m is even, \mathbb{RP}^m is unorientable, and $\mathbb{Z}_{w_1(\sigma)}$ is isomorphic to the orientation local system for \mathbb{RP}^m , so there is a Poincaré duality isomorphism $H^k(\mathbb{RP}^m; A_{w_1(\sigma)}) \cong H_{m-k}(\mathbb{RP}^m; A)$.

We will repeatedly use the following theorem to show some differentials and extensions are trivial in the Adams spectral sequence.

¹⁴This follows from the fact that the signature defines an isomorphism $\sigma : \Omega_4^{SO} \to \mathbb{Z}$, which follows from the fact that \mathbb{CP}^2 , with signature 1, generates Ω_4^{SO} [Tho54, Remarque following Corollaire IV.18].

Theorem 3.22 (Margolis [Mar74]). Let \mathcal{B} be a sub-Hopf algebra of Steenrod algebra and Y be a spectrum with $\widetilde{H}^*(Y; \mathbb{Z}/2) \cong \mathcal{A} \otimes_{\mathcal{B}} \mathbb{Z}/2$ (so that the change-of-rings trick works for computing 2-completed Y-homology). For any spectrum X, there is a splitting

$$(3.23) Y \wedge X \simeq F \vee \overline{X},$$

where F is an Eilenberg-Mac Lane spectrum for a graded $\mathbb{Z}/2$ -vector space and $\widetilde{H}^*(\overline{X};\mathbb{Z}/2)$ has no free summands as an \mathcal{A} -module.

The upshot is that in the Adams spectral sequence for computing $\pi_*(Y \wedge X)_2^{\wedge}$, the piece of the E_2 -page coming from free summands of $\widetilde{H}^*(X;\mathbb{Z}/2)$ as a \mathcal{B} -module do not emit or receive nontrivial differentials, and do not participate in nontrivial extensions.

Lemma 3.24. Let G be a finite group and $E \rightarrow BG$ be a rank-zero virtual vector bundle.

- (1) If $4 \mid n$, $\widetilde{ko}_n(BG^E) \otimes \mathbb{Q} \cong H_0(BG; \mathbb{Q}_{w_1(E)})$; if $4 \nmid n$, $\widetilde{ko}_n(BG^E)$ is torsion.
- (2) The same is true for $\widetilde{ku}_n(BG^E)$, except divisibility by 4 is replaced by divisibility by 2.

Proof. Atiyah-Hirzebruch [AH61] proved that the Chern character defines an equivalence

The Thom isomorphism theorem establishes that $\widetilde{H}_*(BG^E; \mathbb{Q}) \cong H_*(BG; \mathbb{Q}_{w_1(E)})$, and since G is finite, this vanishes above degree zero by Maschke's theorem.

The proof for ko-theory is the same, except first using the complexification map $c: ko \rightarrow ku$:

$$(3.26) ch \circ c \colon ko \wedge H\mathbb{Q} \xrightarrow{\simeq} \bigvee_{k \ge 0} \Sigma^{4k} H\mathbb{Q}.$$

Choosing E to be the trivial bundle shows the conclusions also hold for the torsion in $\widetilde{ko}_*(BG)$ and $\widetilde{ku}_*(BG)$.

Lemma 3.27 (Adem-Milgram). Fix a prime p, and let H be a subgroup of a finite group G with [G : H] coprime to p and P be a Sylow p-subgroup of H. Assume P is abelian and that $N_H(P)/P = N_G(P)/P$; then the restriction map $\rho_{H,G}: H^*(BG; \mathbb{Z}/p) \to H^*(BH; \mathbb{Z}/p)$ is an isomorphism.

Proof. This is a slight strengthening of theorems of Swan [Swa60] and Adem-Milgram [AM04, Theorems II.6.6 and II.6.8], who prove that if K is a finite group with abelian p-Sylow subgroup P, then the restriction map $H^*(BK; \mathbb{Z}/p) \to H^*(BP; \mathbb{Z}/p)^{N_K(P)}$ is an isomorphism. In our setting, the data of P and N(P)/P are identical for G and H, so both restriction maps $r_{P,G} \colon H^*(BG; \mathbb{Z}/p) \to H^*(BP; \mathbb{Z}/p)^N$ and $r_{P,H} \colon H^*(BH; \mathbb{Z}/p) \to H^*(BP; \mathbb{Z}/p)^N$ are isomorphisms. Since $r_{P,G} = r_{P,H} \circ \rho_{G,H}$, we are done.

Lemma 3.28 (Bock-to-Sq¹ lemma). Let $\beta: H^k(\neg; \mathbb{Z}/2) \to H^{k+1}(\neg; \mathbb{Z})$ denote the integral Bockstein. Then $\beta(x) \mod 2 = \operatorname{Sq}^1(x)$.

Proof. The commutative diagram of short exact sequences



induces a commutative diagram of their induced long exact sequences in cohomology; in particular, $\beta \mod 2$ equals the Bockstein for the bottom short exact sequence, which is Sq¹.

In the mixed unoriented case, Theorems 2.11 and 2.24 ask us to study Thom spectra for determinants of representations. We use the following lemma to simplify them.

Lemma 3.30. Let $\lambda: G \to O_d$ be a faithful representation whose image contains a reflection and $V_{\lambda} \to BG$ be the associated vector bundle. Then the splitting of the surjection

$$(3.31) G \xrightarrow{\lambda} O_d \xrightarrow{\pi_0} \mathbb{Z}/2$$

lifts to a splitting of the Thom spectrum $(BG)^{\text{Det}(V_{\lambda})-1}$ as

$$(3.32) (BG)^{\operatorname{Det}(V_{\lambda})-1} \xrightarrow{\simeq} (B\mathbb{Z}/2)^{\sigma-1} \lor M,$$

and the inclusion $\widetilde{H}^*(M; \mathbb{Z}/2) \hookrightarrow \widetilde{H}^*((BG)^{\operatorname{Det}(V_{\lambda})-1}; \mathbb{Z}/2)$ is injective with image a complementary vector space to the subspace spanned by $\{Uw_1(V_{\lambda})^k \mid k \geq 0\}$.

Proof. Let $g \in G$ be an element sent to λ by a reflection. Then $g^2 = 1$, so the maps $\langle g \rangle \hookrightarrow G \twoheadrightarrow \mathbb{Z}/2$ compose to an isomorphism. Upon taking Thom spectra, these can be identified with maps $(B\mathbb{Z}/2)^{\sigma-1} \to (BG)^{\text{Det}(V_{\lambda})-1} \to (B\mathbb{Z}/2)^{\sigma-1}$ composing to (a map homotopy equivalent to) the identity, which splits off $(B\mathbb{Z}/2)^{\sigma-1}$. The image of the map $\widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1};\mathbb{Z}/2) \to \widetilde{H}^*((BG)^{\text{Det}(V_{\lambda})-1};\mathbb{Z}/2)$ is spanned by $\{Uw_1(V_{\lambda})^k \mid k \geq 0\}$, and the image of $\widetilde{H}^*(M;\mathbb{Z}/2)$ is a complementary subspace.

4. Examples: rotations and reflections

4.1. Warmup: reflections. The simplest example of the fermionic crystalline equivalence principle occurs when the spatial symmetry is $\mathbb{Z}/2$ acting by a reflection. This symmetry can mix with $\mu_2 \subset \text{Spin}_d$, and there are two cases. The following principle is well-established in physics literature; see Shiozaki-Shapourian-Ryu [SSR17b] and Song-Huang-Fu-Hermele [SHFH17, §VII].

- If $\mathbb{Z}/2$ and μ_2 do not mix (often written that the reflection squares to 1), then the classification matches the classification of pin⁺ invertible field theories.
- Conversely, if Z/2 and μ₂ do mix (often written that the reflection squares to (−1)^F), the classification matches that of pin[−] invertible field theories.

Condensed-matter theorists also study theories with time-reversal symmetry. Though this is also an antiunitary symmetry that can mix with μ_2 , the classification in terms of pin structures is opposite that of reflections: when time-reversal symmetry does not mix with fermion parity, we get pin⁻, and when it does mix, we get pin⁺. This is also well-established in physics, and is discussed by Kapustin-Thorngren-Turzillo-Wang [KTTW15], Freed-Hopkins [FH16a], and others.

The difference between these two correspondences is a first hint that the fermionic crystalline equivalence principle must be more complicated than the bosonic version; this point is raised by Thorngren-Else [TE18, §V.A] and Cheng-Wang [CW18, §II.C].

d	Class D, spinless	Class D, spin- $1/2$	Class A
	\$4.1.1	\$4.1.2	\$4.1.3
1	$\mathbb{Z}/2$	$\mathbb{Z}/8$	$\mathbb{Z}/4$
2	$\mathbb{Z}/2$	0	0
3	$\mathbb{Z}/16$	0	$\mathbb{Z}/8 \oplus \mathbb{Z}/2$
4	0	0	0

TABLE 1. $\mathbb{Z}/2$ -equivariant phase homology groups for the cases in which $\mathbb{Z}/2$ acts by a reflection. As discussed in §4.1, these arise as the homotopy groups of the Anderson duals of $MTPin^+$, $MTPin^-$, and $MTPin^c$. For this group action, the spinless and spin-1/2 classifications in class A coincide.

4.1.1. Class D, spinless. When the reflection does not mix with the internal symmetry group, our ansatz is exactly that of Freed-Hopkins. In this setting, $\mathbb{Z}/2$ acts on \mathbb{R}^d as $(d-1) + \sigma$, where k denotes the rank-k trivial representation and σ denotes the sign representation. Let f_0^D denote the equivariant local system of symmetry types for the class D spinless case. Arguing as in [FH19a, (3.6)], in space dimension d we see that

(4.1)
$$Ph_0^{\mathbb{Z}/2}(\mathbb{R}^d; f_0^D) \cong [MTSpin \wedge (B\mathbb{Z}/2)^{1-\sigma}, \Sigma^{d+2}I_{\mathbb{Z}}].$$

Using (2.10a), $MTSpin \wedge (B\mathbb{Z}/2)^{1-\sigma} \simeq MTPin^+$, identifying these phase homology groups as homotopy groups of the Anderson dual of $MTPin^+$, as expected. Finally, to obtain the specific groups in Table 1, we use the preexisting calculations of pin⁺ bordism from [Gia73b, KT90a, KT90b].

4.1.2. Class D, spin-1/2. Again $\mathbb{Z}/2$ acts by $d-1+\sigma$, and this time, reflection mixes with fermion parity. Let $f_{1/2}^D$ denote the equivariant local system of symmetry types for this case. The associated bundle to the $\mathbb{Z}/2$ -representation given by reflection is not pin⁻, so by Theorem 2.11,

(4.2)
$$Ph_0^{\mathbb{Z}/2}(\mathbb{R}^d; f_{1/2}^D) \cong [MTSpin \land (B\mathbb{Z}/2)^{\operatorname{Det}(\sigma)-1}, \Sigma^{d+2}I_{\mathbb{Z}}].$$

Because σ is a line bundle, $\text{Det}(\sigma) = \sigma$. Using (2.10b), $MTSpin \wedge (B\mathbb{Z}/2)^{\sigma-1} \simeq MTPin^{-}$, so these phase homology groups are identified with homotopy groups of the Anderson dual of $MTPin^{-}$ as predicted. These bordism groups are calculated in [ABP69, KT90b].

4.1.3. Class A. For spin^c phases (those of Altland-Zirnbauer class A), the spinless and spin-1/2 classifications coincide: V_{λ} is pin^c, so Theorem 2.24 tells us to consider $MTSpin^c \wedge (B\mathbb{Z}/2)^{1-\sigma}$ in both cases, and by (2.10c), this spectrum is equivalent to $MTPin^c$.

Bahri-Gilkey [BG87a, BG87b] compute pin^c bordism groups,¹⁵ giving us the phase homology groups in Table 1.

4.1.4. *Comparison with prior work.* Reflection-equivariant fermionic phases have been studied by many teams of researchers with many methods. Their results agree with each other, and with us.

- Class D, spinless: These phases, especially the Z/16 in d = 3, are studied by Song-Huang-Fu-Hermele [SHFH17, §V.A], Hsieh-Cho-Ryu [HCR16, §IV], Shiozaki-Shapourian-Ryu [SSR17b, §II.B, §II.D], Guo-Ohmori-Putrov-Wan-Wang [GOP⁺20, §10.7], and Mao-Wang [MW20].
- Class D, spin-1/2: Song-Huang-Fu-Hermele [SHFH17, §V.B], Shapourian-Shiozaki-Ryu [SSR17a, SSR17b], Guo-Ohmori-Putrov-Wan-Wang [GOP+20, §10.7], and Bultinck-Williamson-Haegeman-Verstraete [BWHV17, §IX].
- Class A: These phases have been studied by Isobe-Fu [IF15], Hong-Fu [HF17], Shapourian-Shiozaki-Ryu [SSR17a, SSR17b], Song-Huang-Fu-Hermele [SHFH17, §4], and Shiozaki-Shapourian-Gomi-Ryu [SSGR18, §V].

4.2. Inversions. Inversion symmetry is the $\mathbb{Z}/2$ -symmetry on \mathbb{R}^d acting by $(x_1, \ldots, x_d) \mapsto (-x_1, \ldots, -x_d)$. This offers another relatively simple example of the FCEP, but with a new feature in the spin-1/2 case: the classes in $H^2(B\mathbb{Z}/2; \mathbb{Z}/2)$ specified by the extension $1 \to \mathbb{Z}/2 \to \tilde{G} \to \mathbb{Z}/2 \to 1$ and by $w_2(\lambda) + w_1(\lambda)^2$ are not always equal. This does not change very much, as we explain in §4.2.2 below.

d	Class D, spinless	Class D, spin- $1/2$	Class A
	§4.2.1	§4.2.2	§4.2.3
1	$\mathbb{Z}/2$	$\mathbb{Z}/8$	$\mathbb{Z}/4$
2	\mathbb{Z}	$\mathbb{Z}\oplus\mathbb{Z}/8$	$\mathbb{Z}^2\oplus\mathbb{Z}/4$
3	0	$\mathbb{Z}/16$	$\mathbb{Z}/8\oplus\mathbb{Z}/2$
4	0	$\mathbb{Z}\oplus\mathbb{Z}/16$	$\mathbb{Z}^2\oplus\mathbb{Z}/8\oplus\mathbb{Z}/2$

TABLE 2. $\mathbb{Z}/2$ -equivariant phase homology groups for the cases where $\mathbb{Z}/2$ acts as inversion. The symmetry type whose Thom spectrum determines these groups depends on d; see the referenced sections for which symmetry types appear.

¹⁵In low degrees, Beaudry-Campbell [BC18, §5.6] compute low-degree pin^c bordism groups using the Adams spectral sequence over $\mathcal{A}(1)$, using that $MTPin^c \simeq MTSpin \wedge \Sigma^{-2}MU_1 \wedge \Sigma^{-1}MO_1$. One can also compute using the Adams spectral sequence over $\mathcal{E}(1)$, as in §4.4.3; we found this to be a fun and useful exercise for getting comfortable with this variation of the Adams spectral sequence.

4.2.1. Class D. spinless case. First, the case for which inversion symmetry and fermion parity do not mix. The $\mathbb{Z}/2$ -action on \mathbb{R}^d is a direct sum of d copies of the sign representation σ , so as a $\mathbb{Z}/2$ -space, \mathbb{R}^d is denoted $d\sigma$. This case is covered by Freed-Hopkins [FH19a, Example 3.5], and the phase homology groups are

$$[MTSpin \land (B\mathbb{Z}/2)^{d-d\sigma}, \Sigma^{d+2}I_{\mathbb{Z}}]$$

The spectra $MTSpin \wedge (B\mathbb{Z}/2)^{d-d\sigma}$ are periodic in d.

Lemma 4.4. If d' - d is divisible by 4, $MTSpin \wedge (B\mathbb{Z}/2)^{d(1-\sigma)} \simeq MTSpin \wedge (B\mathbb{Z}/2)^{d'(1-\sigma)}$.

Proof. This is an instance of Theorem 1.39, using that spin structures satisfy the 2-out-of-3 property and that, since 4σ is spin, so is $(d'-d)(1-\sigma)$. \square

Thus we have only to determine $MTSpin \wedge (B\mathbb{Z}/2)^{d(1-\sigma)}$ for small d.

- When d = 0, we get $MTSpin \wedge (B\mathbb{Z}/2)_+$.
- When d = 1, (2.10a) tells us MTSpin ∧ (BZ/2)^{1-σ} ≃ MTPin⁺.
 For d = 2, we have MTSpin ∧ (BZ/2)^{2-2σ}.¹⁶
- When d = -1, (2.10b) gives $MTSpin \wedge \Sigma^{-1}(B\mathbb{Z}/2)^{\sigma-1} \simeq MTPin^{-}$.

The low-degree homotopy groups of these spectra that we need are computed by Giambalvo [Gia73b] and Kirby-Taylor [KT90a, KT90b] (the pin⁺ case); Anderson-Brown-Peterson [ABP69] and Kirby-Taylor [KT90b] (the pin⁻ case); Giambalvo [Gia73a] (the case d = 2); and Mahowald-Milgram [MM76] (the spin $\times \mathbb{Z}/2$ case). Thus we obtain the phase homology groups for the spinless class D case in Table 2.

4.2.2. Class D, spin-1/2 case. Now we consider the case where the inversion symmetry and $\mu_2 \subset \operatorname{Pin}_d^-$ mix as specified by the nontrivial extension $1 \to \mu_2 \to \mathbb{Z}/4 \to \mathbb{Z}/2 \to 1$. This is not classified by $w_2 + w_1^2$ of the associated bundle to the spatial representation: in the language of §2, $\lambda \not\cong \xi$. Instead, this extension is classified by $w_2(\sigma) + w_1(\sigma)^2$, and σ is not pin⁻, so if $f_{1/2}^D$ denotes the class D spin-1/2 equivariant local system of symmetry types on \mathbb{R}^d , Theorem 2.11 computes $Ph_*^{\mathbb{Z}/2}(\mathbb{R}^d; f_{1/2}^D)$ using the Thom spectrum of the virtual bundle

(4.5)
$$-V \boxplus (\sigma + \sigma - d\sigma) \cong -V \boxplus (d-2)(1-\sigma).$$

Thus

(4.6)
$$Ph_0^{\mathbb{Z}/2}(\mathbb{R}^d; f_{1/2}^D) \cong [MTSpin \wedge (B\mathbb{Z}/2)^{(d-2)(1-\sigma)}, \Sigma^{d+2}I_{\mathbb{Z}}]$$

and Lemma 4.4 says the domain is again 4-periodic, but differently from the spinless case.

- When d = 0, we have $MTSpin \wedge (B\mathbb{Z}/2)^{2-2\sigma}$.
- When d = 1, we have $MTSpin \wedge (B\mathbb{Z}/2)^{\sigma-1} \simeq MTPin^{-}$.
- When d = 2, we have $MTSpin \wedge (B\mathbb{Z}/2)_+$.
- When d = -1, we have $MTSpin \wedge (B\mathbb{Z}/2)^{1-\sigma} \simeq MTPin^+$.

In the degrees we need, these bordism groups are computed in the same references we gave above in \$4.2.1, and the relevant phase homology groups appear in Table 2.

Remark 4.7. This fourfold periodicity in the tangential structure appears in a few other contexts in mathematical physics, such as recent work of Hason, Komargodski, and Thorngren [HKT20, §4.4] and Córdova, Ohmori, Shao, and Yan [COSY20] applying it to the study of anomalies of domain wall theories as well as work of Tachikawa and Yonekura [TY19, §3] studying anomalies arising in string theory.

4.2.3. Class A. In class A, whether with spinless or spin-1/2 fermions, the FCEP predicts by way of Theorem 2.24 that an inversion symmetry in dimension d leads us to study $MTSpin^c \wedge (B\mathbb{Z}/2)^{d-d\sigma}$. For any vector bundle $V \to X$, $V \oplus V \cong V \otimes \underline{\mathbb{C}}$, and complex vector bundles are spin^c, so by Theorem 1.39, we can remove factors of $2 - 2\sigma$ from $d - d\sigma$ without changing the Thom spectrum, so we want to study $MTSpin^c \wedge (B\mathbb{Z}/2)_+$ when d is even and $MTSpin^c \wedge (B\mathbb{Z}/2)^{1-\sigma} \simeq MTPin^c$ when d is odd.

We discussed pin^c bordism in §4.1.3. Bahri-Gilkey [BG87a, BG87b] also compute $\Omega^{\text{Spin}^c}_*(B\mathbb{Z}/2)$: they establish that the *Smith homomorphism* $\widetilde{\Omega}_n^{\text{Spin}^c}(B\mathbb{Z}/2) \to \Omega_{n-1}^{\text{Pin}^c}$, which sends a spin^c manifold M and principal

¹⁶Campbell [Cam17, §7.8] shows this spectrum is equivalent to $MT(\text{Spin} \times_{\mathbb{Z}/2} \mathbb{Z}/4)$. Bordism for this symmetry type, called spin- $\mathbb{Z}/4$ bordism or spin^{c/2} bordism, is used in several places in recent mathematical physics literature, including [Cam17, Hsi18, FH19a, GEM19, TY19, DL20a, GOP+20, HKT20, WW20a, Wan20, MV21].

 $\mathbb{Z}/2$ -bundle $P \to M$ to the induced pin^c structure on a smooth submanifold representative of the Poincaré dual of $w_1(P) \in H^1(M; \mathbb{Z}/2)$, is an isomorphism for all n; thus we get the groups in Table 2 by applying the universal property (1.7) of $I_{\mathbb{Z}}$ to either $\Omega_*^{\operatorname{Pin}^c}$ or $\Omega_*^{\operatorname{Spin}^c} \oplus \Omega_{*-1}^{\operatorname{Pin}^c}$, depending on dimension.

4.2.4. *Comparison with prior work.* Inversion-symmetric SPT phases are pretty well-studied, even in the fermionic case, and our phase homology calculations reproduce classifications of inversion-symmetric phases in the literature.

- Class D, spinless: These phases are studied by Shiozaki-Xiong-Gomi [SXG18, §V.B] and Cheng-Wang [CW18, §III].
- Class D, spin-1/2: These phases are studied by You-Xu [YX14, §III], Shiozaki-Shapourian-Ryu [SSR17a, SSR17b], Cheng-Wang [CW18, §III], and Shiozaki-Xiong-Gomi [SXG18, §V.A].
- Class A: These phases are studied by You-Xu [YX14, §IV.A.3], Shiozaki-Shapourian-Ryu [SSR17b, §V.B], and Song-Huang-Fu-Hermele [SHFH17, §IV]. Shiozaki-Shapourian-Ryu also study the phases corresponding to the $\mathbb{Z}/2^{k+2}$ summand in $[MTPin^c, \Sigma^{2k+3}I_{\mathbb{Z}}]$ in arbitrary odd dimensions.¹⁷

Remark 4.8. Guo-Ohmori-Putrov-Wan-Wang $[GOP^+20, \$10.8]$ also study inversion-symmetric fermionic phases from a bordism-theoretic perspective, in both the spinless and spin-1/2 cases. Their results disagree with ours, and with the rest of the literature, because they use different symmetry types to model inversion-equivariant fermionic phases.

4.3. Rotations. We turn to the case of phases equivariant for the cyclic group C_n acting by rotation on a plane. These phases have been studied by several groups of authors, and our results are consistent with prior work; see §4.3.4 for more information.

Let $\lambda: C_n \to SO_2$ denote this representation and $V_{\lambda} \to BC_n$ be the associated vector bundle. One can directly check that $C_n \to SO_2$ lifts across $Spin_2 \to SO_2$ iff n is odd.

d	n	Class D, spinless §4.3.1	Class D, spin-1/2 §4.3.2	Class A §4.3.3
2	$\begin{array}{c} 0 \ { m mod} \ 4 \\ 2 \ { m mod} \ 4 \\ 1, 3 \ { m mod} \ 4 \end{array}$	$egin{array}{l} \mathbb{Z}\oplus\mathbb{Z}/(n/2)\ \mathbb{Z}\oplus\mathbb{Z}/(n/2)\ \mathbb{Z}\oplus\mathbb{Z}/n \end{array}$	$egin{array}{lll} \mathbb{Z}\oplus\mathbb{Z}/2n\oplus\mathbb{Z}/2\ \mathbb{Z}\oplus\mathbb{Z}/4n\ \mathbb{Z}\oplus\mathbb{Z}/n \end{array}$	$ \begin{array}{c} \mathbb{Z}^2 \oplus \mathbb{Z}/2n \oplus \mathbb{Z}/(n/2) \\ \mathbb{Z}^2 \oplus \mathbb{Z}/2n \oplus \mathbb{Z}/(n/2) \\ \mathbb{Z}^2 \oplus \mathbb{Z}/n \oplus \mathbb{Z}/n \end{array} $
3	$\begin{array}{c} 0 \ { m mod} \ 4 \\ 2 \ { m mod} \ 4 \\ 1, 3 \ { m mod} \ 4 \end{array}$	0 0 0	0 0 0	0 0 0

TABLE 3. C_n -equivariant phase homology groups for the cases in which C_n acts by rotations. Classification of fermionic phases with a C_n rotation symmetry. For the spinless class D case, these are classified by $[MTSpin \wedge (BC_n)^{2-V_{\lambda}}, \Sigma^{d+1}I_{\mathbb{Z}}]$; for spin-1/2 class D, by $[MTSpin \wedge (BC_n)_+, \Sigma^{d+1}I_{\mathbb{Z}}]$; and for class A, both spinless and spin-1/2, by $[MTSpin^c \wedge (BC_n)_+, \Sigma^{d+1}I_{\mathbb{Z}}]$.

4.3.1. Class D, spinless case. In this case, C_n does not mix with $\mu_2 \subset$ Spin, and Theorem 2.11 reduces Ansatz 1.22 to the computation of $[MTSpin \wedge (BC_n)^{2-V_{\lambda}}, \Sigma^{d+2}I_{\mathbb{Z}}]$ if n is even, or $[MTSpin \wedge (BC_n)_+]$, if n is odd.

Lemma 4.9. $\Omega_3^{SO}(BC_n) \cong \mathbb{Z}/n$, $\Omega_4^{SO}(BC_n) \cong \mathbb{Z}$, and $\Omega_5^{SO}(BC_n)$ is torsion.

Proof. Compute with the Atiyah-Hirzebruch spectral sequence for oriented bordism; it collapses for $p+q \leq 4$, and the 5-line of the E^2 -page is torsion, implying $\Omega_5^{SO}(BC_n)$ is torsion.

Corollary 4.10 (Bruner-Greenlees [BG10, Example 7.3.2, §12.2.D], García-Etxebarria and Montero [GEM19, §C.2]). For *n* odd, $\Omega_3^{\text{Spin}}(BC_n) \cong \mathbb{Z}/n$, $\Omega_4^{\text{Spin}}(BC_n) \cong \mathbb{Z}$, and $\Omega_5^{\text{Spin}}(BC_n)$ is torsion.

¹⁷The presence of this summand follows from the existence of a $\mathbb{Z}/2^{k+2}$ summand in $\Omega_{2k+2}^{\text{Pin}^c}$, which is proven by Bahri-Gilkey [BG87b].

Proof. Because n is odd, BC_n is stably trivial at 2, and $MTSpin \to MTSO$ is an equivalence away from 2.

Theorem 4.11. If n is even, $\widetilde{\Omega}_{3}^{\text{Spin}}((BC_n)^{2-V_{\lambda}}) \cong \mathbb{Z}/(n/2)$, $\widetilde{\Omega}_{4}^{\text{Spin}}((BC_n))^{2-V_{\lambda}}) \cong \mathbb{Z}$, and $\widetilde{\Omega}_{5}^{\text{Spin}}((BC_n)^{2-V_{\lambda}})$ is torsion.

Proof. The computation breaks into 2-primary and odd-primary pieces. The forgetful map $\Omega^{\text{Spin}}_* \to \Omega^{\text{SO}}_*$ is an odd-primary isomorphism, and because $2 - V_{\lambda}$ is orientable, there is a Thom isomorphism $\widetilde{\Omega}^{\text{SO}}_*((BC_n)^{2-V_{\lambda}}) \cong \Omega^{\text{SO}}_*(BC_n)$. Thus, Lemma 4.9 takes care of the odd-primary part.

Write $n = 2^{\ell}m$, where *m* is odd. Then the map $BC_{2^{\ell}} \to BC_n$ is a stable 2-primary equivalence, because it induces an isomorphism on mod 2 cohomology, so for the 2-primary piece it suffices to understand the case $n = 2^{\ell}$. Campbell [Cam17, Theorem 1.8] studies $\Omega_d^{\text{Spin}}((BC_{2^{\ell}})^{2-V_{\lambda}})$, obtaining $\mathbb{Z}/2^{\ell-1}$ when d = 3, \mathbb{Z} when d = 4, and torsion when d = 5, which suffices.¹⁸

4.3.2. Class D, spin-1/2 case. Theorem 2.11 asks us to compute $[MTSpin \wedge (BC_n)_+, \Sigma^{d+2}I_{\mathbb{Z}}]$, which (1.7) tells us in terms of $\Omega^{\text{Spin}}_*(BC_n)$. For n odd, we already saw this in Corollary 4.10.

Proposition 4.12. Let $n \equiv 2 \mod 4$. Then $\Omega_3^{\text{Spin}}(BC_n) \cong \mathbb{Z}/4n$, $\Omega_4^{\text{Spin}}(BC_n) \cong \mathbb{Z}$, and $\Omega_5^{\text{Spin}}(BC_n)$ is torsion.

Proof. Inclusion $BC_2 \to BC_n$ is a 2-local equivalence, so the fact that the 2-torsion is $\mathbb{Z}/8$ in degree 3 and vanishes in degree 4 follows as soon as we know that for $\Omega^{\text{Spin}}_*(BC_2)$. This was originally done by Mahowald-Milgram [MM76] but has been computed in a few other places, including Mahowald [Mah82, Lemma 7.3], Bruner-Greenlees [BG10, Example 7.3.1], Siegemeyer [Sie13, Theorem 2.1.5], and García-Etxebarria and Montero [GEM19, (C.18)]. What remains is odd-primary information, which is equivalent to the odd-primary part of oriented bordism, which we computed in Lemma 4.9.

Proposition 4.13. For $n \equiv 0 \mod 4$, $\Omega_3^{\text{Spin}}(BC_n) \cong \mathbb{Z}/2 \oplus \mathbb{Z}/2n$, $\Omega_4^{\text{Spin}}(BC_n) \cong \mathbb{Z}$, and $\Omega_5^{\text{Spin}}(BC_n)$ is torsion.

Proof. Write $n = 2^{\ell}m$, where *m* is odd. As in the proof of Theorem 4.11, the 2-primary part of the answer is detected by $BC_{2^{\ell}} \to BC_n$, and the odd-primary part of the answer is detected oriented bordism. Davighi-Lohitsiri [DL20a, §A.3] compute $\Omega_k^{\text{Spin}}(BC_{2^{\ell}})$ for $k \leq 6$, giving the 2-primary summand, and for the odd-primary part we use Lemma 4.9.

Botvinnik-Gilkey-Stolz [BGS97, Theorem 2.4], Bruner-Greenlees [BG10, Example 7.3.3], and Siegemeyer [Sie13, §2.2] do special cases of this computation, by a variety of methods.

4.3.3. Class A. The representation of C_n on \mathbb{R}^2 by rotations is unitary (under the standard identification $\mathbb{R}^2 = \mathbb{C}$), hence spin^c, so in both the spinless and spin-1/2 cases, we consider $MTSpin^c \wedge (BC_n)_+$: in the spinless case, we have a Thom isomorphism $MTSpin^c \wedge (BC_n)^{2-V_{\lambda}} \xrightarrow{\simeq} MTSpin^c \wedge (BC_n)_+$, and in the spin-1/2 case, $Det(V_{\lambda})$ is trivial, so Theorem 2.24 also gives us $MTSpin^c \wedge (BC_n)_+$.

Theorem 4.14. The first few spin^c bordism groups of BC_n are

$\Omega_0^{\mathrm{Spin}^c}(BC_{2k}) \cong \mathbb{Z}$	$\Omega_0^{\operatorname{Spin}^c}(BC_{2k+1}) \cong \mathbb{Z}$
$\Omega_1^{\mathrm{Spin}^c}(BC_{2k}) \cong \mathbb{Z}/2k$	$\Omega_1^{\operatorname{Spin}^c}(BC_{2k+1}) \cong \mathbb{Z}/(2k+1)$
$\Omega_2^{\mathrm{Spin}^c}(BC_{2k}) \cong \mathbb{Z}$	$\Omega_2^{\operatorname{Spin}^c}(BC_{2k+1}) \cong \mathbb{Z}$
$\Omega_3^{\mathrm{Spin}^c}(BC_{2k}) \cong \mathbb{Z}/4k \oplus \mathbb{Z}/k$	$\Omega_3^{\operatorname{Spin}^c}(BC_{2k+1}) \cong (\mathbb{Z}/(2k+1))^{\oplus 2}$
$\Omega_4^{\mathrm{Spin}^c}(BC_{2k}) \cong \mathbb{Z}^2$	$\Omega_4^{\operatorname{Spin}^c}(BC_{2k+1}) \cong \mathbb{Z}^2,$

and $\Omega_5^{\mathrm{Spin}^c}(BC_n)$ is torsion for all n.

¹⁸There are a few other computations of $\widetilde{\Omega}_*^{\text{Spin}}((BC_{2\ell})^{2-V_{\lambda}})$ in low degrees by other methods. For $\ell = 1$, see Giambalvo [Gia73b], García-Etxebarria and Montero [GEM19, (C.21)], and Freed-Hopkins [FH19a, §5]. For $\ell > 1$, see Botvinnik-Gilkey [BG97, §5] and Davighi-Lohitsiri [DL20a, §A.4]; Botvinnik-Gilkey only report the orders of the bordism groups, but their computations show that the groups we need are cyclic. Be aware that Campbell and Davighi-Lohitsiri consider a different vector bundle than $2 - V_{\lambda}$, though their calculations apply to this case.

Proof. Write $n = 2^{\ell} \cdot m$, where m is odd. It suffices to compute the 2-primary piece and $\Omega_*^{\text{Spin}^c}(BC_n) \otimes \mathbb{Z}[1/2]$. The inclusion $C_{2^{\ell}} \to C_n$ is stably a 2-primary equivalence, so for the 2-primary piece it suffices to determine $\Omega_*^{\text{Spin}^c}(BC_{2^{\ell}})$. Bahri-Gilkey [BG87b, Theorem 1] compute these groups; when $\ell = 0$ they are $\Omega_*^{\text{Spin}^c}(\text{pt})$, which begins \mathbb{Z} , 0, \mathbb{Z} , 0, \mathbb{Z}^2 , 0; and when $\ell \neq 0$ we have the same free summands as when $\ell = 0$, but additional torsion summands: $\Omega_1^{\text{Spin}^c}(BC_{2^{\ell}}) \cong \mathbb{Z}/2^{\ell}$, and $\Omega_3^{\text{Spin}^c}(BC_{2^{\ell}}) \cong \mathbb{Z}/2^{\ell-1} \oplus \mathbb{Z}/2^{\ell+1}$. After smashing with $H\mathbb{Z}[1/2]$, the forgetful map $MTSpin^c \to MTSO \land (BU_1)_+$ is an equivalence, so

After smashing with $H\mathbb{Z}[1/2]$, the forgetful map $MTSpin^c \to MTSO \wedge (BU_1)_+$ is an equivalence, so $MTSO \wedge (BU_1)_+$ detects all odd-primary torsion in spin^c bordism. To compute this, we use the Atiyah-Hirzebruch spectral sequence

(4.15)
$$E_{p,q}^2 = H_p(BU_1 \times BC_n; \Omega_q^{SO}(\text{pt})) \Longrightarrow \Omega_{p+q}^{SO}(BU_1 \times BC_n).$$

The Künneth theorem implies the first few homology groups of $BU_1 \times BC_n$ are $H_0 = \mathbb{Z}$, $H_1 = \mathbb{Z}/n$, $H_2 = \mathbb{Z}$, $H_3 = (\mathbb{Z}/n)^{\oplus 2}$, $H_4 = \mathbb{Z}$, and $H_5 = (\mathbb{Z}/n)^{\oplus 3}$. When we feed this to the spectral sequence (4.15), there are no nonzero differentials to or from any element in total degree p + q < 5: because $\Omega_i^{SO} = 0$ for i = 1, 2, 3, the only possible nonzero differential would be a $d_4 \colon E_{5,0}^2 \to E_{0,4}^2$, but the splitting $\Omega_*^{SO}(BU_1 \times BC_n) = \Omega_*^{SO}(pt) \oplus \widetilde{\Omega}_*^{SO}(BU_1 \times BC_n)$ splits off the q = 0 line splits off from the rest of the spectral sequence, killing this d_4 . This tells the odd-primary torsion in degrees 0 through 4, and since the 5-line of the E_2 -page is torsion, $\Omega_5^{SO}(BU_1 \times BC_n)$ is also torsion.

4.3.4. Comparison with prior work. Rotation-equivariant phases in class D have been studied by several groups, including Shiozaki-Shapourian-Ryu [SSR17b, §IV.C], Guo-Ohmori-Putrov-Wan-Wang [GOP+20, §10.9], and Freed-Hopkins [FH19a, §5], who all restrict to the case n = 2, and most comprehensively by Cheng-Wang [CW18, §IV, §V], who consider arbitrary n and both the spinless and spin-1/2 cases in d = 2, 3. Freed-Hopkins begin from the same ansatz as us so agreement is no surprise. In the remaining cases, there is almost complete agreement: all classifications compute the same torsion summands, but they all miss the free summand in d = 2. This is not a discrepancy, however: many authors restrict to considering phases whose low-energy effective theories are expected to be topological field theories, which in the ansatz of Freed-Hopkins [FH16a, §§5.3–5.4] amounts to considering the torsion subgroup of the classification using $I_{\mathbb{Z}}MTH$. The non-topological theories corresponding to the free summand have been discussed in a few references, including Freed [Fre19, Lecture 9] and Wan-Wang [WW20a, §7.1]; at present, their mathematical description remains partly conjectural.

Rotation-equivariant phases in class A are studied by Shiozaki-Shapourian-Ryu [SSR17b, §IV.D], Shiozaki-Xiong-Gomi [SXG18, §V.C.1], and Lu-Vishwanath-Khalaf [LVK19]. Shiozaki-Shapourian-Ryu and Lu-Vishwanath-Khalaf's classifications agree with us on torsion but miss the free summand as before, and Shiozaki-Xiong-Gomi's computation completely matches ours. Again, the free summand corresponds to non-topological invertible field theories.

4.4. Rotations and reflections. In this section, we compute the phase homology groups corresponding to phases on \mathbb{R}^d equivariant for the D_{2n} -action of rotations and reflections in a given plane. Zhang-Wang-Yang-Qi-Gu [ZWY⁺20] also study these phases for d = 2 and in class D; we compare our results to theirs in §4.4.5.

Let λ be the standard real 2-dimensional representation of D_{2n} and $V_{\lambda} \to BD_{2n}$ be the associated vector bundle. Let s be a reflection in D_{2n} and r a rotation through the angle $2\pi/n$. Then, define $x, y \in H^1(BD_{2n}; \mathbb{Z}/2) = \text{Hom}(D_{2n}, \mathbb{Z}/2)$ by

(4.16a)
$$x(s^{\ell}r^m) \coloneqq \ell \mod 2$$

$$(4.16b) y(s^{\ell}r^m) \coloneqq m \mod 2$$

In the representation λ , $s^{\ell}r^m \in D_{2n}$ acts by an orientation-reversing endomorphism iff ℓ is odd, so $w_1(V_{\lambda}) = x$.

Proposition 4.17 ([Sna13, Theorem 4.6], [Tei92, §2.3], [Han93, Theorems 5.5 and 5.6]).

- (1) If *n* is odd, $H^*(BD_{2n}; \mathbb{Z}/2) \cong \mathbb{Z}/2[x]$.
- (2) If $n \equiv 0 \mod 4$, $H^*(BD_{2n}; \mathbb{Z}/2) \cong \mathbb{Z}/2[x, y, w]/(xy + y^2)$, where |w| = 2 and $w = w_2(V_\lambda)$.
- (3) If $n \equiv 2 \mod 4$, $H^*(BD_{2n}; \mathbb{Z}/2) \cong \mathbb{Z}/2[x, y]$.

Lemma 4.18. For $n \equiv 2 \mod 4$, $w_2(V_{\lambda}) = xy + y^2$.

d	n	Class D, spinless §4.4.1	Class D, spin-1/2 §4.4.2	Class A, spinless §4.4.3	Class A, spin-1/2 §4.4.4
2	$0 \bmod 4$	$(\mathbb{Z}/2)^{\oplus 2}$	$(\mathbb{Z}/2)^{\oplus 2}$	$\mathbb{Z}/2n$	$\mathbb{Z}/(n/2) \oplus (\mathbb{Z}/2)^{\oplus 2}$
	$2 \mod 4$	$\mathbb{Z}/2$	$(\mathbb{Z}/2)^{\oplus 2}$	$\mathbb{Z}/2n$	$\mathbb{Z}/n\oplus\mathbb{Z}/2$
	$1,3 \bmod 4$	$\mathbb{Z}/2$	0	\mathbb{Z}/n	\mathbb{Z}/n
3	$0 \bmod 4$	$(\mathbb{Z}/2)^{\oplus 4}$	0	$(\mathbb{Z}/2)^{\oplus 4}$	$\mathbb{Z}/8 \oplus \mathbb{Z}/4 \oplus \mathbb{Z}/2$
	$2 \mod 4$	$(\mathbb{Z}/2)^{\oplus 3}$	0	$(\mathbb{Z}/2)^{\oplus 4}$	$\mathbb{Z}/8 \oplus \mathbb{Z}/4 \oplus \mathbb{Z}/2$
	$1,3 \mod 4$	$\mathbb{Z}/16$	0	$\mathbb{Z}/8\oplus\mathbb{Z}/2$	$\mathbb{Z}/8\oplus\mathbb{Z}/2$

TABLE 4. D_{2n} -equivariant phase homology groups, where D_{2n} acts through rotations and reflections. These arise as homotopy groups of Anderson duals of $MTSpin \wedge X_n$ and $MTSpin^c \wedge X_n$, where X_n is one of $(BD_{2n})^{2-V_{\lambda}}$ or $(BD_{2n})^{\text{Det}(V_{\lambda})-1}$. See §4.4 for details and proofs.

Proof. Since $s, r^{n/2} \in D_{2n}$ commute, there is a map $j: \mathbb{Z}/2 \times \mathbb{Z}/2 \to D_{2n}$ sending $(1,0) \mapsto s$ and $(0,1) \mapsto r^{n/2}$. The pullback map $j^*: H^*(BD_{2n}; \mathbb{Z}/2) \to H^*(B\mathbb{Z}/2 \times B\mathbb{Z}/2; \mathbb{Z}/2)$ sends x and y to linearly independent elements of $H^1(B\mathbb{Z}/2 \times B\mathbb{Z}/2; \mathbb{Z}/2)$: one way to see this is to identify the pullback map with the map $\operatorname{Hom}(D_{2n}, \mathbb{Z}/2) \to \operatorname{Hom}(\mathbb{Z}/2 \times \mathbb{Z}/2, \mathbb{Z}/2)$ given by precomposing with j. Thus j^* is an isomorphism on $H^1(-; \mathbb{Z}/2)$. For both BD_{2n} and $B\mathbb{Z}/2 \times B\mathbb{Z}/2$, the mod 2 cohomology ring is the free symmetric algebra on $H^1(-; \mathbb{Z}/2)$, so j^* is an isomorphism of cohomology rings.

Thus we can compute $w_2(V_{\lambda})$ by regarding λ as a $\mathbb{Z}/2 \times \mathbb{Z}/2$ representation. Let $\ell_1 \subset \lambda$ be the fixed locus of s, which is a subspace, and ℓ_2 be its orthogonal complement. Then $\lambda = \ell_1 \oplus \ell_2$ as $(\mathbb{Z}/2 \times \mathbb{Z}/2)$ -representations. Both s and $r^{n/2}$ act nontrivially on ℓ_2 ; on ℓ_1 , s acts trivially and $r^{n/2}$ acts nontrivially. Thus $w(\ell_1) = 1 + j^*(y)$, $w(\ell_2) = 1 + j^*(x) + j^*(y)$, and

(4.19)
$$w_2(j^*V_\lambda) = w_2(\ell_1) + w_1(\ell_1)w_1(\ell_2) + w_2(\ell_2) = j^*(y(x+y)).$$

Lemma 4.20. Suppose n is odd and $i: \mathbb{Z}/2 \hookrightarrow D_{2n}$ is the inclusion of $\langle s \rangle$. Let $V \to BD_{2n}$ be a virtual vector bundle such that $w_1(V)$, as an element $\operatorname{Hom}(D_{2n}, \mathbb{Z}/2)$, is nonzero on s. Then, the induced map of Thom spectra $\hat{i}: (B\mathbb{Z}/2)^{i^*V} \to (BD_{2n})^V$ is a 2-primary homotopy equivalence.

Proof. By the homology Whitehead theorem, it suffices to show $\hat{\imath}$ induces an isomorphism on mod 2 cohomology. The Thom isomorphism rewords our question to be about the map $H^*(BD_{2n}; \mathbb{Z}/2) \to H^*(B\mathbb{Z}/2; \mathbb{Z}/2)$, and Proposition 4.17 tells us that both $H^*(B\mathbb{Z}/2; \mathbb{Z}/2)$ and $H^*(BD_{2n}; \mathbb{Z}/2)$ are abstractly isomorphic to $\mathbb{Z}/2[x]$ with |x| = 1; we will show $i^*x_{BD_{2n}} = x_{B\mathbb{Z}/2}$, implying i^* is a ring isomorphism. Since x is the only nonzero degree-one element and V and i^*V are both unorientable, $x = w_1(V)$ and $i^*x = w_1(i^*V) \neq 0$.

We will need the next calculations to determine the odd-primary torsion subgroups of the phase homology groups we calculate. Recall that $x \in H^1(BD_{2n}; \mathbb{Z}/2)$ is equal to $w_1(V_\lambda)$.

Lemma 4.21 (Handel [Han93, Theorems 5.8, 5.9]).

(4.22)
$$H_*(BD_{2n}; (\mathbb{Z}[1/2])_x) \cong \begin{cases} \mathbb{Z}/n, & n \equiv 1 \mod 4\\ 0, & \text{otherwise.} \end{cases}$$

Handel calculates $H^*(BD_{2n};\mathbb{Z}_x)$; use the universal coefficient theorem to switch to $\mathbb{Z}[1/2]$ -homology.

Proposition 4.23. Suppose $V \to BD_{2n}$ is a rank-zero virtual vector bundle with $w_1(V) = x$. Then the odd-torsion subgroup of $\widetilde{\Omega}_k^{\text{Spin}}((BD_{2n})^V)$ is isomorphic to the odd-torsion subgroup of \mathbb{Z}/n for k = 1, and vanishes for k = 0, 2, 3, and 4.

Proof. Apply the Atiyah-Hirzebruch spectral sequence for the completion of spin bordism at primes other than 2. Since $w_1(V) = x$, the Thom isomorphism identifies $\widetilde{H}_*((BD_{2n})^V) \cong H_*(BD_{2n}; \mathbb{Z}_x)$, and by Lemma 4.21 we know these groups away from 2. The only nonzero entry in the E^2 -page of total degree less than 5 is $E_{1,0}^2 \cong \mathbb{Z}/n$, so the spectral sequence collapses in the desired range and we conclude.

Proposition 4.24. With V as in Proposition 4.23, the odd-torsion subgroup of $\widetilde{\Omega}_k^{\text{Spin}^c}((BD_{2n})^V)$ is isomorphic to the odd-torsion subgroup of \mathbb{Z}/n for k = 1 and 3, and vanishes for k = 0, 2, and 4.

Proof. Use the Atiyah-Hirzebruch spectral sequence for the completion of $MTSpin^c$ at odd primes, just as for Proposition 4.23.

4.4.1. Class D, spinless case. Since we are considering spinless fermions, the FCEP tells us to compute $[MTSpin \wedge (BD_{2n})^{2-V_{\lambda}}, \Sigma^{d+1}I_{\mathbb{Z}}].$

Proposition 4.25. For n odd, the first few spin bordism groups of X_n are

 $\widetilde{\Omega}_0^{\text{Spin}}(X_n) \cong \mathbb{Z}/2$ $\widetilde{\Omega}_1^{\text{Spin}}(X_n) \cong \mathbb{Z}/n$ $\widetilde{\Omega}_2^{\text{Spin}}(X_n) \cong \mathbb{Z}/2$ $\widetilde{\Omega}_3^{\text{Spin}}(X_n) \cong \mathbb{Z}/2$ $\widetilde{\Omega}_4^{\text{Spin}}(X_n) \cong \mathbb{Z}/16,$

and $\widetilde{\Omega}_5^{\text{Spin}}(X_n)$ is torsion.

Proof. To compute the 2-torsion subgroups of these bordism groups, apply Lemma 4.20 with $2 - V_{\lambda}$ to get a 2-primary stable equivalence $(BD_{2n})^{2-V_{\lambda}} \simeq (B\mathbb{Z}/2)^{1-\sigma}$, then (2.10a) to get $MTSpin \wedge (B\mathbb{Z}/2)^{1-\sigma} \simeq MTPin^+$. Low-degree pin⁺ bordism groups are calculated in [Gia73b, KT90a, KT90b]. For the odd-torsion subgroups, use Proposition 4.23.

Now we turn to the case where $n \equiv 2 \mod 4$.

Theorem 4.26. When $n \equiv 2 \mod 4$, the first few spin bordism groups of X_n are

$$\begin{split} \widetilde{\Omega}_0^{\mathrm{Spin}}(X_n) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_1^{\mathrm{Spin}}(X_n) &\cong \mathbb{Z}/n\\ \widetilde{\Omega}_2^{\mathrm{Spin}}(X_n) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_3^{\mathrm{Spin}}(X_n) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_4^{\mathrm{Spin}}(X_n) &\cong (\mathbb{Z}/2)^{\oplus 3}, \end{split}$$

and $\widetilde{\Omega}_5^{\text{Spin}}(X_n)$ is torsion.

As usual, this together with the universal property (1.7) of $I_{\mathbb{Z}}$ gives the $n \equiv 2 \mod 4$ entries in Table 4.

Proof. We will use the Adams spectral sequence at the prime 2 to compute $\widetilde{\Omega}_d^{\mathrm{Spin}}(X_n)$ for $d \leq 7$. This only sees 2-primary information, but we already calculated the odd-torsion subgroup in Proposition 4.23. Recall that $w_1(V_\lambda) = x$ and (from Lemma 4.18) $w_2(V_\lambda) = xy + y^2$; thus $w_1(2 - V_\lambda) = x$ and $w_2(2 - V_\lambda) = x^2 + xy + y^2$. This tells us the Steenrod squares in $\widetilde{H}^*(X_n; \mathbb{Z}/2)$, e.g. $\mathrm{Sq}^1(U) = Ux$ and $\mathrm{Sq}^2(U) = U(x^2 + xy + y^2)$. Continuing in this vein determines the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(X_n; \mathbb{Z}/2)$ in low degrees, as shown in Figure 1, left. We obtain a splitting as $\mathcal{A}(1)$ -modules:

(4.27)
$$H^*(X_n; \mathbb{Z}/2) \cong \mathcal{A}(1) \oplus \Sigma R_0 \oplus \Sigma^2 \mathcal{A}(1) \oplus \Sigma^4 \mathcal{A}(1) \oplus \Sigma^4 \mathcal{A}(1) \oplus P.$$

The $\mathcal{A}(1)$ -module \mathbb{R}_0 is defined to be $\widetilde{H}^*((\mathbb{B}\mathbb{Z}/2)^{1-\sigma};\mathbb{Z}/2)$; the copy appearing here is the indecomposable summand containing Uy. The submodule P contains no elements of degree below 6, so is irrelevant for our low-degree computations; we need to determine $\operatorname{Ext}(M)$ for the remaining summands. For $\Sigma^k \mathcal{A}(1)$, there is a single $\mathbb{Z}/2$ summand in topological degree k and filtration 0, and for $\Sigma \mathbb{R}_0$, see [GMM68, §2] or [BC18, Figure 24]. Putting these together, we display the E_2 -page of this Adams spectral sequence in Figure 1, right. In this range, a combination of h_1 -equivariance and Margolis' theorem (Theorem 3.22) forces all differentials to vanish, and Margolis' theorem implies there are no hidden extensions, so we are done.

Finally, consider the case that $n \equiv 0 \mod 4$.



FIGURE 1. Left: the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*((BD_{2n})^{2-V_{\lambda}}; \mathbb{Z}/2)$ in low degrees, when $n \equiv 2 \mod 4$. Here $\alpha \coloneqq x^4y + y^5$. The submodule pictured here contains all elements of degree at most 5. Right: the E_2 -page of the corresponding Adams spectral sequence computing *ko*-theory.

Theorem 4.28. Let $n \equiv 0 \mod 4$.

$$\begin{split} \widetilde{\Omega}_{0}^{\mathrm{Spin}}(X_{n}) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_{1}^{\mathrm{Spin}}(X_{n}) &\cong \mathbb{Z}/n\\ \widetilde{\Omega}_{2}^{\mathrm{Spin}}(X_{n}) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_{3}^{\mathrm{Spin}}(X_{n}) &\cong (\mathbb{Z}/2)^{\oplus 2}\\ \widetilde{\Omega}_{4}^{\mathrm{Spin}}(X_{n}) &\cong (\mathbb{Z}/2)^{\oplus 4}, \end{split}$$

and $\widetilde{\Omega}_5^{\text{Spin}}(X_n)$ is torsion.

Proof of Theorem 4.28. First, by Proposition 4.23, the only odd-primary torsion in $\widetilde{\Omega}_k^{\text{Spin}}(X_n)$ for $k \leq 4$ is in degree 1. Draw the Atiyah-Hirzebruch spectral sequence

(4.29)
$$E_{p,q}^2 = \widetilde{H}_p(X_n; \Omega_q^{\text{Spin}}) \Longrightarrow \widetilde{\Omega}_{p+q}^{\text{Spin}}(X_n).$$

After applying the Thom isomorphism, this needs as input $H_*(BD_{2n}; \mathbb{Z}_x)$ and $H_*(BD_{2n}; \mathbb{Z}/2)$. The former can be determined using Handel's calculation [Han93, Theorem 5.8] of $H^*(BD_{2n}; \mathbb{Z}_x)$, and the latter can be determined from Proposition 4.17; in both cases use the universal coefficient theorem to pass from homology to cohomology. We obtain $E_{1,0}^2 \cong \mathbb{Z}/n$ and $E_{0,1}^2 \cong \mathbb{Z}/2$, so there are three options for $\widetilde{\Omega}_1^{\text{Spin}}(X_n)$: \mathbb{Z}/n , $\mathbb{Z}/n \oplus \mathbb{Z}/2$, or $\mathbb{Z}/2n$. We will address this ambiguity later.

Using Proposition 4.17, $w_1(2 - V_\lambda) = x$ and $w_2(2 - V_\lambda) = w + x^2$. Hence $\operatorname{Sq}^1(U) = Ux$ and $\operatorname{Sq}^2(U) = U(w + x^2)$. We also need the Steenrod squares of x, y, and w. For degree reasons, $\operatorname{Sq}(x) = x + x^2$ and $\operatorname{Sq}(y) = y + y^2$.

Lemma 4.30 ([Mal11, §4.1]). $Sq(w) = w + wx + w^2$.

These and the Cartan formula determine the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(X_n; \mathbb{Z}/2)$. In Figure 2, left, we display this structure in low degrees.

In particular,

(4.31)
$$\widetilde{H}^*(X_n;\mathbb{Z}/2) \cong \mathcal{A}(1) \oplus \Sigma R_2 \oplus \Sigma^2 \mathbb{Z}/2 \oplus \Sigma^4 J \oplus \Sigma^4 \mathcal{A}(1) \oplus \Sigma^4 \mathcal{A}(1) \oplus \Sigma^5 \mathcal{O} \oplus P,$$

where P is 5-connected, and we define \mathbb{R}_2 , J, and \hat{O} as follows. First, \mathbb{R}_2 is defined to be the kernel of the augmentation map $\mathcal{A}(1) \to \mathbb{Z}/2$; the indecomposable summand in (4.31) isomorphic to $\Sigma \mathbb{R}_2$ is generated by Uy and Uy^2 . The Joker is the $\mathcal{A}(1)$ -module $J := \mathcal{A}(1)/(\mathrm{Sq}^3)$; here it is generated by Uw^2 . Finally,



FIGURE 2. Left: the low-degree mod 2 cohomology of $(BD_{2n})^{2-V_{\lambda}}$ over $\mathcal{A}(1)$, $n \equiv 0 \mod 4$. This summand contains all elements in degrees 5 and below. The dashed line indicates that the $\mathbb{Z}/2^r$ Bockstein maps Uy to Uw, which we need in Lemma 4.32. Right: the E_2 -page of the Adams spectral sequence computing $\tilde{ko}_*((BD_{2n})^{2-V_{\lambda}})_2^{\wedge}$. See Lemma 4.32 for how to address the differential in topological degree 2 and Lemma 4.35 to show the differential in topological degree 5 vanishes.

 $\hat{\mathcal{O}} := \mathcal{A}(1)/(\mathrm{Sq}^1, \mathrm{Sq}^2\mathrm{Sq}^3)$ and is called the *upside-down question mark*; here it is generated by Uw^2y . For each of these summands M in (4.31), $\mathrm{Ext}_{\mathcal{A}(1)}^{s,t}(M, \mathbb{Z}/2)$ is known in the degrees relevant to us – except for P, which is too high-degree to affect our calculations anyways.

- For $\Sigma^k \mathcal{A}(1)$ there is a single $\mathbb{Z}/2$ in bidegree s = 0, t = k.
- For R_2 , J, and ∂ , see [BC18, Figure 29].¹⁹
- For $\mathbb{Z}/2$, see [BC18, Figure 20].

Put these together to obtain the E_2 -page as in Figure 2, right. Lemma 3.24 tells us the E_{∞} -page is torsion, so there must be nonzero differentials in the range shown, though not necessarily the d_2 s pictured.

The first nonzero differential is a d_r from the 2-line to the 1-line; by h_0 -equivariance, it kills the entire yellow tower in the 2-line. Since a d_r differential decreases t - s by 1 and increases s by r, on the E_{r+1} -page, the 2-line contains only the first r summands of the orange tower, and the 3-line contains only the orange $\mathbb{Z}/2$ summand in degree s = 0. There can be no further differentials to or from the 1- or 2-lines, so we obtain $\mathbb{Z}/2^r$ in degree 1 and $\mathbb{Z}/2$ in degree 2.

Lemma 4.32. 2^r is the largest power of 2 dividing n, i.e. $\widetilde{\Omega}_1^{\text{Spin}}(X_n) \cong \mathbb{Z}/n$.

Proof. The May-Milgram theorem [MM81] identifies Adams spectral sequence differentials between towers with Bockstein spectral sequence differentials. What it means here is that the lemma statement is equivalent to the statement that the Bockstein map $\beta: H^1(-;\mathbb{Z}/2^r) \to H^2(-;\mathbb{Z}/2)$ associated to the short exact sequence

$$(4.33) 0 \longrightarrow \mathbb{Z}/2 \longrightarrow \mathbb{Z}/2^{r+1} \longrightarrow \mathbb{Z}/2^r \longrightarrow 0$$

carries a preimage of Uy to Uw. Both of these classes are in the image of the pullback map induced by $(B\mathbb{Z}/n)^{2-V} \to (BD_{2n})^{2-V}$, and the Bockstein is natural with respect to the Thom isomorphism, so we just have to check this in the cohomology of $B\mathbb{Z}/n$, where it is true [Cam17, DL20a].

The next differential that might be nonzero, and which is the only possibly nonzero differential to or from an element of degree 3 or 4, is $d_2: E_2^{0,5} \to E_2^{2,6}$. If this $d_2 = 0$, there is also an extension problem in degree t - s = 4 of the form

(4.34)
$$0 \longrightarrow \mathbb{Z}/2 \longrightarrow \widetilde{\Omega}_4^{\mathrm{Spin}}(X_n) \longrightarrow \mathbb{Z}/2 \oplus \mathbb{Z}/2 \oplus \mathbb{Z}/2 \longrightarrow 0.$$

¹⁹The first calculations of $\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(\mathbb{R}_2,\mathbb{Z}/2)$ and $\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(J,\mathbb{Z}/2)$ that we know of are due to Adams-Priddy [AP76, §3].

Lemma 4.35. This d_2 vanishes, and the extension (4.34) splits.

Proof. We will prove this by mapping to a simpler Adams spectral sequence that has already been studied, as depicted in Figure 3.

Because V_{λ} is the pullback of the tautological bundle $V_2 \to BO_2$ along $B\lambda \colon BD_{2n} \to BO_2$, we obtain a map of Thom spectra $f \colon X_n = (BD_{2n})^{2-V_{\lambda}} \to (BO_2)^{2-V_2}$; the codomain is often denoted $\Sigma^2 MTO_2$. Under f, our $Uw \in \widetilde{H}^2(X_n; \mathbb{Z}/2)$ is the pullback of $Uw_2 \in \widetilde{H}^2(\Sigma^2 MTO_2)$.

The spin bordism of $\Sigma^2 MTO_2$ is identified with the bordism theory of the group $\operatorname{Pin}^{\tilde{c}+} := (\operatorname{Pin}^+ \ltimes \operatorname{Spin}_2)/\mu_2$. Invertible field theories for this tangential structure are believed to correspond to invertible topological phases of Altland-Zirnbauer type AII [FH16a, (9.25), (10.2)].²⁰

Several authors study the Adams spectral sequence for $\Omega_*^{\text{Pin}^{\tilde{c}+}} \cong \widetilde{\Omega}_*^{\text{Spin}}(\Sigma^2 M TO_2)$ in low degrees, including Freed-Hopkins [FH16a, Figure 5, case s = -2], Campbell [Cam17, Example 6.10], and Wan-Wang-Zheng [WWZ20, §6.2.3]. Their work shows $Uw_2 \in \widetilde{H}^2(\Sigma^2 M TO_2; \mathbb{Z}/2)$ generates a $\Sigma^2 \mathbb{Z}/2$ summand as an $\mathcal{A}(1)$ -submodule of $\widetilde{H}^*(\Sigma^2 M TO_2; \mathbb{Z}/2)$, and therefore f^* restricts to an isomorphism from that $\Sigma^2 \mathbb{Z}/2$ summand to our $\Sigma^2 \mathbb{Z}/2$ summand generated by Uw. This means the submodule of the E_2 -page for $\widetilde{\Omega}_*^{\text{Spin}}(X_n)$ coming from $\Sigma^2 \mathbb{Z}/2$ maps isomorphically onto the submodule of the E_2 -page for $\widetilde{\Omega}_*^{\text{Spin}}(\Sigma^2 M TO_2)$ coming from the $\Sigma^2 \mathbb{Z}/2$ generated by Uw_2 — and crucially, in that spectral sequence, $E_2^{0.5} \cong 0$. See the commutative diagram of pink arrows in Figure 3. Thus the image of our d_2 under f vanishes, and the map between these spectral sequences on $E_2^{2,6}$'s (the targets of these d_2 s) is an isomorphism, so our d_2 also vanishes.



FIGURE 3. The map $X_n \to \Sigma^2 MTO_2$ induces a map between the Adams spectral sequences computing their *ko*-theory groups. We use this in Lemma 4.35 to show the pictured d_2 vanishes, as the square of pink arrows in the above figure is commutative. The right-hand side of this figure, which displays $\text{Ext}(\tilde{H}^*(\Sigma^2 MTO_2; \mathbb{Z}/2))$, is adapted from Campbell [Cam17, Figure 6.9].

Now suppose (4.34) does not split; then, there are elements $x, y \in \widetilde{\Omega}_4^{\text{Spin}}(X_n)$ such that x = 2y and the image of x image in the E_{∞} -page of the Adams spectral sequence is the nonzero element of $E_{\infty}^{2,6} \cong \mathbb{Z}/2$. Then f maps this $\mathbb{Z}/2$ isomorphically onto a $\mathbb{Z}/2$ in the E_{∞} -page for $\Sigma^2 M TO_2$, so $f_*(x) \neq 0$. But $\Omega_4^{\text{Pin}^{\tilde{c}+}} \cong (\mathbb{Z}/2)^{\oplus 3}$ [FH16a, Theorem 9.87], so no matter where y maps to, $2y = x \mapsto 0$, which is a problem.

We have thus determined $\widetilde{\Omega}_d^{\text{Spin}}(X_n)_2^{\wedge}$ for d = 3, 4, so we are done.

4.4.2. Class D, spin-1/2 case.

Lemma 4.36. V_{λ} is not pin⁻.

Proof. For n even, this follows by pulling back along $BC_n \to BD_{2n}$: we saw in §4.3 that the pullback is not spin, so V_{λ} cannot be pin⁻. For n odd, pull back along the map $B\mathbb{Z}/2 \to BD_{2n}$ induced by the inclusion of a reflection; the pullback is not pin⁻, so neither is V_{λ} .

Therefore by Theorem 2.11, we consider $X_n := (BD_{2n})^{\text{Det}(V_{\lambda})-1}$.

 $^{^{20}\}mbox{For}$ further discussion, see also Metlitski [Met15] and Seiberg-Witten [SW16, §A.4].

Proposition 4.37. For n odd, the first few spin bordism groups of X_n are

 $\widetilde{\Omega}_{0}^{\text{Spin}}(X_{n}) \cong \mathbb{Z}/2$ $\widetilde{\Omega}_{1}^{\text{Spin}}(X_{n}) \cong \mathbb{Z}/2n$ $\widetilde{\Omega}_{2}^{\text{Spin}}(X_{n}) \cong \mathbb{Z}/8$ $\widetilde{\Omega}_{3}^{\text{Spin}}(X_{n}) \cong 0$ $\widetilde{\Omega}_{4}^{\text{Spin}}(X_{n}) \cong 0,$

and $\widetilde{\Omega}_5^{\text{Spin}}(X_n)$ is torsion.

Proof. To compute the 2-torsion subgroups of these bordism groups, apply Lemma 4.20 with $\text{Det}(V_{\lambda}) - 1$ get a 2-primary stable equivalence $(BD_{2n})^{Det(V_{\lambda})-1} \simeq (B\mathbb{Z}/2)^{\sigma-1}$, then (2.10b) to get $MTSpin \wedge (B\mathbb{Z}/2)^{1-\sigma} \simeq MTPin^-$. Low-degree pin⁻ bordism groups are calculated in [ABP69, KT90b]. For the odd-torsion subgroups, use Proposition 4.23.

Theorem 4.38. When $n \equiv 2 \mod 4$, the first few bordism groups of X_n are

$$\begin{split} \widetilde{\Omega}_0^{\mathrm{Spin}}(X_n) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_1^{\mathrm{Spin}}(X_n) &\cong \mathbb{Z}/n \oplus \mathbb{Z}/2\\ \widetilde{\Omega}_2^{\mathrm{Spin}}(X_n) &\cong \mathbb{Z}/8 \oplus \mathbb{Z}/4\\ \widetilde{\Omega}_3^{\mathrm{Spin}}(X_n) &\cong \mathbb{Z}/2 \oplus \mathbb{Z}/2\\ \widetilde{\Omega}_4^{\mathrm{Spin}}(X_n) &\cong 0, \end{split}$$

and $\widetilde{\Omega}_5^{\text{Spin}}(X_n)$ is torsion.

Proof. We establish a 2-primary equivalence $MTSpin \wedge X_n \simeq MTPin^- \wedge (B\mathbb{Z}/2)_+$, so the free and 2-torsion part of the spin bordism groups of X are isomorphic to the pin⁻ bordism groups of $B\mathbb{Z}/2$. Once we finish this, we use work of Guo-Ohmori-Putrov-Wan-Wang [GOP+20, §7.2.1] computing $\Omega_k^{\text{Pin}^-}(B\mathbb{Z}/2)$ in degrees 5 and below to get the 2-primary part; for the odd-primary torsion, we use Proposition 4.23 as usual.

Lemma 4.39. The inclusion $i: \mathbb{Z}/2 \times \mathbb{Z}/2 \hookrightarrow D_{2n}$ given by a reflection and a half-turn induces a 2-primary equivalence of Thom spectra $(B(\mathbb{Z}/2 \times \mathbb{Z}/2))^{i^* \operatorname{Det}(V_\lambda) - 1} \xrightarrow{\simeq} (BD_{2n})^{\operatorname{Det}(V_\lambda) - 1}$.

Proof. The map $Bi: B(\mathbb{Z}/2 \times \mathbb{Z}/2) \to BD_{2n}$ induces an equivalence on mod 2 cohomology, and therefore by the Thom isomorphism theorem also induces an equivalence on the mod 2 cohomology of the Thom spectra in question. This suffices by the stable Whitehead theorem.

The stable bundle $i^* \text{Det}(V_\lambda) \to B(\mathbb{Z}/2 \times \mathbb{Z}/2)$ splits as an exterior direct sum $\sigma \boxplus \underline{0}$, where $\sigma \to B\mathbb{Z}/2$ is the tautological line bundle. Therefore the Thom spectrum also splits: $(B(\mathbb{Z}/2 \times \mathbb{Z}/2))^{i^* \text{Det}(V_\lambda)-1} \simeq (B\mathbb{Z}/2)^{\sigma-1} \wedge (B\mathbb{Z}/2)_+$. Therefore by (2.10b),

(4.40)
$$MTSpin \wedge (BD_{2n})^{\text{Det}(V_{\lambda})-1} \simeq MTSpin \wedge (B\mathbb{Z}/2)^{\sigma-1} \wedge (B\mathbb{Z}/2)_{+} \simeq MTPin^{-} \wedge (B\mathbb{Z}/2)_{+}.$$

Finally, let $n \equiv 0 \mod 4$. Recall $H^*(BD_{2n}; \mathbb{Z}/2) \cong \mathbb{Z}/2[x, y, w]/(xy + y^2)$ with |x| = |y| = 1 and |w| = 2, so $\operatorname{Sq}(x) = x + x^2$ and $\operatorname{Sq}(y) = y + y^2$, and from Lemma 4.30, $\operatorname{Sq}(w) = w + wx + w^2$. The Stiefel-Whitney classes of $\operatorname{Det}(V_\lambda)$ tell us that if U is the Thom class, $\operatorname{Sq}^1(U) = Ux$ and $\operatorname{Sq}^2(U) = 0$ in the cohomology of X_n .

Theorem 4.41. For $n \equiv 0 \mod 4$, the first few bordism groups of X_n are

$$\Omega_0^{\mathrm{Spin}}(X_n) \cong \mathbb{Z}/2$$
$$\widetilde{\Omega}_1^{\mathrm{Spin}}(X_n) \cong \mathbb{Z}/n \oplus \mathbb{Z}/2$$
$$\widetilde{\Omega}_2^{\mathrm{Spin}}(X_n) \cong \mathbb{Z}/8 \oplus \mathbb{Z}/4$$
$$\widetilde{\Omega}_3^{\mathrm{Spin}}(X_n) \cong \mathbb{Z}/2 \oplus \mathbb{Z}/2$$
$$\widetilde{\Omega}_4^{\mathrm{Spin}}(X_n) \cong 0,$$

and $\widetilde{\Omega}_5^{\text{Spin}}(X_n)$ is torsion.

Proof. First, by Proposition 4.23, the only odd-primary torsion in $\widetilde{\Omega}_k^{\text{Spin}}(X_n)$ for $k \leq 4$ is in degree 1. Draw the Atiyah-Hirzebruch spectral sequence

(4.42)
$$E_{p,q}^2 = \widetilde{H}_p(X_n; \Omega_q^{\text{Spin}}) \Longrightarrow \widetilde{\Omega}_{p+q}^{\text{Spin}}(X).$$

After applying the Thom isomorphism, this needs as input $H_*(BD_{2n}; \mathbb{Z}_x)$ and $H_*(BD_{2n}; \mathbb{Z}/2)$. The former can be determined using Handel's calculation [Han93, Theorem 5.8] of $H^*(BD_{2n}; \mathbb{Z}_x)$, and the latter can be determined from Proposition 4.17; in both cases use the universal coefficient theorem to pass from homology to cohomology. Since $E_{1,0}^2 \cong \mathbb{Z}/n$ and $E_{0,1}^2 \cong \mathbb{Z}/2$, there are three options for $\widetilde{\Omega}_1^{\text{Spin}}(X_n)$: $\mathbb{Z}/n, \mathbb{Z}/n \oplus \mathbb{Z}/2$, or $\mathbb{Z}/2n$. We will learn which one is correct in our analysis of the 2-primary part below.

For the 2-primary part, we use the Adams spectral sequence as usual. By Lemma 3.30, a choice of a reflection in D_{2n} induces a splitting

(4.43)
$$X_n \xrightarrow{\simeq} (B\mathbb{Z}/2)^{\sigma-1} \lor M_n,$$

such that the map $\widetilde{H}^*(M_n; \mathbb{Z}/2) \to \widetilde{H}^*(X_n; \mathbb{Z}/2)$ is injective with image complementary to the subspace spanned by $\{Ux^i \mid i \geq 0\}$. We focus on $MTSpin \wedge M_n$, adding in the summands arising from $MTSpin \wedge (B\mathbb{Z}/2)^{\sigma-1} \simeq MTPin^-$ at the end. The $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(M_n; \mathbb{Z}/2)$ is determined by its image in $\widetilde{H}^*(X_n; \mathbb{Z}/2)$, which we know using Sq¹ and Sq² of x, y, w, and U via the Cartan formula. Using this, we draw this $\mathcal{A}(1)$ -module structure in Figure 4, left.



FIGURE 4. Left: the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(M_n; \mathbb{Z}/2)$ in low degrees. The pictured summand contains all elements in degrees 4 and below. Right: the Ext of this module, which is the E_2 -page of the Adams spectral sequence converging to $\widetilde{ko}_*(M_n)$. See the proof of Theorem 4.41 for more information.

As $\mathcal{A}(1)$ -modules,

(4.44)
$$\widehat{H}^*(M_n;\mathbb{Z}/2) \cong \Sigma R_1 \oplus \Sigma^2 \partial \oplus \Sigma^3 \mathcal{A}(1) \oplus \Sigma^3 \mathcal{A}(1) \oplus P,$$

where P is 4-connected; we will see below that the 4-line is empty, so there are no nonzero differentials from $\operatorname{Ext}(P)$ to anything we care about. Here ΣR_1 is the indecomposable summand containing Uy. For the $\Sigma^k \mathcal{A}(1)$ summands, we know the Ext; for ΣR_1 , see [BC18, Figure 26], and for $\Sigma^2 \partial$, see [BC18, Figure 29]. Assembling these, we display the E_2 -page for $t - s \leq 5$ in Figure 4, right. Lemma 3.24 implies $\widetilde{\Omega}_5^{\operatorname{Spin}}(X_n)$ is torsion, as claimed, and that there must be a differential d_r from the infinite tower in topological degree 2 to the infinite tower in topological degree 1, though it might not be the d_2 pictured.²¹ Margolis' theorem and h_0 -equivariance rule out any other nonzero differentials to or from elements with $t - s \leq 4$. Therefore in this range, $E_{r+1} = E_{\infty}$. The infinite tower in topological degree 2 is killed by the differential, as are all but rof the $\mathbb{Z}/2$ summands of the infinite tower in topological degree 1. The first few 2-completed spin bordism groups of M_n are therefore $\mathbb{Z}/2^r$ in degree 1, $\mathbb{Z}/4$ in degree 2, $\mathbb{Z}/2 \oplus \mathbb{Z}/2$ in degree 3, and 0 in degrees 0 and 4.

²¹In fact, r is the largest number such that $2^r \mid n$. Like in the proof of Lemma 4.32, one can deduce this using the Bockstein from Uy to Uw and the May-Milgram theorem.

Finally, we add in the pin⁻ bordism summands as computed in [ABP69, KT90b]: a $\mathbb{Z}/2$ in degrees 0 and 1, a $\mathbb{Z}/8$ in degree 2, and 0 otherwise. In particular, since the 2-torsion subgroup of $\widetilde{\Omega}_1^{\text{Spin}}(X_n)$ is of the form $\mathbb{Z}/2 \oplus \mathbb{Z}/2^r$, $\widetilde{\Omega}_1^{\text{Spin}}(X_n) \cong \mathbb{Z}/n \oplus \mathbb{Z}/2$.

4.4.3. Class A, spinless case. In this case, Theorem 2.24 asks us to consider $X_n := MTSpin^c \wedge (BD_{2n})^{2-V_{\lambda}}$. **Theorem 4.45.** For n odd, the first few spin bordism groups of X_n are

$$\widetilde{\Omega}_{0}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/2$$

$$\widetilde{\Omega}_{1}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/n$$

$$\widetilde{\Omega}_{2}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/4$$

$$\widetilde{\Omega}_{3}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/n$$

$$\widetilde{\Omega}_{4}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/8 \oplus \mathbb{Z}/2$$

and $\widetilde{\Omega}_5^{\text{Spin}^c}(X_n)$ is torsion.

Proof. To compute the 2-torsion subgroups of these bordism groups, apply Lemma 4.20 with $2-V_{\lambda}$ to get a 2-primary stable equivalence $(BD_{2n})^{2-V_{\lambda}} \simeq (B\mathbb{Z}/2)^{1-\sigma}$, then (2.10c) to get $MTSpin^{c} \wedge (B\mathbb{Z}/2)^{1-\sigma} \simeq MTPin^{c}$. The pin^c bordism groups we need are calculated by Bahri-Gilkey [BG87a, BG87b]. For the odd-torsion subgroups, use Proposition 4.24.

Theorem 4.46. Let $n \equiv 2 \mod 4$; then the low-degree spin^c bordism of X_n is

$$\widetilde{\Omega}_{0}^{\operatorname{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/2$$

$$\widetilde{\Omega}_{1}^{\operatorname{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/n$$

$$\widetilde{\Omega}_{2}^{\operatorname{Spin}^{c}}(X_{n}) \cong (\mathbb{Z}/2)^{\oplus 2}$$

$$\widetilde{\Omega}_{3}^{\operatorname{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/2n$$

$$\widetilde{\Omega}_{4}^{\operatorname{Spin}^{c}}(X_{n}) \cong (\mathbb{Z}/2)^{\oplus 4},$$

and $\widetilde{\Omega}_5^{{\rm Spin}^c}(X_n)$ is torsion.

Proof. First, Proposition 4.24 computes the odd-torsion subgroups: a \mathbb{Z}/n in degrees 1 and 3, and nothing else below degree 5.

To compute the 2-primary information we use the Adams spectral sequence over $\mathcal{E}(1)$, which converges to $\widetilde{ku}_*(X_n)$, together with Anderson-Brown-Peterson's isomorphism $\widetilde{\Omega}_n^{\mathrm{Spin}^c}(X_n) \stackrel{\simeq}{\to} \widetilde{ku}_n(X_n) \oplus \widetilde{ku}_{n-4}(X_n)$ for $n \leq 7$ [ABP67].

The $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(X_n; \mathbb{Z}/2)$ that we calculated in (4.27) and displayed in Figure 1, left, determines the $\mathcal{E}(1)$ -module structure: as $\mathcal{E}(1)$ -modules, $\mathcal{A}(1) \cong \mathcal{E}(1) \oplus \Sigma^2 \mathcal{E}(1)$. Therefore

(4.47)
$$\widetilde{H}^*(X_n;\mathbb{Z}/2)\cong \mathcal{E}(1)\oplus \Sigma R_0\oplus \Sigma^2 \mathcal{E}(1)\oplus \Sigma^2 \mathcal{E}(1)\oplus \Sigma^4 \mathcal{E}(1)\oplus \Sigma^4 \mathcal{E}(1)\oplus \Sigma^4 \mathcal{E}(1)\oplus P_2$$

where P is 5-connected; we draw a picture of this $\mathcal{E}(1)$ -module in Figure 5, left.

Next Ext. For $\Sigma^k \mathcal{E}(1)$, there is a unique $\mathbb{Z}/2$ summand, in degree s = 0, t = k; for ΣR_0 , we must work a little harder.

Proposition 4.48. $\operatorname{Ext}_{\mathcal{E}(1)}^{s,t}(\mathbb{R}_0,\mathbb{Z}/2)$ is given in Figure 6, right.

Proof. Our proof uses as input $\operatorname{Ext}_{\mathcal{E}(1)}(N_1)$, where N_1 is defined to be the \mathcal{A} -module $\Sigma^{-1}\widetilde{H}^*(\mathbb{RP}^2;\mathbb{Z}/2)$, with two $\mathbb{Z}/2$ summands connected by a Sq¹; this in turn defines its $\mathcal{A}(1)$ - and $\mathcal{E}(1)$ -module structures. Davis-Mahowald [DM81, §2] calculate $\operatorname{Ext}_{\mathcal{E}(1)}(N_1)$ as a graded vector space but we also need its $H^{*,*}(\mathcal{E}(1))$ -module structure.

Let $\langle Q_1 \rangle \subset \mathcal{E}(1)$ denote the subalgebra generated by Q_1 , which is a two-dimensional vector space over $\mathbb{Z}/2$. As $\mathcal{E}(1)$ -modules, $N_1 \cong \mathcal{E}(1) \otimes_{\langle Q_1 \rangle} \mathbb{Z}/2$, so by the change-of-rings theorem (3.4), there are isomorphisms of $H^{*,*}(\mathcal{E}(1))$ -modules

(4.49)
$$\operatorname{Ext}_{\mathcal{E}(1)}(N_1) \cong \operatorname{Ext}_{\langle Q_1 \rangle}(\mathbb{Z}/2) \cong \mathbb{Z}/2[v_1],$$


FIGURE 5. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*((BD_{2n})^{2-V_{\lambda}}; \mathbb{Z}/2), n \equiv 2 \mod 4$, in low degrees. The pictured submodule contains all elements in degrees 5 and below. Right: the Adams E_2 -page computing $\widetilde{ku}_*((BD_{2n})^{2-V_{\lambda}})$.

with $v_1 \in \operatorname{Ext}_{\mathcal{E}(1)}^{1,3}(N_1, \mathbb{Z}/2)$. The rightmost isomorphism in (4.49) uses Koszul duality [BC18, Remark 4.5.4], which applies because $\langle Q_1 \rangle$ is an exterior algebra.

Now for R_0 , we use the extension of $\mathcal{E}(1)$ -modules

$$(4.50) 0 \longrightarrow \Sigma^2 R_0 \longrightarrow R_0 \longrightarrow N_1 \longrightarrow 0$$

drawn in Figure 6, left.



FIGURE 6. Left: the extension (4.50). Right: the long exact sequence it induces of Ext groups. See the proof of Proposition 4.48 for why the long exact sequence looks like this; the key feature is that there are no elements in odd topological degree, so all boundary maps vanish. The dashed lines are h_0 -extensions which are not implied by the long exact sequence, but are shown in the proof of Proposition 4.48.

At first, all we know is $\text{Ext}(N_1)$. Because this lives solely in even topological degrees, and $\Sigma^2 R_0$ is 2-connected, the long exact sequence diagram is empty in topological degree 1, so the boundary map

(4.51)
$$\delta \colon \operatorname{Ext}_{\mathcal{E}(1)}^{s,s+1}(\mathbb{R}_0,\mathbb{Z}/2) \to \operatorname{Ext}_{\mathcal{E}(1)}^{s,s}(N_1,\mathbb{Z}/2)$$

vanishes, which tells us the line t-s = 0 in $\operatorname{Ext}(R_0)$ consists of a single $\mathbb{Z}/2$ summand in filtration 0. Therefore the line t-s = 2 in the long exact sequence diagram consists of two $\mathbb{Z}/2$ summands: one in filtration 1 coming from N_1 , and one in filtration 0 coming from $\Sigma^2 R_0$. Since the 1-line of the diagram is empty and $\operatorname{Ext}(N_1)$ is concentrated in even degrees, the 3-line of the diagram is empty, so there are no differentials to the 2-line. Continuing in this way produces Figure 6, right.

Finally, acting by $h_0 \in H^{*,*}(\mathcal{E}(1))$ defines an isomorphism

(4.52)
$$\operatorname{Ext}_{\mathcal{E}(1)}^{0,2}(R_0, \mathbb{Z}/2) \to \operatorname{Ext}_{\mathcal{E}(1)}^{1,3}(R_0, \mathbb{Z}/2).$$

This can be checked directly from the definition: begin with the unique nontrivial map $R_0 \to \Sigma^2 \mathbb{Z}/2$ and act on it by an extension representing h_0 (namely the extension $0 \to \Sigma \mathbb{Z}/2 \to N_1 \to \mathbb{Z}/2 \to 0$); the result is a nontrivial extension.

With $\operatorname{Ext}(\Sigma R_0)$ in hand, we return to our goal of computing $\widetilde{ku}_*(X_n)$. We draw the E_2 -page of the Adams spectral sequence in (5), right. Margolis' theorem (Theorem 3.22) forces all differentials in this range to vanish, except possible differentials with target the 7-line, and there can be no hidden extensions in the range depicted. Thus for n = 2k < 7, $\widetilde{ku}_*(X_n) \cong (\mathbb{Z}/2)^{\oplus k+1}$ and for n = 2k + 1 < 8, $\widetilde{ku}_*(X_n) \cong \mathbb{Z}/2^{k+1}$; we finish with the fact that the map $MTSpin^c \to ku \vee \Sigma^4 ku$ is 7-connected, so we can read off the spin^c bordism groups from the ku-homology groups.

Theorem 4.53. If $n \equiv 0 \mod 4$, write $n = 2^k m$ with m odd. The first few spin^c bordism groups of X_n are

$$\Omega_0^{\operatorname{Spin}^c}(X_n) \cong \mathbb{Z}/2$$

$$\widetilde{\Omega}_1^{\operatorname{Spin}^c}(X_n) \cong \mathbb{Z}/n$$

$$\widetilde{\Omega}_2^{\operatorname{Spin}^c}(X_n) \cong (\mathbb{Z}/2)^{\oplus 2}$$

$$\widetilde{\Omega}_3^{\operatorname{Spin}^c}(X_n) \cong \mathbb{Z}/2n$$

$$\widetilde{\Omega}_4^{\operatorname{Spin}^c}(X_n) \cong (\mathbb{Z}/2)^{\oplus 4}$$

and $\widetilde{\Omega}_5^{{\rm Spin}^c}(X_n)$ is torsion.

Proof of Theorem 4.53. By Proposition 4.24, the odd-primary torsion is isomorphic to the odd-primary torsion of \mathbb{Z}/n in degrees 1 and 3 and vanishes in degrees 0, 2, and 4.

At 2, we use the Adams spectral sequence. We described the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(X; \mathbb{Z}/2)$ in (4.31) and draw it in Figure 2; this determines the $\mathcal{E}(1)$ -module structure, with isomorphisms of $\mathcal{E}(1)$ -modules $\mathcal{A}(1) \cong \mathcal{E}(1) \oplus \Sigma^2 \mathcal{E}(1), R_2 \cong \widetilde{O} \oplus \Sigma \mathcal{E}(1)$ and $J \cong \mathcal{E}(1) \oplus \Sigma^2 \mathbb{Z}/2$. Hence as $\mathcal{E}(1)$ -modules,

 $(4.54) \quad \widetilde{H}^*(X;\mathbb{Z}/2) \cong \mathcal{E}(1) \oplus \Sigma^{\hat{\mathcal{O}}} \oplus \Sigma^2 \mathcal{E}(1) \oplus \Sigma^2 \mathcal{E}(1) \oplus \Sigma^2 \mathbb{Z}/2 \oplus \Sigma^4 \mathcal{E}(1) \oplus \Sigma^4 \mathcal{E}(1) \oplus \Sigma^4 \mathcal{E}(1) \oplus \Sigma^5 \hat{\mathcal{O}} \oplus P,$

where P is 5-connected. We draw this $\mathcal{E}(1)$ -module in Figure 7, left.



FIGURE 7. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*((BD_{2n})^{2-V_{\lambda}}; \mathbb{Z}/2)$, $n \equiv 0 \mod 4$, in low degrees. The pictured submodule contains all elements in degrees 5 and below. The gray dashed line indicates that the $\mathbb{Z}/2^r$ Bockstein maps a preimage of Uy to Uw, which we use in the proof of Theorem 4.53. Right: the E_2 -page for the Adams spectral sequence computing $\widetilde{Ku}_*((BD_{2n})^{2-V_{\lambda}})$. The two pictured differentials are related by a v_1 -action.

We calculated $\text{Ext}(\mathbb{Z}/2)$ in (3.7), and Adams-Priddy [AP76, §3] show

(4.55)
$$\operatorname{Ext}_{\mathcal{E}(1)}^{s,t}(\dot{\mathcal{O}}, \mathbb{Z}/2) \cong \operatorname{Ext}_{\mathcal{E}(1)}^{s+1,t+1}(\mathbb{Z}/2, \mathbb{Z}/2),$$

with the isomorphism intertwining the $H^{*,*}(\mathcal{E}(1))$ -actions. We can therefore draw the E_2 -page of the Adams spectral sequence in Figure 7, right. We hide most v_1 -actions to declutter the diagram.

The first differential that could be nonzero is from the 2-line to the 1-line; as differentials are h_0 -equivariant, if a d_r differential is nonzero on one summand in the tower on the 2-line, then it is nonzero on the entire tower, so we refer to differentials between towers. The May-Milgram theorem [MM81] characterizes differentials between towers: there is a d_r differential between those two towers iff the Bockstein $\beta: H^1(-;\mathbb{Z}/2^r) \to H^2(-;\mathbb{Z}/2)$ carries a preimage of Uy to Uw. The Thom isomorphism is natural with respect to this Bockstein, so it suffices to know whether $\beta(y) = w$ in $H^2(BD_{2n};\mathbb{Z}/2)$, and we saw this in the proof of Lemma 4.32, where ris the largest number such that $2^r \mid n$. This means that the 2-torsion in $\widetilde{\Omega}_1^{\text{Spin}^c}(X_n)$ is isomorphic to that of \mathbb{Z}/n , so along with our odd-torsion computation we see that $\widetilde{\Omega}_1^{\text{Spin}^c}(X_n) \cong \mathbb{Z}/n$.

The other differential we need to resolve in range goes from the tower in the 4-line to the tower in the 3-line. Action by $v_1 \in ku_2$ carries the tower in the 2-line to the tower in the 4-line, and the tower in the 1-line to the tower in the 3-line, and differentials are v_1 -equivariant, so there is also a d_r differential between these towers. As seen in Figure 7, right, on the E_{∞} -page there are $r + 1 \mathbb{Z}/2$ summands on the 3-line, all connected, so together with our odd-torsion computation we see that $\widetilde{\Omega}_3^{\text{Spin}^c}(X_n) \cong \mathbb{Z}/2n$.

There can be no other nonzero differentials in range, and Margolis' theorem precludes any hidden extensions, so we are done. $\hfill \Box$

4.4.4. Class A, spin-1/2 case.

Lemma 4.56. V_{λ} is pin^c iff n is odd.

Proof. For n odd, we saw that inclusion of a reflection defines a map $B\mathbb{Z}/2 \to BD_{2n}$ which is an isomorphism on mod 2 cohomology. Therefore we can compute Stiefel-Whitney classes of V_{λ} by pulling back to $B\mathbb{Z}/2$, and we saw that the pullback bundle is stably equivalent to a line bundle, so $w_2 = 0$.

For *n* even, recall that V_{λ} is pin^{*c*} iff $\beta(w_2(V_{\lambda})) = 0$, where $\beta : H^k(-; \mathbb{Z}/2) \to H^{k+1}(-; \mathbb{Z})$ is the integral Bockstein. Lemma 3.28 means it suffices to show $\operatorname{Sq}^1(w_2(V_{\lambda})) \neq 0$. In the notation of Proposition 4.17, for $n \equiv 2 \mod 4, w_2(V_{\lambda}) = xy + y^2$, and $\operatorname{Sq}^1(xy + y^2) = x^2y + xy^2 \neq 0$. For $n \equiv 0 \mod 4, w_2(V_{\lambda}) = w$, and by Lemma 4.30, $\operatorname{Sq}^1(w) \neq 0$.

Therefore for *n* odd, we consider $X_n \coloneqq (BD_{2n})^{2-V_{\lambda}}$. We computed $\Omega_k^{\text{Spin}^c}(X_n)$ for $k \leq 4$ in Theorem 4.45. For *n* even, Theorem 2.24 directs us to the spin^{*c*} bordism of $X_n \coloneqq (BD_{2n})^{\text{Det}(V_{\lambda})-1}$.

Theorem 4.57. If $n \equiv 2 \mod 4$, the first few spin^c bordism groups of X_n are

$$\Omega_{0}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/2$$

$$\widetilde{\Omega}_{1}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/n$$

$$\widetilde{\Omega}_{2}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/4 \oplus \mathbb{Z}/2$$

$$\widetilde{\Omega}_{3}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/n \oplus \mathbb{Z}/2$$

$$\widetilde{\Omega}_{4}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/8 \oplus \mathbb{Z}/4 \oplus \mathbb{Z}/2.$$

Because Lemma 3.24 implies $\widetilde{\Omega}_5^{\text{Spin}^c}(X_n)$ is torsion, the phase homology groups for this symmetry type are $\mathbb{Z}/n \oplus \mathbb{Z}/2$ for d = 2 and $\mathbb{Z}/8 \oplus \mathbb{Z}/4 \oplus \mathbb{Z}/2$ for d = 3.

The 2-local equivalence $MTSpin \wedge X_n \simeq MTPin^- \wedge (B\mathbb{Z}/2)_+$ we used in Theorem 4.38 implies a 2-local equivalence $MTSpin^c \wedge X_n \simeq MTPin^c \wedge (B\mathbb{Z}/2)_+$, so when n = 2, these are also the pin^c bordism groups of $\mathbb{Z}/2$. This may be of independent interest.

Proof. We can read the odd-primary torsion off of Proposition 4.24. For 2-primary torsion we use the Adams spectral sequence over $\mathcal{E}(1)$ as usual. Recall from the proof of Theorem 4.38 that $(BD_{2n})^{\text{Det}(V_{\lambda})-1} \simeq (B\mathbb{Z}/2)^{\sigma-1} \wedge (B\mathbb{Z}/2)_+$. Guo-Ohmori-Putrov-Wan-Wang [GOP+20, §7.2.1] determine the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1} \wedge B\mathbb{Z}/2; \mathbb{Z}/2)$ in low degrees. Using their work, and the isomorphisms of $\mathcal{E}(1)$ -modules $\mathcal{A}(1) \cong \mathcal{E}(1) \oplus \mathcal{E}(1)$ and $R_5 \cong \mathcal{E}(1) \oplus \Sigma R_0$, there is an isomorphism of $\mathcal{E}(1)$ -modules

(4.58)
$$\widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1} \wedge B\mathbb{Z}/2; \mathbb{Z}/2) \cong \Sigma \mathcal{E}(1) \oplus \Sigma^2 R_0 \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus P,$$

where P is 4-connected. Since we began with $(B\mathbb{Z}/2)^{\sigma-1} \wedge (B\mathbb{Z}/2)_+$, this does not account for everything; the disjoint basepoint gives us another summand equivalent to

(4.59)
$$MTSpin^{c} \wedge (B\mathbb{Z}/2)^{\sigma-1} \simeq MTPin^{c}$$

ARUN DEBRAY

by (2.10c). We will add in the pin^c bordism groups coming from this summand, which can be read off from the work of Bahri-Gilkey [BG87a, BG87b], after running the Adams spectral sequence for the other summand.

Returning to (4.58), we will see momentarily that $E_2^{s,t}$ is empty when t-s = 4 and $s \ge 2$, which precludes differentials from the 5-line to the 4-line and therefore means that P does not affect the calculations we make. In Figure 8, left, we draw (4.58). We computed $\operatorname{Ext}(R_0)$ in Proposition 4.48, so we can draw the E_2 -page of the Adams spectral sequence for $ku_*((B\mathbb{Z}/2)^{\sigma-1} \wedge B\mathbb{Z}/2)$, as in Figure 8, right. In the degrees we care about, this collapses, and we deduce the spin^c bordism of $(B\mathbb{Z}/2)^{\sigma-1} \wedge B\mathbb{Z}/2$ and combine it with pin^c bordism to conclude.



FIGURE 8. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1} \wedge B\mathbb{Z}/2; \mathbb{Z}/2)$ in low degrees. The pictured submodule contains all elements in degrees 4 and below. Right: Ext of this submodule, which is the E_2 -page of the Adams spectral sequence computing $\widetilde{ku}_*(M_n)$ for $t-s \leq 4$. See the proof of Theorem 4.57 for more information.

Theorem 4.60. If $n \equiv 0 \mod 4$, the first few spin^c bordism groups of X_n are

$$\widetilde{\Omega}_{0}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/2$$

$$\widetilde{\Omega}_{1}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/n$$

$$\widetilde{\Omega}_{2}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/4 \oplus \mathbb{Z}/2$$

$$\widetilde{\Omega}_{3}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/(n/2) \oplus (\mathbb{Z}/2)^{\oplus 2}$$

$$\widetilde{\Omega}_{4}^{\mathrm{Spin}^{c}}(X_{n}) \cong \mathbb{Z}/8 \oplus \mathbb{Z}/4 \oplus \mathbb{Z}/2.$$

Because Lemma 3.24 implies $\widetilde{\Omega}_5^{\text{Spin}^c}(X_n)$ is torsion, the phase homology groups for this symmetry type are $\mathbb{Z}/(n/2) \oplus (\mathbb{Z}/2)^{\oplus 2}$ for d = 2 and $\mathbb{Z}/8 \oplus \mathbb{Z}/4 \oplus \mathbb{Z}/2$ for d = 3.

Proof. We closely follow the proof of Theorem 4.41. For odd-primary torsion, use Proposition 4.24 to see that the odd-primary torsion in the range we care about is isomorphic to the odd torsion in \mathbb{Z}/n in degrees 1 and 3, and is 0 in degrees 0, 2, and 4.

On to the prime 2. In Theorem 4.41, we established a splitting $X_n \simeq (B\mathbb{Z}/2)^{\sigma-1} \vee M_n$, allowing us to focus solely on $\widetilde{\Omega}^{\text{Spin}}_*(M_n)$: $MTSpin^c \wedge (B\mathbb{Z}/2)^{\sigma-1} \simeq MTPin^c$ (2.10c), and we know pin^c bordism groups thanks to Bahri-Gilkey [BG87a, BG87b]. In (4.44), we determined the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(M_n; \mathbb{Z}/2)$ in low degrees, and the isomorphisms of $\mathcal{E}(1)$ -modules $R_1 \cong \mathbb{Z}/2 \oplus \Sigma R_0$ and $\mathcal{A}(1) \cong \mathcal{E}(1) \oplus \Sigma^2 \mathcal{E}(1)$ mean that as $\mathcal{E}(1)$ -modules,

(4.61)
$$H^*(M_n; \mathbb{Z}/2) \cong \Sigma \mathbb{Z}/2 \oplus \Sigma^2 R_0 \oplus \Sigma^2 \mathcal{O} \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus P,$$

where P is 4-connected. A priori, $\operatorname{Ext}(P)$ could have nonzero differentials to elements of the 4-line, but we will see that this does not happen without needing to compute $\operatorname{Ext}(P)$. In Figure 9, left, we draw (4.61). To determine the E_2 -page of the Adams spectral sequence, see (3.7) for $\operatorname{Ext}(\mathbb{Z}/2)$, Proposition 4.48 for $\operatorname{Ext}(R_0)$, and (4.55) for $\operatorname{Ext}(\hat{O})$. We draw the E_2 -page of the Adams spectral sequence for $\widetilde{ku}_*(M_n)$, as in Figure 9, right — though for legibility, most v_1 -actions are hidden. Lemma 3.24 implies there must be differentials in this range, though not necessarily the d_2 s pictured.



FIGURE 9. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*(M_n; \mathbb{Z}/2)$ in low degrees. The pictured submodule contains all elements in degrees 4 and below. The dashed line indicates a $\mathbb{Z}/2^r$ Bockstein, which we use to resolve a differential. Right: Ext of this submodule, which is the E_2 -page of the Adams spectral sequence computing $\widetilde{ku}_*(M_n)$ for $t - s \leq 4$. Most v_1 -actions are hidden for readability. See the proof of Theorem 4.60 for more information.

For $\widetilde{\Omega}_{1}^{\text{Spin}^{c}}(M_{n})$ to be torsion, there must be a differential d_{r} from the 2-line to the 1-line; then, $\widetilde{\Omega}_{1}^{\text{Spin}^{c}}(M_{n}) \cong \mathbb{Z}/2^{r}$, and since $\Omega_{1}^{\text{Pin}^{c}} \cong 0$, $\widetilde{\Omega}_{1}^{\text{Spin}^{c}}(M_{n}) \cong \mathbb{Z}/2^{r}$ as well. Differentials between towers, such as this d_{r} , are characterized by the May-Milgram theorem [MM81], and just as in the proof of Theorem 4.53, we conclude r is the largest natural number such that $2^{r} \mid n$. Combining this with our odd-torsion calculation, $\widetilde{\Omega}_{1}^{\text{Spin}^{c}}(M_{n}) \cong \mathbb{Z}/n$.

Continuity in increasing topological degree, this d_r kills the entire orange tower in the 2-line, and we infer $\widetilde{\Omega}_2^{\text{Spin}^c}(M_n) \cong \mathbb{Z}/2$. The green and blue summands in the 3-line survive and split off by Margolis' theorem. v_1 -equivariance of differentials implies that $d_r: E_2^{s,4+s} \to E_2^{s+2,s+3}$ is nonzero, and again maps the orange tower to the dark red tower, leaving a single $\mathbb{Z}/2$ summand in $E_3^{1,4}$. There can be no further differentials to the 3-line, so $\widetilde{\Omega}_3^{\text{Spin}^c}(M_n)_2^{\Lambda} \cong \mathbb{Z}/2^{r-1} \oplus (\mathbb{Z}/2)^{\oplus 2}$. Our odd-primary calculation then tells us that $\widetilde{\Omega}_3^{\text{Spin}^c}(M_n) \cong \mathbb{Z}/(n/2) \oplus (\mathbb{Z}/2)^{\oplus 2}$. Finally, the orange tower in the 4-line is killed by the d_r we most recently discussed, and the two light red $\mathbb{Z}/2$ summands in the 4-line cannot emit or receive differentials. Thus as promised Ext(P) does not have nonzero differentials to the 4-line, so we conclude by adding the pin^c bordism summands back in.

4.4.5. Comparison with [ZWY⁺20]. Interacting fermionic phases equivariant for a dihedral group D_{2n} acting by rotations and reflections have also been studied by Zhang-Wang-Yang-Qi-Gu [ZWY⁺20], who considered both spinless and spin-1/2 phases in dimension 2 + 1 for all n, and in Altland-Zirnbauer class D. They also study systems without a spatial symmetry, using the extended supercohomology classification of Wang-Gu [WG18, WG20] to classify these phases and discuss the FCEP for dihedral groups. We find complete agreement with their results except for phases with spinless fermions when $n \equiv 0 \mod 4$, where we predict $\mathbb{Z}/2 \oplus \mathbb{Z}/2$ and they predict $\mathbb{Z}/2$. This appears to arise from a calculation error: as we note below in Remark 4.62, the comparison map between supercohomology and the Anderson dual of spin bordism is an isomorphism for this symmetry type.

Remark 4.62. The phases we classify are realized by the extended supercohomology classifications of Wang-Gu [WG18, WG20] and Kapustin-Thorngren [KT17].²² Gaiotto-Johnson-Freyd [GJF19, §§5.4–5.6] determine that the extended supercohomology classification à la [KT17, WG18] is the cohomology of $(BD_{2n})^{2-V_{\lambda}}$ or $(BD_{2n})^{\text{Det}(V_{\lambda})-1}$ with respect to a spectrum they call fGP[×]₂₂, which is equivalent to the (-3)-connected cover of $I_{\mathbb{Z}}MTSpin$. Wang-Gu's refinement in [WG20] corresponds instead to the spectrum fGP[×], equivalent to the (-7)-connected cover of $I_{\mathbb{Z}}MTSpin$.²³

²²These classifications concern phases with an internal D_{2n} symmetry, but the fermionic crystalline equivalence principle allows us to pass back and forth.

 $^{^{23}}$ The reader may at this point wonder why our classification is a generalized *homology* theory, while these extended supercohomology classifications are generalized *cohomology* theories. This is a subtle point. The passage between homology and cohomology occurs because in these dimensions, we may approximate *MTSpin* by *KO* due to Anderson-Brown-Peterson's [ABP67] study of the connectivity of the Atiyah-Bott-Shapiro map [ABS64], then use that *KO* is shifted Anderson self-dual [And69,

ARUN DEBRAY

The connective covering maps induce comparison maps from the classifications of fermionic phases using extended supercohomology to the classification of fermionic phases under our ansatz. For fGP[×], the map is sufficiently connected as to be an isomorphism between the classifications of (d + 1)-dimensional phases for all $d \leq 5$. For fGP[×]_{≤ 2}, the map is not always an isomorphism even for d = 2: the cokernel when computing supercohomology of X is $\tilde{H}^0(X;\mathbb{Z})$, and this is nonzero e.g. for $X = (BC_n)^{2-V_{\lambda}}$ from §4.3. But for dihedral groups, $\tilde{H}^0((BD_{2n})^{\xi};\mathbb{Z})$ vanishes whenever $\xi \to BD_{2n}$ is a rank-0 unorientable virtual vector bundle, so in this case the comparison map is an isomorphism.

4.5. D_{2n} acting by rotations. The dihedral group D_{2n} can act on \mathbb{R}^3 in an orientation-preserving manner, where $C_n \subset D_{2n}$ acts by rotations in a plane and a preimage of the generator of $D_{2n}/C_n \cong \mathbb{Z}/2$ acts by a rotation perpendicular to that plane. Said differently, this point group is defined by a representation $\lambda: D_{2n} \to SO_3$ which decomposes as $\rho \oplus \sigma$, where ρ is the standard two-dimensional representation by rotations and reflections, and $\sigma: D_{2n} \to O_1$ is the sign representation, which is the determinant of ρ . Confusingly, this point group is sometimes called "three-dimensional dihedral symmetry;" in this convention, the three-dimensional action by $\rho \oplus \mathbb{R}$ is called *pyramidal symmetry*.

As far as we know, interacting fermionic phases for this D_{2n} symmetry have not been studied in the literature before.

n	Class D, spinless §4.5.1	Class D, spin-1/2 §4.5.2	Class A, spinless §4.5.3	Class A, spin-1/2 §4.5.4
$0 \mod 4$	$\mathbb{Z}/2$	$(\mathbb{Z}/2)^{\oplus 2}$	0	$(\mathbb{Z}/2)^{\oplus 2}$
$2 \mod 4$	0	$(\mathbb{Z}/2)^{\oplus 2}$	0	$(\mathbb{Z}/2)^{\oplus 3}$
$1,3 \bmod 4$	0	0	0	0

TABLE 5. D_{2n} -equivariant phase homology groups, where D_{2n} acts faithfully on \mathbb{R}^3 by rotations. These arise as homotopy groups of Anderson duals of $MTSpin \wedge X_n$ and $MTSpin^c \wedge X_n$, where X_n is one of $(BD_{2n})^{3-V_{\lambda}}$ or $(BD_{2n})_+$. See §4.5 for details and proofs.

For any representation $\phi: D_{2n} \to O_d$, let $V_{\phi} \to BD_{2n}$ denote the associated vector bundle.

Lemma 4.63.

- (1) If n is odd, V_{λ} is pin^c but not pin⁻.
- (2) If n is even, V_{λ} is not pin^c.

Proof. For (2), we show that if β is the integral Bockstein, $\beta w_2(V_\lambda) \neq 0$. By Lemma 3.28, it suffices to show $\operatorname{Sq}^1(w_2(V_\lambda)) \neq 0$. For $n \equiv 2 \mod 4$,

(4.64a)
$$w_2(V_{\lambda}) = w_2(V_{\rho}) + w_1(V_{\rho})w_1(V_{\sigma}) + w_2(V_{\sigma}) = x^2 + xy + y^2,$$

and $Sq^{1}(x^{2} + xy + y^{2}) = x^{2}y + xy^{2}$. For $n \equiv 0 \mod 4$,

(4.64b)
$$w_2(V_{\lambda}) = w_2(V_{\rho}) + w_1(V_{\rho})w_1(V_{\sigma}) + w_2(V_{\sigma}) = w + x^2,$$

and $\operatorname{Sq}^1(w + x^2) = wx$, so in neither case is $V_{\lambda} \operatorname{pin}^c$.

Now (1). Choose $i: \mathbb{Z}/2 \hookrightarrow D_{2n}$ given by a splitting of $D_{2n} \twoheadrightarrow D_{2n}/C_n \cong \mathbb{Z}/2$; restricting to $\mathbb{Z}/2$ along i, λ decomposes as $2\sigma \oplus \mathbb{R}$. Therefore $i^*V_{\lambda} \to B\mathbb{Z}/2$ is spin^c but not spin: $w_2(2V_{\sigma}) = w_1(V_{\sigma})^2 = x^2$, and for any vector bundle $V, V \oplus V$ admits a complex structure, hence a spin^c structure. In particular, $\beta(w_2(i^*V_{\lambda})) \neq 0$. The maps $\mathbb{Z}/2 \hookrightarrow D_{2n} \twoheadrightarrow \mathbb{Z}/2$ compose to the identity, so the induced maps on cohomology also compose to the identity. Therefore $\beta(w_2(V_{\lambda})) \neq 0$ too.

These propositions are the analogues of Propositions 4.23 and 4.24, helping us calculate odd-primary torsion in phase homology groups.

FMS07, HS14, Ric16, HLN20] to pass between $I_{\mathbb{Z}}KO$ -homology and $\Sigma^4 KO$ -cohomology. See Freed-Hopkins [FH19a, §5.1] for further discussion.

Lemma 4.65 (Handel [Han93, Theorems 5.2, 5.3]).

(4.66)
$$\widetilde{H}_k(BD_{2n};\mathbb{Z}[1/2]) \cong \begin{cases} \mathbb{Z}/n, & k \equiv 3 \mod 4\\ 0, & \text{otherwise.} \end{cases}$$

As usual, Handel computes $H^*(BD_{2n};\mathbb{Z})$, and it is up to us to change to homology with $\mathbb{Z}[1/2]$ coefficients.

Proposition 4.67. Suppose V is a rank-zero oriented virtual vector bundle.

- (1) The odd-torsion subgroup of $\widetilde{\Omega}_{k}^{\text{Spin}}((BD_{2n})^{V})$ is isomorphic to the odd-torsion subgroup of \mathbb{Z}/n when k = 3 and vanishes for all other $k \leq 6$.
- (2) The odd-torsion subgroup of $\widetilde{\Omega}_k^{\text{Spin}^c}((BD_{2n})^V)$ is isomorphic to the odd-torsion subgroup of \mathbb{Z}/n when k = 3 and k = 5 and vanishes for all other $k \leq 6$.

Proof. It suffices to work at odd primes. There are odd-primary equivalences $MTSpin \to MTSO$ and $MTSpin^c \to MTSO \wedge (BU_1)_+$; moreover, since V is oriented, there is a Thom isomorphism $MTSO \wedge (BD_{2n})_+ \xrightarrow{\sim} MTSO \wedge (BD_{2n})^V$. Therefore it suffices to study $\widetilde{\Omega}_*^{SO}(BD_{2n})$ for (1) and $\widetilde{\Omega}_*^{SO}(BD_{2n} \wedge BU_1)$ for (2) after completing at an odd prime p. Using Proposition 4.67 for input, as well as the Künneth formula to determine $H_*(BD_{2n} \wedge BU_1)_p^{\wedge}$, one sees that the Atiyah-Hirzebruch spectral sequences computing these bordism groups collapse for degree reasons in total degree 6 and below.

4.5.1. Class D, spinless case. Let f_0^D denote the equivariant local system of symmetry types for this case. Theorem 2.11 tells us that to compute $Ph_*^{D_{2n}}(\mathbb{R}^3, f_0^D)$, we should study the spin bordism of $X_n := (BD_n)^{3-V_{\lambda}}$.

Proposition 4.68. Suppose n is odd. Then

$$\begin{split} \widetilde{\Omega}_{0}^{\mathrm{Spin}}(X_{n}) &\cong \mathbb{Z} \\ \widetilde{\Omega}_{1}^{\mathrm{Spin}}(X_{n}) &\cong \mathbb{Z}/4 \\ \widetilde{\Omega}_{2}^{\mathrm{Spin}}(X_{n}) &\cong 0 \\ \widetilde{\Omega}_{3}^{\mathrm{Spin}}(X_{n}) &\cong \mathbb{Z}/n \\ \widetilde{\Omega}_{4}^{\mathrm{Spin}}(X_{n}) &\cong \mathbb{Z} \\ \widetilde{\Omega}_{5}^{\mathrm{Spin}}(X_{n}) &\cong \mathbb{Z}/16 \\ \widetilde{\Omega}_{6}^{\mathrm{Spin}}(X_{n}) &\cong 0, \end{split}$$

and therefore $Ph_0^{D_{2n}}(\mathbb{R}^3, f_0^D) \cong 0.$

Proof. Proposition 4.67 shows that $\widetilde{\Omega}_{k}^{\text{Spin}}(X_{n})$ lacks odd-primary torsion for k = 4, 5, so it suffices to work at 2. The inclusion $\mathbb{Z}/2 \hookrightarrow D_{2n}$ induces an isomorphism $H^{*}(BD_{2n}; \mathbb{Z}/2) \to H^{*}(B\mathbb{Z}/2; \mathbb{Z}/2)$, as we saw in the proof of Lemma 4.20, hence by naturality of the Thom isomorphism gives an isomorphism

(4.69)
$$\widetilde{H}^*(X_n; \mathbb{Z}/2) \xrightarrow{\cong} \widetilde{H}^*((B\mathbb{Z}/2)^{3-V_\lambda|_{B\mathbb{Z}/2}}; \mathbb{Z}/2).$$

Restricted to $\mathbb{Z}/2$, $\lambda \cong 2\sigma \oplus \mathbb{R}$, so by the stable Whitehead theorem, (4.69) gives a stable 2-primary equivalence $X_n \simeq (B\mathbb{Z}/2)^{2-2\sigma}$. Campbell [Cam17, §7.8] computes $\widetilde{\Omega}_k^{\text{Spin}}((B\mathbb{Z}/2)^{2-2\sigma})$, obtaining the free and 2-torsion summands we claim in the theorem statement.²⁴

Proposition 4.70 (Pedrotti [Ped17, Theorem 8.0.8]). For $n \equiv 2 \mod 4$, $\widetilde{\Omega}_4^{\text{Spin}}(X_n) \cong \mathbb{Z}$, and by Lemma 3.24 $\widetilde{\Omega}_5^{\text{Spin}}(X_n)$ is torsion. Therefore $Ph_0^{D_{2n}}(\mathbb{R}^3, f_0^D)$ vanishes.

Remark 4.71. Pedrotti reports this computation in terms of w_1 and w_2 of $3 - V_{\lambda}$, rather than λ itself, so we should check that our characteristic classes agree with his: we want $w_1(3-V_{\lambda}) = 0$ and $w_2(3-V_{\lambda}) = x^2 + xy + y^2$. Indeed Im $(\lambda) \subset$ SO₃, so V_{λ} is orientable, and from (4.64a) that $w_2(V_{\lambda}) = x^2 + xy + y^2$. Since $w_1(V_{\lambda}) = 0$, these are also w_1 and w_2 of $3 - V_{\lambda}$, as desired.

Proposition 4.72 (Pedrotti [Ped17, Theorem 9.0.14]). For $n \equiv 0 \mod 4$, $\widetilde{\Omega}_4^{\text{Spin}}(X_n) \cong \mathbb{Z} \oplus \mathbb{Z}/2$, and by Lemma 3.24 $\widetilde{\Omega}_5^{\text{Spin}}(X_n)$ is torsion. Therefore $Ph_0^{D_{2n}}(\mathbb{R}^3, f_0^D) \cong \mathbb{Z}/2$.

²⁴Campbell computes only through dimension 5, and Beaudry-Campbell [BC18, Figure 26] shows how to extend Campbell's computation to dimension 6.

Pedrotti takes as input $w_1(3 - V_\lambda) = 0$ and $w_2(3 - V_\lambda) = w + x^2$, which agrees with the classes of V_λ (e.g. (4.64b)). Beware that what we call x he calls y, and vice versa!

4.5.2. Class D, spin-1/2 case. Let $f_{1/2}^D$ denote the equivariant local system of symmetry types for this case. Lemma 4.63 and Theorem 2.11 tell us that to compute $Ph_*^{D_{2n}}(\mathbb{R}^3, f_{1/2}^D)$, we should study the spin bordism of $(BD_n)^{\text{Det}(V_\lambda)-1}$. Since V_λ is orientable, this is isomorphic to $\Omega_4^{\text{Spin}}(BD_{2n})$.

Proposition 4.73. Suppose *n* is odd. Then $\Omega_4^{\text{Spin}}(BD_{2n}) \cong \mathbb{Z}$ and $\Omega_5^{\text{Spin}}(BD_{2n}) \cong 0$, so $Ph_0^{D_{2n}}(\mathbb{R}^3, f_{1/2}^D) \cong 0$.

Proof. The proof is almost the same as that of Proposition 4.68: by Proposition 4.67, there is no odd-primary torsion, and $B\mathbb{Z}/2 \to BD_{2n}$ induces an isomorphism on mod 2 cohomology, hence also on 2-local spin bordism, and Mahowald-Milgram [MM76] show $\Omega_4^{\text{Spin}}(B\mathbb{Z}/2) \cong \mathbb{Z}$ and $\Omega_5^{\text{Spin}}(B\mathbb{Z}/2) \cong 0.^{25}$

Proposition 4.74. For *n* even, $\Omega_4^{\text{Spin}}(BD_{2n}) \cong \mathbb{Z} \oplus (\mathbb{Z}/2)^{\oplus 2}$.

Proof. Pedrotti [Ped17, Theorems 8.0.4 and 9.0.3] shows $\Omega_4^{\text{Spin}}(BD_{2n}) \cong \mathbb{Z} \oplus H_4(BD_{2n};\mathbb{Z})$, and the latter is computed by Handel [Han93, Theorem 5.2].

Bruner-Greenlees [BG10, Corollary 8.5.9] also compute this when n is a power of 2. By Lemma 3.24, $\Omega_5^{\text{Spin}}(BD_{2n})$ is torsion, so $Ph_0^D(\mathbb{R}^3, f_{1/2}^D) \cong \mathbb{Z}/2 \oplus \mathbb{Z}/2$.

4.5.3. Class A, spinless case. Let f_0^A denote the equivariant local system of symmetry types in the spinless type A case. By Lemma 4.63, we should compute $\widetilde{\Omega}_*^{\text{Spin}^c}(X_n)$, where $X_n := (BD_{2n})^{3-V_{\lambda}}$.

When n is odd, V_{λ} is spin^c, so there is a Thom isomorphism $MTSpin^{c} \wedge X_{n} \simeq MTSpin^{c} \wedge (BD_{2n})_{+}$.

Theorem 4.75. Suppose n is odd. Then

$$\Omega_0^{\text{Spin}^c}(BD_{2n}) \cong \mathbb{Z}$$
$$\Omega_1^{\text{Spin}^c}(BD_{2n}) \cong \mathbb{Z}/2$$
$$\Omega_2^{\text{Spin}^c}(BD_{2n}) \cong \mathbb{Z}$$
$$\Omega_3^{\text{Spin}^c}(BD_{2n}) \cong \mathbb{Z}/4n$$
$$\Omega_4^{\text{Spin}^c}(BD_{2n}) \cong \mathbb{Z}^2$$
$$\Omega_5^{\text{Spin}^c}(BD_{2n}) \cong \mathbb{Z}/8n \oplus \mathbb{Z}/2$$
$$\Omega_6^{\text{Spin}^c}(BD_{2n}) \cong \mathbb{Z}^2.$$

Therefore $Ph_0^{D_{2n}}(\mathbb{R}^3, f_0^A) \cong 0.$

Proof. Proposition 4.67 accounts for the odd-primary torsion, so we just have to work at 2. The map $\mathbb{Z}/2 \hookrightarrow D_{2n}$ induced by a choice of reflection defines an isomorphism on mod 2 cohomology, therefore by the stable Whitehead theorem is a 2-local stable equivalence. Therefore it defines an isomorphism $\Omega^{\text{Spin}^c}_*(B\mathbb{Z}/2)^{\wedge}_2 \to \Omega^{\text{Spin}^c}_*(BD_{2n})^{\wedge}_2$, and the spin^c bordism of $B\mathbb{Z}/2$ is computed by Bahri-Gilkey [BG87a, BG87b].

Theorem 4.76. Suppose $n \equiv 2 \mod 4$. Then the first few spin^c bordism groups of X_n are

$$\Omega_0^{\operatorname{Spin}^c}(X) \cong \mathbb{Z}$$
$$\widetilde{\Omega}_1^{\operatorname{Spin}^c}(X) \cong (\mathbb{Z}/2)^{\oplus 2}$$
$$\widetilde{\Omega}_2^{\operatorname{Spin}^c}(X) \cong \mathbb{Z}$$
$$\widetilde{\Omega}_3^{\operatorname{Spin}^c}(X) \cong \mathbb{Z}/n \oplus (\mathbb{Z}/2)^{\oplus 2}$$
$$\widetilde{\Omega}_4^{\operatorname{Spin}^c}(X) \cong \mathbb{Z}^2,$$

and $\widetilde{\Omega}_5^{{\rm Spin}^c}(X)$ is torsion.

²⁵This is also been computed by other methods by Mahowald [Mah82, Lemma 7.3], Bruner-Greenlees [BG10, Example 7.3.1], Siegemeyer [Sie13, Theorem 2.1.5], and García-Etxebarria and Montero [GEM19, (C.18)].

Proof. The odd-torsion subgroups can be read off of (4.67). For the 2-primary part, we use the Adams spectral sequence over $\mathcal{E}(1)$. Letting U denote the Thom class, we saw $w_1(V_{\lambda}) = 0$, so $\operatorname{Sq}^1(U) = 0$, and (4.64a) $w_2(V_{\lambda}) = x^2 + xy + y^2$, so $\operatorname{Sq}^2(U) = U(x^2 + xy + y^2)$. Using this and the Cartan formula, we have an $\mathcal{E}(1)$ -module isomorphism

(4.77)
$$\widetilde{H}^*(X_n; \mathbb{Z}/2) \cong \mathcal{O} \oplus \Sigma \mathcal{E}(1) \oplus \Sigma \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus P,$$

where P is 4-connected. We draw this in Figure 10, left. A priori Ext(P) could have nonzero differentials to the 4-line and therefore affect our computation, but we will see that this cannot happen without needing to determine Ext(P).



FIGURE 10. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*(X_n; \mathbb{Z}/2)$ when $n \equiv 2 \mod 4$. The pictured summand contains all elements in degrees 4 and below. Right: the E_2 -page of the corresponding Adams spectral sequence computing $\widetilde{ku}_*(X_n)_2^{\wedge}$.

We calculated $\operatorname{Ext}_{\mathcal{E}(1)}(\mathcal{O})$ in (4.55), so we can draw the E_2 -page of the Adams spectral sequence in Figure 10, right. h_0 -equivariance rules out nonzero differentials in degrees 3 and below, but a priori there could be a nonzero differential from the 5-line to then 4-line. To rule this out, use Lemma 3.24 to see that $\widetilde{ku}_4(X_n)$ has one free summand. Therefore there cannot be any nonzero differentials to the 4-line: h_0 -equivariance would mean that if there were such a differential, it would kill all but finitely many summands in the 4-line of the E_2 -page, preventing $\widetilde{ku}_4(X_n)$ from having a free part. \Box

Theorem 4.78. When $n \equiv 0 \mod 4$, the first few spin^c bordism groups of X_n are

$$\begin{split} \widetilde{\Omega}_{0}^{\mathrm{Spin}^{c}}(X_{n}) &\cong \mathbb{Z} \\ \widetilde{\Omega}_{1}^{\mathrm{Spin}^{c}}(X_{n}) &\cong (\mathbb{Z}/2)^{\oplus 2} \\ \widetilde{\Omega}_{2}^{\mathrm{Spin}^{c}}(X_{n}) &\cong \mathbb{Z} \\ \widetilde{\Omega}_{3}^{\mathrm{Spin}^{c}}(X_{n}) &\cong \mathbb{Z}/n \oplus (\mathbb{Z}/2)^{\oplus 2} \\ \widetilde{\Omega}_{4}^{\mathrm{Spin}^{c}}(X_{n}) &\cong \mathbb{Z}^{2}, \end{split}$$

and $\widetilde{\Omega}_5^{\text{Spin}^c}(X_n)$ is torsion. Therefore $Ph_0^{D_{2n}}(\mathbb{R}^3, f_0^A) \cong 0$.

Proof. The odd-torsion subgroups are calculated in Proposition 4.67. For the 2-torsion, we use the Adams spectral sequence over $\mathcal{E}(1)$. Recall that $w_1(V_{\lambda}) = 0$ and (from (4.64b)) $w_2(V_{\lambda}) = w + x^2$, so $w_1(3 - V_{\lambda}) = 0$ and $w_2(3 - V_{\lambda}) = w + x^2$. Thus in $\tilde{H}^*(X_n; \mathbb{Z}/2)$, Sq¹(U) = 0 and Sq²(U) = U(w + x^2). Using this and the Cartan formula, we can compute the $\mathcal{E}(1)$ -action on $\tilde{H}^*(X_n; \mathbb{Z}/2)$, and find that

(4.79)
$$\widetilde{H}^*(X_n; \mathbb{Z}/2) \cong \mathcal{O} \oplus \Sigma \mathcal{E}(1) \oplus \Sigma \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^3 \mathbb{Z}/2 \oplus \Sigma^4 \mathcal{O} \oplus P$$

where P is 4-connected. We draw this in Figure 11, left. We will see in a moment that Ext(P) has no nonzero differentials to elements in degree 4 and below, which means we can ignore it in our computations. We calculated $Ext_{\mathcal{E}(1)}(\mathcal{O})$ in (4.55), so we can draw the E_2 -page of the Adams spectral sequence in Figure 10, right. Margolis' theorem (Theorem 3.22) implies the only possible nonzero differentials from an element of topological degree 4 or below are the differentials from a tower in the 4-line to the blue tower in the 3-line,



FIGURE 11. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*(X_n; \mathbb{Z}/2)$ when $n \equiv 0 \mod 4$. The pictured summand contains all elements in degrees 4 and below. The gray dashed line indicates a $\mathbb{Z}/2^r$ Bockstein, where r is the largest number for which $2^r \mid n$; this is not part of the $\mathcal{E}(1)$ -module structure, but we use it in Theorem 4.78 to resolve a differential. Right: the E_2 -page of the corresponding Adams spectral sequence computing $\widetilde{ku}_*(X_n)_2^{\wedge}$; v_1 -actions are hidden for legibility. We will see in Theorem 4.78 that there is a d_r from the purple tower in the 4-line to the 3-line, though it is not always the d_2 pictured.

and Lemma 3.24 implies $\widetilde{ku}_4(X_n)$ has free rank 1, so for some r this differential d_r is nonzero. Moreover, its source must be the purple tower: the red tower is in the image of $v_1: E_2^{s,s+2} \to E_2^{s+1,s+5}$, so if $d_r(x) = y$ for any element x of the red tower in degree 4, then y is also in the image of v_1 , but the blue tower is not in this image. Therefore we know that d_r kills the entire purple tower in degree 4, and the red tower survives to the E_{∞} -page: the red tower supports no nonzero differentials to the 3-line, and if there were a differential from the 5-line to the red tower, h_0 -linearity guarantees it would kill all but finitely many summands of the red tower, contradicting Lemma 3.24.

It remains only to determine the value of r. In $H^*(BD_{2n}; \mathbb{Z}/2)$, the $\mathbb{Z}/2^k$ Bockstein carries (a preimage of) wy to w^2 , where k is the largest number such that $2^k \mid n$. This can be checked by, e.g., pulling back to BC_n , where this Bockstein is discussed by [Cam17, DL20a]. The Thom isomorphism theorem implies the $\mathbb{Z}/2^k$ Bockstein sends (a preimage of) Uwy to Uw^2 , and therefore by the May-Milgram theorem [MM81], r = k.

4.5.4. Class A, spin-1/2 case. Let $f_{1/2}^A$ denote the equivariant local system of symmetry types in the spin-1/2 type A case. In this case the ansatz tells us to study $\Omega_*^{\text{Spin}^c}(BD_{2n})$.

Proposition 4.80. For *n* odd, $Ph_0^{D_{2n}}(\mathbb{R}^3, f_{1/2}^A) = 0.$

Proof. This follows from our computation of $\Omega_k^{\text{Spin}^c}(BD_{2n})$ in Theorem 4.75.

Theorem 4.81. Suppose $n \equiv 2 \mod 4$. Then

$$\Omega_{0}^{\mathrm{Spin}^{c}}(BD_{2n}) \cong \mathbb{Z}$$

$$\Omega_{1}^{\mathrm{Spin}^{c}}(BD_{2n}) \cong (\mathbb{Z}/2)^{\oplus 2}$$

$$\Omega_{2}^{\mathrm{Spin}^{c}}(BD_{2n}) \cong \mathbb{Z} \oplus \mathbb{Z}/2$$

$$\Omega_{3}^{\mathrm{Spin}^{c}}(BD_{2n}) \cong \mathbb{Z}/n \oplus (\mathbb{Z}/4)^{\oplus 2} \oplus (\mathbb{Z}/2)^{\oplus 2}$$

$$\Omega_{4}^{\mathrm{Spin}^{c}}(BD_{2n}) \cong \mathbb{Z}^{2} \oplus (\mathbb{Z}/2)^{\oplus 3}$$

$$\Omega_{5}^{\mathrm{Spin}^{c}}(BD_{2n}) \cong \mathbb{Z}/n \oplus (\mathbb{Z}/8)^{\oplus 2} \oplus \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 5}$$

$$\Omega_{6}^{\mathrm{Spin}^{c}}(BD_{2n}) \cong \mathbb{Z}^{2} \oplus (\mathbb{Z}/2)^{\oplus 6}.$$

Proof. We calculated the odd-primary torsion in these bordism groups in Proposition 4.67; now the 2-primary part. The inclusion $\mathbb{Z}/2 \times \mathbb{Z}/2 \to D_{2n}$ given by a reflection and a rotation by π induces an isomorphism on mod 2 cohomology, so by the stable Whitehead theorem, $BD_{2n} \to B(\mathbb{Z}/2 \times \mathbb{Z}/2)$ is an equivalence after stabilizing

and 2-completing. Therefore $\Omega^{\text{Spin}^c}_*(BD_{2n})_2^{\wedge} \xrightarrow{\cong} \Omega^{\text{Spin}^c}_*(B(\mathbb{Z}/2 \times \mathbb{Z}/2))_2^{\wedge}$; since $MTSpin^c \to ku \vee \Sigma^4 ku$ is an isomorphism in degrees 8 and below, it suffices to know $ku_*(B(\mathbb{Z}/2 \times \mathbb{Z}/2))$.

Ossa [Oss89, Proposition 3] computes $ku_*(B(\mathbb{Z}/2 \times \mathbb{Z}/2))$ by establishing an equivalence

(4.82)
$$ku \wedge B\mathbb{Z}/2 \wedge B\mathbb{Z}/2 \simeq (ku \wedge \Sigma^2 B\mathbb{Z}/2) \vee \Sigma^2 H(\mathbb{Z}/2[u,v]),$$

where the third term refers to a generalized Eilenberg-Mac Lane spectrum on the graded abelian group $\mathbb{Z}/2[u, v]$.²⁶ Using the stable splitting

(4.83)
$$\Sigma^{\infty}(B\mathbb{Z}/2 \times B\mathbb{Z}/2)_{+} \simeq \mathbb{S} \vee \Sigma^{\infty}B\mathbb{Z}/2 \vee \Sigma^{\infty}B\mathbb{Z}/2 \vee \Sigma^{\infty}(B\mathbb{Z}/2 \wedge B\mathbb{Z}/2),$$

we see that $ku_*(B(\mathbb{Z}/2 \times \mathbb{Z}/2))$ can be assembled from the following pieces.

- (1) $ku_*(pt)$, which contributes \mathbb{Z} in even degrees and 0 in odd degrees.
- (2) Two copies of $ku_*(B\mathbb{Z}/2)$. Hashimoto [Has83, Theorem 3.1] shows each copy vanishes in even degrees and is isomorphic to $\mathbb{Z}/2^{k+1}$ in odd degree 2k + 1.
- (3) $\widetilde{ku}_*(\Sigma^2 B\mathbb{Z}/2)$. Hashimoto (*ibid.*) shows this vanishes in even degrees and is isomorphic to $\mathbb{Z}/2^k$ in odd degree 2k + 1.
- (4) $\pi_*(\Sigma^2 H\mathbb{Z}/2[u,v])$, which contributes 0 in degrees 0 and 1 and $(\mathbb{Z}/2)^{\oplus (k-1)}$ in degrees $k \geq 2$.

Putting this together and adding in the odd-primary torsion, we obtain $ku_k(BD_{2n})$ for $k \leq 6$; using the Anderson-Brown-Peterson isomorphism $\Omega_k^{\text{Spin}^c}(BD_{2n}) \cong ku_k(BD_{2n}) \oplus ku_{k-4}(BD_{2n})$, valid for k < 8, we obtain the bordism groups in the theorem statement.

Theorem 4.84. Suppose $n \equiv 0 \mod 4$. Then

$$\Omega_0^{\operatorname{Spin}^c}(BD_{2n}) \cong \mathbb{Z}
\Omega_1^{\operatorname{Spin}^c}(BD_{2n}) \cong (\mathbb{Z}/2)^{\oplus 2}
\Omega_2^{\operatorname{Spin}^c}(BD_{2n}) \cong \mathbb{Z} \oplus \mathbb{Z}/2
\Omega_3^{\operatorname{Spin}^c}(BD_{2n}) \cong \mathbb{Z}/n \oplus (\mathbb{Z}/4)^{\oplus 2}
\Omega_4^{\operatorname{Spin}^c}(BD_{2n}) \cong \mathbb{Z}^2 \oplus (\mathbb{Z}/2)^{\oplus 2},$$

and $\Omega_5^{\operatorname{Spin}^c}(BD_{2n})$ is torsion. Therefore $\operatorname{Ph}_0^{D_{2n}}(\mathbb{R}^3, f_{1/2}^A) \cong (\mathbb{Z}/2)^{\oplus 2}$.

Proof. Since $\Omega^{\text{Spin}^c}_*(BD_{2n}) \cong \Omega^{\text{Spin}^c}_*(\text{pt}) \oplus \widetilde{\Omega}^{\text{Spin}^c}_*(BD_{2n})$, we will focus on $\widetilde{\Omega}^{\text{Spin}^c}_*(BD_{2n})$, adding on $\Omega^{\text{Spin}^c}_*(\text{pt})$ at the end. We also focus on the 2-primary story: the odd-primary torsion is calculated in Proposition 4.67.

Recall from Proposition 4.17 that $H^*(BD_{2n}; \mathbb{Z}/2) \cong \mathbb{Z}/2[x, y, w]/(xy + y^2)$, with |x| = |y| = 1 and |w| = 2. A choice of reflection induces a section of $BD_{2n} \to B\mathbb{Z}/2$, and therefore there is a spectrum M_n and a splitting

(4.85)
$$\Sigma^{\infty} BD_{2n} \xrightarrow{\simeq} M_n \vee \Sigma^{\infty} B\mathbb{Z}/2.$$

such that as a subspace of $\widetilde{H}^*(BD_{2n}; \mathbb{Z}/2)$, $\widetilde{H}^*(M_n; \mathbb{Z}/2)$ is complementary to the subspace S spanned by $\{x^n \mid n \geq 0\}$, because S is the image of the pullback map $\widetilde{H}^*(B\mathbb{Z}/2; \mathbb{Z}/2) \to \widetilde{H}^*(BD_{2n}; \mathbb{Z}/2)$. Bahri-Gilkey [BG87a, BG87b] show that $\widetilde{\Omega}_4^{\text{Spin}^c}(B\mathbb{Z}/2) \cong 0$, so we just have to understand $\widetilde{\Omega}_4^{\text{Spin}^c}(M_n)$.

We will use the Adams spectral sequence over $\mathcal{E}(1)$ to show that $\widetilde{ku}_0(M_n) \cong 0$ and $\widetilde{ku}_4(M_n) \cong (\mathbb{Z}/2)^{\oplus 2}$, which suffices to prove the theorem. For degree reasons, $\operatorname{Sq}(x) = x + x^2$ and $\operatorname{Sq}(y) = y + y^2$, and in Lemma 4.30 we saw $\operatorname{Sq}(w) = w + wx + w^2$. Using this, we find that as $\mathcal{E}(1)$ -modules,

(4.86)
$$\widetilde{H}^*(M_n;\mathbb{Z}/2) \cong \Sigma R_0 \oplus \Sigma^2 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{O} \oplus \Sigma^4 \mathbb{Z}/2 \oplus \Sigma^4 \mathcal{E}(1) \oplus \Sigma^4 \mathcal{E}(1) \oplus P_1$$

where P is 5-connected, and therefore too highly connected to affect our calculations. We draw this in Figure 12, left. We have already computed Ext(M) for the remaining summands M: see Proposition 4.48 for $\text{Ext}(R_0)$, (4.55) for $\text{Ext}(\hat{O})$, and (3.7) for $\text{Ext}(\mathbb{Z}/2)$. Therefore we obtain the E_2 -page of the Adams spectral sequence in Figure 12, right.

²⁶Ossa's splitting (4.82) or its analogue on homotopy groups has also been proven in several other ways: see Johnson-Wilson [JW97], Bruner [Bru99, Corollary 3.3], Bruner-Greenlees [BG03, Example 4.11.2], Powell [Pow14], and Bruner-Mira-Stanley-Snaith [BMSS15, Theorem 2.12].



FIGURE 12. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*(M_n; \mathbb{Z}/2)$ in low degrees. The gray dashed line indicates a $\mathbb{Z}/2^r$ Bockstein, where r is the largest number such that $2^r \mid n$; this is not part of the $\mathcal{E}(1)$ -module structure, but we will use it in Theorem 4.84 to resolve a differential. Right: the E_2 -page of the Adams spectral sequence computing $\widetilde{Ku}(M_n)^{\wedge}_2$. v_1 -actions are hidden for readability. We will see in Theorem 4.84 that there is a nonzero differential from the 4-line to the 3-line, though it is not necessarily the d_2 pictured.

The 0-line is empty, so $\widetilde{ku}_0(M_n) \cong 0$, as promised. Lemma 3.24 implies $\widetilde{ku}_3(M_n)$ is torsion; therefore there must be a nonzero differential d_r from the purple tower in the 4-line to the yellow tower in the 3-line. As in previous examples (Lemma 4.32 and Theorems 4.53 and 4.60), k is the largest number such that $2^k \mid n$: the $\mathbb{Z}/2^k$ Bockstein sends a preimage of wy to w^2 , which can be checked after pulling back to BC_n as usual. The May-Milgram theorem [MM81] then identifies r = k. Therefore from the E_{r+1} -page onward, the green tower is gone, and the 4-line consists only of the two $\mathbb{Z}/2$ summands in Adams filtration zero, so $\widetilde{ku}_4(M_n) \cong (\mathbb{Z}/2)^{\oplus 2}$.

5. EXAMPLES: TETRAHEDRAL, OCTAHEDRAL, AND ICOSAHEDRAL SYMMETRIES

Point group	Ref.	Class D, spinless	Class D, spin- $1/2$	Class A, spinless	Class A, spin- $1/2$
Chiral tet. (A_4, T) Pyrit. $(A_4 \times \mathbb{Z}/2, T_h)$ Full tet. (S_4, T_d)	\$5.1 \$5.2 \$5.3	$0 \\ (\mathbb{Z}/2)^{\oplus 3} \\ \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 2}$	$0 \ \mathbb{Z}/2 \ 0$	$0 \\ \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 3} \\ (\mathbb{Z}/2)^{\oplus 4}$	$egin{array}{l} 0 \ \mathbb{Z}/8 \oplus (\mathbb{Z}/2)^{\oplus 3} \ \mathbb{Z}/8 \oplus (\mathbb{Z}/2)^{\oplus 2} \end{array}$
Chiral oct. (S_4, O) Full oct. $(S_4 \times \mathbb{Z}/2, O_h)$	\$5.4 \$5.5	$0 \ (\mathbb{Z}/2)^{\oplus 4}$	$\mathbb{Z}/2\ (\mathbb{Z}/2)^{\oplus 2}$	$0 \ \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 4}$	$\frac{\mathbb{Z}/2}{\mathbb{Z}/8 \oplus \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 4}}$
Chiral icos. (A_5, I) Full icos. $(A_5 \times \mathbb{Z}/2, I_h)$	$\frac{5.6}{5.7}$	$0 \ (\mathbb{Z}/2)^{\oplus 3}$	$0 \ \mathbb{Z}/2$	$0 \ \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 3}$	$0 \ \mathbb{Z}/8 \oplus (\mathbb{Z}/2)^{\oplus 3}$

TABLE 6. Phase homology groups in dimension 3 + 1 equivariant with respect to various tetrahedral, octahedral, and icosahedral symmetries and the ways they can mix with fermion parity. See the referenced sections for how the fermionic crystalline equivalence principle associates this data with symmetry types for invertible TFTs.

5.1. Chiral tetrahedral symmetry. We compute phase homology groups equivariant for a chiral tetrahedral symmetry $\lambda: A_4 \to SO_3$. As far as we know, this point group has not yet been considered by physicists in the setting of fermionic phases. We will show that our ansatz implies there are no nontrivial phases with either spinless or spin-1/2 fermions in both class D and class A. As usual, $V_{\lambda} \to BA_4$ denotes the vector bundle associated to λ .

Proposition 5.1. $H^*(BA_4; \mathbb{Z}/2) \cong \mathbb{Z}/2[u, v, w]/(u^3 + v^2 + w^2 + vw)$, where |u| = 2 and |v| = |w| = 3. Sq $(u) = u + v + w + u^2$, Sq $(v) = v + u^2 + uw + v^2$, and Sq $(w) = w + u^2 + uv + w^2$. Except for the Steenrod operations, this result can be found in several places, such as [Kin] and [AM04, Theorem III.1.3], so we will be brief.

Proof sketch. Use the Lyndon-Hochschild-Serre spectral sequence [Lyn48, Ser50, HS53] for the short exact sequence $1 \to \mathbb{Z}/2 \times \mathbb{Z}/2 \to A_4 \to \mathbb{Z}/3 \to 1$; the mod 2 cohomology of $\mathbb{Z}/3$ is trivial, so the spectral sequence collapses, and

(5.2)
$$H^*(BA_4; \mathbb{Z}/2) \cong H^0(B\mathbb{Z}/3; H^*(B\mathbb{Z}/2 \times B\mathbb{Z}/2; \mathbb{Z}/2)) = H^*(B\mathbb{Z}/2 \times B\mathbb{Z}/2; \mathbb{Z}/2)^{\mathbb{Z}/3}.$$

We can choose this $\mathbb{Z}/3$ -action to be such that a generator of $\mathbb{Z}/3$ acts on $\mathbb{Z}/2 \times \mathbb{Z}/2 = \{1, \alpha, \beta, \alpha + \beta\}$ by $\alpha \mapsto \alpha + \beta$, $\beta \mapsto \alpha$, and $\alpha + \beta \mapsto \beta$. In a mild abuse of notation, we identify $\mathbb{Z}/2 \times \mathbb{Z}/2$ with $H^1(B\mathbb{Z}/2 \times B\mathbb{Z}/2; \mathbb{Z}/2) \cong \operatorname{Hom}(\mathbb{Z}/2 \times \mathbb{Z}/2, \mathbb{Z}/2)$: these are dual $\mathbb{Z}/2$ -vector spaces, and we have a basis for one, which we identify with the dual basis vectors of the other. Thus $H^*(B\mathbb{Z}/2 \times B\mathbb{Z}/2; \mathbb{Z}/2) \cong \mathbb{Z}/2[\alpha, \beta]$.

The unique nonzero degree-2 cohomology class fixed by $\mathbb{Z}/3$ is $u \coloneqq \alpha^2 + \alpha\beta + \beta^2$, and two linearly independent degree-3 classes fixed by $\mathbb{Z}/3$ are $v \coloneqq \alpha^3 + \alpha^2\beta + \beta^3$ and $w \coloneqq \alpha^3 + \alpha\beta^2 + \beta^3$, whence the relation.

For the Steenrod squares, the identification in (5.2) of $H^*(BA_4; \mathbb{Z}/2)$ as a subalgebra of $H^*(B\mathbb{Z}/2 \times B\mathbb{Z}/2; \mathbb{Z}/2)$ is the pullback map for $B\mathbb{Z}/2 \times B\mathbb{Z}/2 \to BA_4$, hence \mathcal{A} -equivariant, so we can compute $\operatorname{Sq}(u)$ in $H^*(B\mathbb{Z}/2 \times B\mathbb{Z}/2; \mathbb{Z}/2)$; the computation follows from $\operatorname{Sq}(\alpha) = \alpha + \alpha^2$ and $\operatorname{Sq}(\beta) = \beta + \beta^2$.

Lemma 5.3. $w_1(V_{\lambda}) = 0$ and $w_2(V_{\lambda}) = u$.

Proof. Since V_{λ} is orientable, $w_1(V_{\lambda}) = 0$, and since V_{λ} is not spin, $w_2(V_{\lambda}) \neq 0$. Since $H^2(BA_4; \mathbb{Z}/2) \cong \mathbb{Z}/2 \cdot u$, $w_2(V_{\lambda}) = u$.

One way to see that this representation is not spin is to look at the *binary tetrahedral group* 2T, defined to be the preimage of $A_4 \subset SO_3$ under the double cover $Spin_3 \rightarrow SO_3$. If V_{λ} were spin, 2T would be a split extension of A_4 by μ_2 , but it is not split.

5.1.1. Class D, spinless case. If A_4 does not mix with the symmetry type, our ansatz reduces to that of Freed-Hopkins, which reduces the computation of these A_4 -equivariant phase homology groups to the computation of $[MTSpin \wedge (BA_4)^{3-V_{\lambda}}, \Sigma^5 I_{\mathbb{Z}}]$.

Theorem 5.4. The first few spin bordism groups of $X := (BA_4)^{3-V_{\lambda}}$ are

$$\widetilde{\Omega}_{0}^{\text{Spin}}(X) \cong \mathbb{Z}
\widetilde{\Omega}_{1}^{\text{Spin}}(X) \cong \mathbb{Z}/3
\widetilde{\Omega}_{2}^{\text{Spin}}(X) \cong 0
\widetilde{\Omega}_{3}^{\text{Spin}}(X) \cong \mathbb{Z}/6
\widetilde{\Omega}_{4}^{\text{Spin}}(X) \cong \mathbb{Z}
\widetilde{\Omega}_{5}^{\text{Spin}}(X) \cong \mathbb{Z}/18 \oplus \mathbb{Z}/2
\widetilde{\Omega}_{5}^{\text{Spin}}(X) \cong \mathbb{Z}/2
\widetilde{\Omega}_{7}^{\text{Spin}}(X) \cong \mathbb{Z}/9.$$

Thus if f_0^D denotes the A_4 -equivariant local system of symmetry types for this case, $Ph_0^{A_4}(\mathbb{R}^3, f_0^D) = 0$.

Proof. At the prime 2, we use the Adams spectral sequence; if p is an odd prime, the map $\widetilde{\Omega}^{\text{Spin}}_*(X) \to \widetilde{\Omega}^{\text{SO}}_*(X)$ is an isomorphism on p-torsion, and we will determine the p-torsion part of $\widetilde{\Omega}^{\text{SO}}_*(X)$.

First, the 2-primary piece. Letting U denote the mod 2 Thom class as usual, $\operatorname{Sq}^{1}(U) = 0$ and $\operatorname{Sq}^{2}(U) = Uu$. This and Proposition 5.1 allow us to determine the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^{*}(X;\mathbb{Z}/2)$ in low degrees, as depicted in Figure 13, left.

Hence as $\mathcal{A}(1)$ -modules,

(5.5)
$$H^*(X;\mathbb{Z}/2) \cong \mathcal{O} \oplus \Sigma^3 \mathcal{A}(1) \oplus \Sigma^5 \mathcal{A}(1) \oplus P,$$

where P is 8-connected. Because we only care about degrees 6 and below, P is irrelevant for us, and for the remaining summands in (5.5), $\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(-,\mathbb{Z}/2)$ has already been computed. For $\Sigma^k \mathcal{A}(1)$, there's a single $\mathbb{Z}/2$



FIGURE 13. Left: the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*((BA_4)^{3-V_{\lambda}}; \mathbb{Z}/2)$ in low degrees. This submodule contains all elements of degree at most 8. Right: the E_2 -page of the Adams spectral sequence calculating $\widetilde{ko}_*((BA_4)^{3-V_{\lambda}})$, given by $\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(\widetilde{H}^*((BA_4)^{3-V_{\lambda}};\mathbb{Z}/2),\mathbb{Z}/2)$.

with s = 0, t = k; for \mathcal{O} , see [BC18, Figure 29]. We put this together and display the E_2 -page for our spectral sequence in Figure 13, right. A combination of h_0 -equivariance and Margolis' theorem (Theorem 3.22) rules out nontrivial differentials and hidden extensions. Therefore the 2-primary part of $\widetilde{\Omega}_k^{\text{Spin}}(X)$ has a single free summand each in degrees 0 and 4, is 0 in degrees 1 and 2, is $\mathbb{Z}/2$ in degrees 3 and 6, and is $\mathbb{Z}/2 \oplus \mathbb{Z}/2$ in degree 5.

For the odd-primary part, we use the fact that $\Omega^{\text{Spin}}_* \to \Omega^{\text{SO}}_*$ is an equivalence after inverting 2. Moreover, because λ factors through SO₃, $V_{\lambda} \to BA_4$ is orientable, so there is a Thom isomorphism $\widetilde{\Omega}^{\text{SO}}_*(X) \xrightarrow{\cong} \Omega^{\text{SO}}_*(BA_4)$. Hence we just need the odd-primary part of $\Omega^{\text{SO}}_*(BA_4)$, which is isomorphic to the odd-primary part of $\Omega^{\text{Spin}}_*(BA_4)$. In the degrees we care about, this is isomorphic to $ko_*(BA_4)$, and Bruner-Greenlees [BG10, §7.7.E] show that the odd-primary torsion in $ko_*(BA_4)$ below degree 6 consists of $\mathbb{Z}/3$ summands in degrees 1 and 3 and $\mathbb{Z}/9$ summands in degrees 5 and 7.

5.1.2. Class D, spin-1/2 case. In this case, the symmetries mix as specified by the group extension giving the binary tetrahedral group.

Theorem 5.6. The A_4 -equivariant phase homology group for the class D, spin-1/2 symmetry type in 3d is trivial.

Proof. Let $f_{1/2}^D$ denote the local system on \mathbb{R}^3 assigned to this symmetry type. Since V_{λ} is not pin⁻ (if it were, it would be pin⁻ and orientable, hence spin), Theorem 2.11 says $Ph_0^{A_4}(\mathbb{R}^3; f_{1/2}^D) \cong [MTSpin \wedge (BA_4)_+, \Sigma^5 I_{\mathbb{Z}}]$. Bruner-Greenlees [BG10, §7.7.E] show $ko_4(BA_4) \cong \mathbb{Z}$ and $ko_5(BA_4)$ is torsion, so this phase homology group vanishes.

5.1.3. Class A. Let f_0^A and $f_{1/2}^A$ be the A_4 -equivariant local systems of symmetry types for spinless, resp. spin-1/2 fermions in class A.

Lemma 5.7. $V_{\lambda} \rightarrow BA_4$ is not pin^c.

Proof. If $\beta: H^2(-;\mathbb{Z}/2) \to H^3(-;\mathbb{Z})$ denotes the integral Bockstein, we want to show $\beta w_2(V_\lambda) \neq 0$. By Lemma 3.28, it suffices to show $\operatorname{Sq}^1(w_2(V_\lambda)) \neq 0$. Lemma 5.3 gives $w_2(V_\lambda) = b$, and $\operatorname{Sq}^1b = ab + c$.

Therefore for spin-1/2 fermions, Theorem 2.24 computes $Ph_*^{A_4}(\mathbb{R}^3; f_{1/2}^A)$ in terms of the spin^c bordism of $(BA_4)^{\text{Det}(V_\lambda)-1}$. Since V_λ is orientable, this is isomorphic to the spin^c bordism of BA_4 . For spinless fermions, we use $(BA_4)^{3-V_\lambda}$, as usual.

Theorem 5.8. The low-degree spin^c bordism groups of $X := (BA_4)^{3-V_{\lambda}}$ and BA_4 are

$$\begin{split} \widetilde{\Omega}_{0}^{\mathrm{Spin}^{c}}(X) &\cong \mathbb{Z} & \Omega_{0}^{\mathrm{Spin}^{c}}(BA_{4}) \cong \mathbb{Z} \\ \widetilde{\Omega}_{1}^{\mathrm{Spin}^{c}}(X) &\cong \mathbb{Z}/3 & \Omega_{1}^{\mathrm{Spin}^{c}}(BA_{4}) \cong \mathbb{Z}/3 \\ \widetilde{\Omega}_{2}^{\mathrm{Spin}^{c}}(X) &\cong \mathbb{Z} & \Omega_{2}^{\mathrm{Spin}^{c}}(BA_{4}) \cong \mathbb{Z} \oplus \mathbb{Z}/2 \\ \widetilde{\Omega}_{3}^{\mathrm{Spin}^{c}}(X) &\cong \mathbb{Z}/6 \oplus \mathbb{Z}/3 & \Omega_{3}^{\mathrm{Spin}^{c}}(BA_{4}) \cong \mathbb{Z}/6 \oplus \mathbb{Z}/3 \\ \widetilde{\Omega}_{4}^{\mathrm{Spin}^{c}}(X) &\cong \mathbb{Z}^{2} & \Omega_{4}^{\mathrm{Spin}^{c}}(BA_{4}) \cong \mathbb{Z}^{2}, \end{split}$$

and in both cases, $\Omega_5^{\text{Spin}^c}$ is torsion. Hence both $Ph_0^{A_4}(\mathbb{R}^3; f_0^A)$ and $Ph_0^{A_4}(\mathbb{R}^3; f_{1/2}^A)$ vanish.

Proof. We use the equivalence $MTSpin^c \simeq ku \vee \Sigma^4 ku$ in degrees below 8, then the Adams spectral sequence over $\mathcal{E}(1)$ to compute ku-homology at the prime 2.

For the case of spinless fermions, use the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(X;\mathbb{Z}/2)$ from (5.5) (drawn in Figure 13, left) to compute that the $\mathcal{E}(1)$ -module structure is

(5.9)
$$\widetilde{H}^*(X;\mathbb{Z}/2) \cong \mathcal{O} \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^5 \mathcal{E}(1) \oplus \Sigma^5 \mathcal{E}(1) \oplus P$$

where P is 6-connected. We draw this in Figure 14, left. We computed $\operatorname{Ext}_{\mathcal{E}(1)}^{s,t}(\mathcal{O}, \mathbb{Z}/2)$ in (4.55), and P is too high-degree to be relevant to us, so the E_2 -page of the Adams spectral sequence for $\widetilde{ku}_*(X)$ is given in Figure 14, right. Margolis' theorem (Theorem 3.22) implies this spectral sequence collapses and there are no extension problems, so we conclude.



FIGURE 14. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*((BA_4)^{3-V_{\lambda}}; \mathbb{Z}/2)$ in low degrees. The picture includes all elements in degrees 6 and below. Right: $\operatorname{Ext}_{\mathcal{E}(1)}^{s,t}(\widetilde{H}^*((BA_4)^{3-V_{\lambda}}; \mathbb{Z}/2), \mathbb{Z}/2),$ the E_2 -page of the Adams spectral sequence for $\widetilde{ku}_*((BA_4)^{3-V_{\lambda}})$.

On to the spin-1/2 case. As before, $ku_*(BA_4)$ splits as $ku_*(\text{pt}) \oplus \widetilde{ku}_*(BA_4)$, and we focus on the latter. Bruner-Greenlees [BG03, §2.6] show that 2-locally, there is an equivalence

(5.10)
$$ku \wedge BA_4 \simeq (ku \wedge \Sigma^2 BC_2) \vee \bigvee_{\alpha} \Sigma^{n_{\alpha}} H\mathbb{Z}/2$$

for some collection of integers α ; moreover, their calculation of $ku^*(BA_4)$ [BG03, Theorem 2.6.3] implies the only $n_{\alpha} < 8$ (i.e. the ones relevant for us) are $n_1 = 2$ and $n_2 = 6$. This, together with Hashimoto's computation of $\tilde{ku}_*(B\mathbb{Z}/2)$ [Has83, Theorem 3.1], tells us $ku_*(BA_4)_2^{\wedge}$ in the degrees we need.

We still need to determine the odd-primary torsion.

Lemma 5.11. Let p be an odd prime; then, the inclusion $\mathbb{Z}/3 \hookrightarrow A_4$ sending a generator to $(1\ 2\ 3)$ induces a p-primary stable equivalence $\Sigma^{\infty}(B\mathbb{Z}/3)_+ \to \Sigma^{\infty}(BA_4)_+$.

Proof. Since $|A_4| = 2^2 \cdot 3$, for any $p \ge 5$, the maps $BA_4 \to \text{pt}$ and $B\mathbb{Z}/3 \to \text{pt}$ are *p*-local stable equivalences, so we only have to address p = 3. In this case, Lemma 3.27 implies the inclusion $j: \mathbb{Z}/3 \to A_4$ as the subgroup generated by (1 2 3) induces an isomorphism $H^*(BA_4; \mathbb{Z}/3) \to H^*(B\mathbb{Z}/3; \mathbb{Z}/3)$, so we conclude by the mod *p* Whitehead theorem [Ser53, Chapitre III, Théorème 3].

The Thom isomorphism theorem then implies $\widetilde{H}^*(X; \mathbb{Z}[1/2]) \to \widetilde{H}^*((B\mathbb{Z}/3)^{3-j^*V_{\lambda}}; \mathbb{Z}[1/2])$ is an isomorphism, so arguing in a similar way, there is a *p*-primary stable equivalence $X_{\lambda} \simeq (B\mathbb{Z}/3)^{3-V_{\lambda}}$. Thus, for the purpose of computing the odd-torsion subgroups of $\Omega_*^{\text{Spin}^c}(BA_4)$ and $\widetilde{\Omega}_*^{\text{Spin}^c}(X)$, we can just work with $\mathbb{Z}/3$.

As a $\mathbb{Z}/3$ -representation, j^*V_{λ} is isomorphic to the direct sum of a trivial representation and the real 2-dimensional representation given by rotation. Each of these is spin^c, the latter because it is unitary, so there is a Thom isomorphism $MTSpin^c \wedge (B\mathbb{Z}/3)^{3-j^*V_{\lambda}} \cong MTSpin^c \wedge (B\mathbb{Z}/3)_+$, so in both the spinless and spin-1/2 cases, we just need the 3-torsion in $\Omega_*^{\text{Spin}^c}(B\mathbb{Z}/3)$, which we computed in Theorem 4.14.

5.2. **Pyritohedral symmetry.** Pyritohedral symmetry is the action of $G := A_4 \times \mathbb{Z}/2$ on \mathbb{R}^3 in which A_4 acts as the orientation-preserving symmetries of a tetrahedron and $\mathbb{Z}/2$ acts through inversion; let λ denote this representation and $V_{\lambda} \to BG$ be the associated vector bundle. Because G splits as a direct product, it is easier to analyze than full tetrahedral symmetry (i.e. chiral tetrahedral symmetry together with a reflection), as we will see in this and the next section.

5.2.1. Spinless case. Let $X := (BG)^{3-V_{\lambda}}$. By the twisted Künneth formula, $H^*(X)$ is 2-torsion; therefore $\widetilde{\Omega}_*^{\text{Spin}}(X)$ also lacks odd-primary torsion. so we just have to work with the Adams spectral sequence at p = 2. In the rest of this section, all cohomology is with $\mathbb{Z}/2$ coefficients unless otherwise stated.

Proposition 5.12. The first several spin bordism groups of $(BG)^{3-V_{\lambda}}$ are

$$\begin{split} &\widetilde{\Omega}_{0}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong \mathbb{Z}/2 \\ &\widetilde{\Omega}_{1}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong 0 \\ &\widetilde{\Omega}_{2}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong \mathbb{Z}/2 \\ &\widetilde{\Omega}_{3}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong \mathbb{Z}/2 \\ &\widetilde{\Omega}_{4}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong (\mathbb{Z}/2)^{\oplus 3} \\ &\widetilde{\Omega}_{5}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong (\mathbb{Z}/2)^{\oplus 3} \\ &\widetilde{\Omega}_{6}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong \mathbb{Z}/16 \oplus (\mathbb{Z}/2)^{\oplus 2} \\ &\widetilde{\Omega}_{7}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong (\mathbb{Z}/2)^{\oplus 2}. \end{split}$$

Proof. We employ a trick to reduce the amount of direct computations. We will replace $(3 - V_{\lambda}) \rightarrow BG$ with a virtual vector bundle $E \rightarrow BG$ with the same first two Stiefel-Whitney classes, but which splits as an exterior sum over BA_4 and $B\mathbb{Z}/2$. The Thom spectrum $(BG)^E$ has two nice properties: the Adams E_2 -page for calculating $\tilde{ko}_*((BG)^E)$ is isomorphic to that of $\tilde{ko}_*(X)$, but $(BG)^E$ also splits as a smash product of Thom spectra over BA_4 and $B\mathbb{Z}/2$, simplifying the calculation of said E_2 -page. Because we do not construct a map from $\tilde{ko}_*((BG)^E)$ to $\tilde{ko}_*((BG)^{3-V_{\lambda}})$ or vice versa, this isomorphism does not allow us to deduce any differentials, but we will see that all differentials in range vanish for formal reasons, so this is no problem.

The Künneth formula and Proposition 5.1 together imply

(5.13)
$$H^*(BG) \cong \mathbb{Z}/2[x, u, v, w]/(u^3 + v^2 + w^2 + vw),$$

where |x| = 1, |u| = 2, and |v| = |w| = 3, and that $Sq(x) = x + x^2$ and the Steenrod squares of u, v, and w are as in Proposition 5.1.

Lemma 5.14. The first two Steifel-Whitney classes of V are $w_1(V_{\lambda}) = x$ and $w_2(V_{\lambda}) = u + x^2$.

Proof. Since this representation contains orientation-reversing symmetries, $w_1(V_{\lambda})$ must be nonzero, so is x. For w_2 , we saw in Lemma 5.3 that when one restricts to $A_4 \subset A_4 \times \mathbb{Z}/2$, one has $w_2(V_{\lambda}|_{BA_4}) = u$; when one restricts to $\mathbb{Z}/2$, this is 3 copies of the sign representation, hence has $w_2(V_{\lambda}|_{B\mathbb{Z}/2}) = x^2$. Let $E \to BG$ be the virtual vector bundle

$$(5.15) E \coloneqq 4 - (V_{\lambda}|_{BA_4} \boxplus -\sigma),$$

where $\sigma \to B\mathbb{Z}/2$ is the tautological line bundle. The Whitney sum formula implies for $i = 1, 2, w_i(E) = w_i(3 - V_\lambda)$. Feeding this to the Thom isomorphism gives isomorphisms of $\mathcal{A}(1)$ -modules

(5.16)
$$\widetilde{H}^*((BG)^{3-V_{\lambda}}) \cong \widetilde{H}^*((BG)^E)$$

hence also isomorphisms of the E_2 -pages of the corresponding Adams spectral sequences. Because $E \to BG$ is an external sum,

(5.17)
$$(BG)^E \simeq (BA_4)^{3-V_\lambda} \wedge (B\mathbb{Z}/2)^{\sigma-1}$$

We know the $\mathcal{A}(1)$ -module structures on the low-degree cohomology of both summands, and the Künneth formula tells us to tensor them together (over $\mathbb{Z}/2$) to determine the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*((BG)^E)$.

In (5.5), we computed the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*((BA_4)^{3-V_\lambda})$ in low degrees, and split off two $\Sigma^k \mathcal{A}(1)$ summands. Margolis' theorem (Theorem 3.22) promotes that to a splitting of spectra

(5.18)
$$ko \wedge (BA_4)^{3-V_{\lambda}} \simeq \Sigma^3 H\mathbb{Z}/2 \vee \Sigma^5 H\mathbb{Z}/2 \vee Y,$$

such that as an \mathcal{A} -module,

(5.19)
$$\widetilde{H}^*(Y) \cong \mathcal{A} \otimes_{\mathcal{A}(1)} (\mathcal{O} \oplus P),$$

where P is 7-connected. When we smash $(B\mathbb{Z}/2)^{\sigma-1}$ back in, each $\Sigma^k H\mathbb{Z}/2 \wedge (B\mathbb{Z}/2)^{\sigma-1}$ contributes a summand of $\widetilde{H}_{n-k}((B\mathbb{Z}/2)^{\sigma-1})$ to $\widetilde{ko}_n((BG)^E)$, i.e. a $\mathbb{Z}/2$ -summand in each degree $\ell \geq k$. The upshot for $\mathcal{A}(1)$ -modules is

(5.20)
$$\Sigma^{k} \mathcal{A}(1) \otimes_{\mathbb{Z}/2} \widetilde{H}^{*}((B\mathbb{Z}/2)^{\sigma-1}) \cong \bigoplus_{\ell \ge k} \Sigma^{\ell} H\mathbb{Z}/2.$$

By (5.16), these summands are also present in $\widetilde{H}^*((BG)^{3-V_{\lambda}})$, and Margolis' theorem lifts this to split off corresponding $\Sigma^{\ell} H\mathbb{Z}/2$ summands. Therefore there is a spectrum Y' such that

(5.21)
$$\widetilde{ko}_n((BG)^E) \cong \pi_n(Y') \oplus \widetilde{H}_{n-3}((B\mathbb{Z}/2)^{\sigma-1}) \oplus \widetilde{H}_{n-5}((B\mathbb{Z}/2)^{\sigma-1})$$

and as \mathcal{A} -modules,

(5.22)
$$\widetilde{H}^*(Y') \cong \mathcal{A} \otimes_{\mathcal{A}(1)} (\mathcal{O} \oplus P) \otimes_{\mathbb{Z}/2} \widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1}).$$

The change-of-rings theorem (3.4) thus applies to the E_2 -page of the Adams spectral sequence calculating $\pi_*(Y')$, yielding

(5.23)
$$E_2^{s,t} \cong \operatorname{Ext}_{\mathcal{A}(1)}^{s,t}((\mathcal{O} \oplus P) \otimes_{\mathbb{Z}/2} \widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1}), \mathbb{Z}/2).$$

We will work with this spectral sequence, adding in the summands corresponding to $\Sigma^3 H\mathbb{Z}/2$ and $\Sigma^5 H\mathbb{Z}/2$ afterwards.

Our first order of business is to compute the tensor product in (5.23). The $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1})$ can be found in [BC18, Figure 4].

Lemma 5.24. There is an isomorphism of $\mathcal{A}(1)$ -modules $\mathcal{O} \otimes_{\mathbb{Z}/2} \widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1}) \cong \mathcal{A}(1) \oplus \Sigma^2 R_0 \oplus \Sigma^4 \mathcal{A}(1) \oplus P$, where P is 7-connected.

Proof. Compute directly, by hand or by computer.

By (5.21) and (5.22), we can work with (5.23), then add in the $\mathbb{Z}/2$ summands coming from the $\Sigma^k H\mathbb{Z}/2$ summands at the end. Lemma 5.24 tells us the E_2 -page of (5.23) is

(5.25)
$$E_2^{s,t} \cong \operatorname{Ext}(\mathcal{A}(1) \oplus \Sigma^2 R_0 \oplus \Sigma^4 \mathcal{A}(1) \oplus P).$$

Since P is 7-connected, its Ext is concentrated in degrees irrelevant to us, and we ignore it. $Ext(\Sigma^2 R_0)$ is computed in the degrees we need by Beaudry-Campbell [BC18, Figures 23, 24]; using this, the E_2 -page

of (5.23) is



Margolis' theorem and h_1 -equivariance of differentials immediately imply there are no nontrivial differentials or extension problems below degree 8, so we conclude.

5.2.2. Class D, spin-1/2 case. Let $f_{1/2}^D$ denote the equivariant local system of symmetry types corresponding to spin-1/2 fermions for a pyritohedral symmetry in class D. Theorem 2.11 computes the equivariant phase homology associated to $f_{1/2}^D$ in terms of the spin bordism of $X := (BA_4 \times B\mathbb{Z}/2)^{\text{Det}(V_\lambda)-1}$. The isomorphism $\text{Det}(V_\lambda) \cong 0 \boxplus \sigma$ provides an isomorphism $X \simeq (BA_4)_+ \wedge (B\mathbb{Z}/2)^{\sigma-1}$, Lemma 3.30 thus implies the spin bordism of this spectrum computes the pin⁻ bordism of BA_4 , which could be independently interesting.

Theorem 5.27. The first few spin bordism groups of X are

$$\begin{split} \widetilde{\Omega}_{0}^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_{1}^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_{2}^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/8 \oplus \mathbb{Z}/2\\ \widetilde{\Omega}_{3}^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/2 \oplus \mathbb{Z}/2\\ \widetilde{\Omega}_{4}^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/2. \end{split}$$

Since $\widetilde{\Omega}_5^{\mathrm{Spin}}(X)$ is torsion by Lemma 3.24, $Ph_0^{A_4 \times \mathbb{Z}/2}(\mathbb{R}^3; f_{1/2}^D) \cong \mathbb{Z}/2.$

Proof. By the twisted Künneth formula, $H^*(X)$ has no odd-primary torsion, and therefore neither does $\widetilde{\Omega}^{\text{Spin}}_*(X)$, so it suffices to work at the prime 2, which we do.

Use Lemma 3.30 to split $X \simeq (B\mathbb{Z}/2)^{\sigma-1} \vee M$, where the map $\widetilde{H}^*(M;\mathbb{Z}/2) \to \widetilde{H}^*(X;\mathbb{Z}/2)$ is injective with image a complimentary subspace to $\mathbb{Z}/2 \cdot \{Ux^k \mid k \ge 0\}$.

As usual, $w_1(\text{Det}(V_\lambda) - 1) = w_1(V_\lambda) = x$ and $w_2(\text{Det}(V_\lambda) - 1) = 0$. We also need to know the \mathcal{A} -action on $H^*(BG; \mathbb{Z}/2)$; the Künneth formula determines this using as input the \mathcal{A} -action on $H^*(BA_4; \mathbb{Z}/2)$, which we computed in Proposition 5.1, and the \mathcal{A} -action on $H^*(B\mathbb{Z}/2; \mathbb{Z}/2)$, which is standard. Using this, we can determine the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(M; \mathbb{Z}/2)$. We obtain an isomorphism of $\mathcal{A}(1)$ -modules

(5.28)
$$\widetilde{H}^*(M;\mathbb{Z}/2) \cong \Sigma^2 R_3 \oplus \Sigma^3 \mathcal{A}(1) \oplus \Sigma^3 \mathcal{A}(1) \oplus \Sigma^4 \mathcal{A}(1) \oplus P,$$

where P is 4-connected. We will see in Figure 15, right, that for $t - s \leq 4$, $E_2^{s,t}$ is concentrated in Adams filtration 0; this and the 4-connectedness of P imply its contribution to the E_2 -page cannot affect the spectral sequence in degrees $t - s \leq 4$, which is all we need. We draw these summands, except for P, in Figure 15, left.

Freed-Hopkins [FH16a, Figure 5, case s = 3] and Beaudry-Campbell [BC18, Figures 32, 33] calculate $Ext(R_3)$ in the range we need, and we can draw the E_2 -page of the Adams spectral sequence in Figure 15, right. This collapses, so we add in the pin⁻ bordism summands we need from [ABP69, KT90b] to obtain the groups in the theorem.

5.2.3. Class A, spinless case. Let f_0^A denote the equivariant local system of symmetry types corresponding to spinless fermions in class A and $X \coloneqq (BA_4 \times B\mathbb{Z}/2)^{3-V_{\lambda}}$; then we saw that $Ph_0^{A_4 \times \mathbb{Z}/2}(\mathbb{R}^3; f_0^A)$ is determined by $\widetilde{\Omega}_*^{\text{spin}^c}(X)$.



FIGURE 15. Left: the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(M; \mathbb{Z}/2)$ in low degrees. This picture includes all summands in degrees 4 and below. Here $\alpha := u^2 x^2 + v^2 + w^2$. Right: the E_2 -page of the corresponding Adams spectral sequence.

Theorem 5.29. The first few spin^c bordism groups of X are

$$\begin{split} &\widetilde{\Omega}_{0}^{\mathrm{Spin}^{c}}(X) \cong \mathbb{Z}/2 \\ &\widetilde{\Omega}_{1}^{\mathrm{Spin}^{c}}(X) \cong 0 \\ &\widetilde{\Omega}_{2}^{\mathrm{Spin}^{c}}(X) \cong (\mathbb{Z}/2)^{\oplus 2} \\ &\widetilde{\Omega}_{3}^{\mathrm{Spin}^{c}}(X) \cong \mathbb{Z}/2 \\ &\widetilde{\Omega}_{4}^{\mathrm{Spin}^{c}}(X) \cong \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 3} \\ &\widetilde{\Omega}_{5}^{\mathrm{Spin}^{c}}(X) \cong (\mathbb{Z}/2)^{\oplus 3} \\ &\widetilde{\Omega}_{6}^{\mathrm{Spin}^{c}}(X) \cong \mathbb{Z}/8 \oplus (\mathbb{Z}/2)^{\oplus 6} \\ &\widetilde{\Omega}_{7}^{\mathrm{Spin}^{c}}(X) \cong (\mathbb{Z}/2)^{\oplus 5}, \end{split}$$

so $Ph_0^{A_4 \times \mathbb{Z}/2}(\mathbb{R}^3; f_0^A) \cong \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 3}$.

Proof. The twisted Thom isomorphism and twisted Künneth formula imply $\widetilde{H}^*(X;\mathbb{Z})$ is 2-torsion. Therefore for any odd prime p, the mod p Whitehead theorem [Ser53, Chapitre III, Théorème 3] implies $\widetilde{\Omega}_*^{\text{spin}^c}(X)$ also has no p-torsion. This leaves only p = 2, for which we use the Adams spectral sequence over $\mathcal{E}(1)$.

We determined the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*((BG)^{3-V_{\lambda}})$ as given in (5.25), together with an $\Sigma^{\ell}\mathcal{A}(1)$ for $\ell = 3, 4$, and two $\Sigma^{\ell}\mathcal{A}(1)$ summands for $\ell \geq 5$. This determines the $\mathcal{E}(1)$ -module structure: as $\mathcal{E}(1)$ -modules, $\mathcal{A}(1) \cong \mathcal{E}(1) \oplus \Sigma^2 \mathcal{E}(1)$, and $R_0 \cong H$, so

(5.30)
$$H^*(X;\mathbb{Z}/2) \cong \Sigma^2 H \oplus V' \otimes_{\mathbb{Z}/2} \mathcal{E}(1) \oplus P,$$

where V' is a graded $\mathbb{Z}/2$ -vector space with a basis of homogeneous elements in degrees 0, 2, 3, 4, 4, 5, 5, 5, 6, 6, 6, 7, 7, 7, and 7, and P is 7-connected. Therefore the E_2 -page is



By Margolis' theorem, there are no nontrivial differentials or extension problems in this range.

5.2.4. Class A, spin-1/2 case. To compute the $(A_4 \times \mathbb{Z}/2)$ -equivariant phase homology groups for the local system $f_{1/2}^A$ specified by the spin-1/2 extension in class A Theorem 2.24 asks us to investigate the spin^c bordism of $X := (BA_4 \times B\mathbb{Z}/2)^{\text{Det}(V_{\lambda})-1} \simeq (BA_4 \times B\mathbb{Z}/2)^{0 \oplus \sigma-1}$; we know V_{λ} is not pin^c because we saw in Lemma 5.7 that the pullback of V_{λ} along $BA_4 \to BA_4 \times B\mathbb{Z}/2$ is not pin^c.

Theorem 5.32. The first few spin^c bordism groups of X are

$$\widetilde{\Omega}_{0}^{\text{Spin}^{c}}(X) \cong \mathbb{Z}/2$$

$$\widetilde{\Omega}_{1}^{\text{Spin}^{c}}(X) \cong 0$$

$$\widetilde{\Omega}_{2}^{\text{Spin}^{c}}(X) \cong \mathbb{Z}/4 \oplus \mathbb{Z}/2$$

$$\widetilde{\Omega}_{3}^{\text{Spin}^{c}}(X) \cong (\mathbb{Z}/2)^{\oplus 2}$$

$$\widetilde{\Omega}_{4}^{\text{Spin}^{c}}(X) \cong \mathbb{Z}/8 \oplus (\mathbb{Z}/2)^{\oplus 3}$$

By Lemma 3.24, $\widetilde{\Omega}_5^{\mathrm{Spin}^c}(X)$ is torsion, so $Ph_0^{A_4 \times \mathbb{Z}/2}(\mathbb{R}^3; f_{1/2}^A) \cong \mathbb{Z}/8 \oplus (\mathbb{Z}/2)^{\oplus 3}$.

Proof. We reuse our work from §5.2.2. We saw that $X \simeq (B\mathbb{Z}/2)^{\sigma-1} \vee M$, and we gave the low-degree cohomology of M as an $\mathcal{A}(1)$ -module in (5.28) (and drew it in Figure 15, left). This determines the $\mathcal{E}(1)$ -module structure on it, so we can calculate spin^c bordism of M using the Adams spectral sequence. For the other summand, we have $MTSpin^c \wedge (B\mathbb{Z}/2)^{\sigma-1} \simeq MTPin^c$, so we direct-sum in the pin^c bordism groups computed by Bahri-Gilkey [BG87a, BG87b].

There are isomorphisms of $\mathcal{E}(1)$ -modules $\mathcal{A}(1) \cong \mathcal{E}(1) \oplus \Sigma^2 \mathcal{E}(1)$ and $R_3 \cong \Sigma^2 \mathcal{E}(1) \oplus \Sigma^4 R_0$. Therefore as an $\mathcal{E}(1)$ -module,

(5.33)
$$\widetilde{H}^*(M;\mathbb{Z}/2) \cong \Sigma^2 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^4 R_0 \oplus \Sigma^4 \mathcal{E}(1) \oplus P,$$

where P is 4-connected. As usual for these cases, we will see that $\operatorname{Ext}(\tilde{H}^*(M;\mathbb{Z}/2),\mathbb{Z}/2)$ has no nonzero elements with t-s=4 and s>1, so P does not affect our calculations. See Figure 16, left, for a picture of the $\mathcal{E}(1)$ -module structure on $\tilde{H}^*(M;\mathbb{Z}/2)$. Look up $\operatorname{Ext}(\Sigma^4 R_0)$ in Proposition 4.48 to obtain the E_2 -page of the Adams spectral sequence as in Figure 16, right. This collapses, so we add in the pin^c bordism summands and conclude.

I could get used to Adams spectral sequences like this one. But alas, they are not all this easy, as we will see in the next section.

5.3. Full tetrahedral symmetry. The full group of symmetries of the tetrahedron, including reflections, is the symmetric group S_4 , acting via the representation $\lambda: S_4 \to O_3$, which is isomorphic to the quotient of the four-dimensional real permutation representation by the fixed line $\mathbb{R} \cdot (1, 1, 1, 1)$.

Proposition 5.34 ([Ngu09, §2.3]). $H^*(BS_4; \mathbb{Z}/2) \cong \mathbb{Z}/2[a, b, c]/(ac)$, with |a| = 1, |b| = 2, and |c| = 3. The Steenrod squares of the generators are $Sq(a) = a + a^2$, $Sq(b) = b + ab + c + b^2$, and $Sq(c) = c + bc + c^2$.²⁷

²⁷The ring structure on $H^*(BS_4; \mathbb{Z}/2)$ was known earlier, due to Cardenas [Car65]; see [AM04, Example VI.1.13].



FIGURE 16. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*(M; \mathbb{Z}/2)$ in low degrees. This picture includes all summands in degrees 4 and below. Here $\alpha \coloneqq u^2 + vx + wx$. Right: the E_2 -page of the corresponding Adams spectral sequence.

Let $V_{\lambda} \to BS_4$ denote the associated vector bundle to λ .

Proposition 5.35. $w_1(V_{\lambda}) = a$, $w_2(V_{\lambda}) = b$, and $w_3(V_{\lambda}) = c$.

Proof. Since λ does not factor through $SO_3 \subset O_3$, V_λ is unorientable. Thus $w_1(V_\lambda) \neq 0$, and a is the only nonzero element of $H^1(BS_4; \mathbb{Z}/2)$, so $w_1(V_\lambda) = a$. For w_2 , we calculated in Lemma 5.3 that $w_2(V_\lambda|_{A_4}) \neq 0$, so w_2 cannot vanish in BS_4 . Our options are a^2 , b, and a^2+b . Let $\mathbb{Z}/2 \subset S_4$ be generated by a transposition; then as a $\mathbb{Z}/2$ -representation $\lambda \cong \mathbb{R}^2 \oplus \sigma$, so $w_2(V_\lambda|_{\mathbb{Z}/2}) = 0$. The map $H^*(BS_4; \mathbb{Z}/2) \to H^*(B\mathbb{Z}/2; \mathbb{Z}/2) \cong \mathbb{Z}/2[x]$ sends $b, c \mapsto 0$ and $a \mapsto x$, so the constraint $w_2(V_\lambda|_{\mathbb{Z}/2}) = 0$ rules out $w_2(V_\lambda) = a^2$ and $w_2(V_\lambda) = a^2 + b$, forcing us to conclude $w_2(V_\lambda) = b$. Finally, $w_3(V_\lambda) = c$ follows from the Wu formula.

We need the next calculation to determine the odd-primary torsion subgroups of the phase homology groups that we calculate.

Lemma 5.36. Suppose $V \to BS_4$ is a rank-zero virtual vector bundle with $w_1(V) = x$. Then the inclusion $i: S_3 \hookrightarrow S_4$ defines an isomorphism

(5.37)
$$\widetilde{H}_*((BS_3)^{i^*V}) \otimes \mathbb{Z}[1/2] \to \widetilde{H}_*((BS_4)^V) \otimes \mathbb{Z}[1/2].$$

Proof. The commutative diagram of short exact sequences



induces a map between their Lyndon-Hochschild-Serre spectral sequences with signatures

(5.39)
$$E_{p,q}^2 = H_p(B\mathbb{Z}/2; \underline{H_q(BA_k; \mathbb{Z}[1/2])} \otimes (\mathbb{Z}[1/2])_x) \Longrightarrow H_{p+q}(BS_k; (\mathbb{Z}[1/2])_{w_1(V)}),$$

where $\underline{H}_q(BA_4; \mathbb{Z}[1/2])$ means the local system on $B\mathbb{Z}/2$ induced by the action of $\mathbb{Z}/2$ on A_k as specified by the extension $1 \to A_k \to S_k \to \mathbb{Z}/2 \to 1$, and x is the generator of $H^1(B\mathbb{Z}/2; \mathbb{Z}/2)$.

We claim the map on these spectral sequences is an isomorphism on E^2 -pages. By Lemma 5.11, the map $H_*(BA_4; \mathbb{Z}[1/2]) \to H_*(BA_3; \mathbb{Z}[1/2])$ is an isomorphism, and this isomorphism intertwines the $\mathbb{Z}/2$ -actions on $H_*(BA_k; \mathbb{Z}[1/2]) \otimes (\mathbb{Z}[1/2])_x$, because (5.38) commutes. Therefore it induces an isomorphism on all E^r -pages, hence also on what these spectral sequences converge to.

ARUN DEBRAY

The top row in (5.38) can be identified with $1 \to \mathbb{Z}/3 \to D_6 \to \mathbb{Z}/2 \to 1$, so by the same lines of reasoning as in Propositions 4.23 and 4.24 we deduce

(5.40a)
$$\widetilde{\Omega}_{k}^{\mathrm{Spin}}((BS_{4})^{V}) \otimes \mathbb{Z}[1/2] \cong \begin{cases} \mathbb{Z}/3, & k = 1\\ 0, & k = 0, 2, 3, 4 \end{cases}$$

(5.40b)
$$\widetilde{\Omega}_{k}^{\text{Spin}^{c}}((BS_{4})^{V}) \otimes \mathbb{Z}[1/2] \cong \begin{cases} \mathbb{Z}/3, & k = 1, 3\\ 0, & k = 0, 2, 4. \end{cases}$$

5.3.1. Class D, spinless case. As usual in the spinless case for unorientable representations, the ansatz asks us to let $X := (BS_4)^{3-V_{\lambda}}$ and consider $MTSpin \wedge X$.

Theorem 5.41. The first few spin bordism groups of X are

$$\begin{split} \widetilde{\Omega}_0^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_1^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/3\\ \widetilde{\Omega}_2^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_3^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_4^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 2}, \end{split}$$

and $\widetilde{\Omega}_5^{\text{Spin}}(X)$ is torsion,

Proof. For odd-primary information, see Equation (5.40a). For 2-primary information, we will again use the Adams spectral sequence over $\mathcal{A}(1)$. Our first task is to write down $\widetilde{H}^*(X;\mathbb{Z}/2)$ as an $\mathcal{A}(1)$ -module in low degrees, using Proposition 5.35 to deduce $w_1(3 - V_{\lambda}) = a$ and $w_2(3 - V_{\lambda}) = a^2 + b$. We describe this $\mathcal{A}(1)$ -module structure in low degrees in Figure 17, left.



FIGURE 17. Left: the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*((BS_4)^{3-V_{\lambda}}; \mathbb{Z}/2)$ in low degrees. This submodule contains all elements of degree at most 7. Right: the E_2 -page of the Adams spectral sequence computing $\widetilde{ko}_*((BS_4)^{3-V_{\lambda}})$.

Let $\Sigma^4 N_2$ denote the submodule generated by Ub^2 and Ubc, which is a nontrivial extension of J by ΣJ .²⁸ Then there is an isomorphism

(5.42)
$$\widetilde{H}^*(X;\mathbb{Z}/2) \cong \mathcal{A}(1) \oplus \Sigma^2 N_1 \oplus \Sigma^4 \mathcal{A}(1) \oplus \Sigma^4 N_2 \oplus \Sigma^6 \mathcal{A}(1) \oplus P,$$

²⁸We propose calling N_2 the *butterfly*; it also appears in [WWZ20, Figure 16].

The indecomposable summand isomorphic to $\Sigma^2 N_1$ is generated by Ub, and P has no elements in degrees below 8, and therefore is irrelevant for our low-degree computations. As before, we know what a $\Sigma^k \mathcal{A}(1)$ summand contributes to the E_2 -page. To compute $\text{Ext}(N_1)$, we use a well-known explicit (12-shifted) 4-periodic minimal resolution²⁹

(5.43)

$$N_1 \stackrel{f_0}{\leftarrow} \mathcal{A}(1) \stackrel{f_1}{\leftarrow} \Sigma^2 \mathcal{A}(1) \oplus \Sigma^3 \mathcal{A}(1) \stackrel{f_2}{\leftarrow} \Sigma^4 \mathcal{A}(1) \oplus \Sigma^5 \mathcal{A}(1) \stackrel{f_3}{\leftarrow} \Sigma^7 \mathcal{A}(1) \stackrel{f_4}{\leftarrow} \Sigma^{12} \mathcal{A}(1) \stackrel{\Sigma^{12} f_1}{\leftarrow} \Sigma^{14} \mathcal{A}(1) \oplus \Sigma^{15} \mathcal{A}(1) \stackrel{\Sigma^{12} f_2}{\leftarrow} \dots$$

The dimension of $\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(N_1, \mathbb{Z}/2)$ is the number of summands of $\Sigma^t \mathcal{A}(1)$ in the sth module in the extension.³⁰ This (shifted up by 2 for $\Sigma^2 N_1$) gives the orange summands in Figure 17, right.

For $\Sigma^4 N_2$, we use a convenient shortcut: the kernel of the map f_2 in (5.43) is isomorphic to $\Sigma^4 N_2$. Thus, the sequence (5.43) except for the first two terms forms a minimal resolution for $\Sigma^4 N_2$, so for every $s, t \ge 0$, there is an isomorphism

(5.45)
$$\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(N_2, \mathbb{Z}/2) \cong \operatorname{Ext}_{\mathcal{A}(1)}^{s+2,t+4}(N_1, \mathbb{Z}/2)$$

equivariant for the $H^{*,*}(\mathcal{A}(1))$ -actions on both sides. This gives us the blue summands in Figure 17, right. Now we can draw the E_2 -page for the Adams spectral sequence for $\widetilde{\Omega}_*^{\text{Spin}}(X)$, and do so in Figure 17, right.

Margolis' theorem and h_i -equivariance of differentials imply there is a single differential in this range that could be nonzero, namely the pictured $d_2: E_2^{0,5} \to E_2^{2,6}$.

Proposition 5.46. $d_2: E_2^{0,5} \to E_2^{2,6}$ vanishes; equivalently, $\widetilde{ko}_4(X)$ has more than eight elements.

We will prove this using the Atiyah-Hirzebruch spectral sequence in Theorem 5.56.

Assuming Proposition 5.46 for now, there are no further differentials in the range we care about, but we must address four extension questions in degrees 4, 5, and 6:

$$(5.47a) 0 \longrightarrow \mathbb{Z}/2 \longrightarrow A \longrightarrow \mathbb{Z}/2 \oplus \mathbb{Z}/2 \longrightarrow 0$$

(5.47b)
$$0 \longrightarrow \mathbb{Z}/2 \longrightarrow \widetilde{ko}_4(X) \longrightarrow A \longrightarrow 0$$

(5.47c)
$$0 \longrightarrow \mathbb{Z}/2 \longrightarrow \widetilde{ko}_5(X) \longrightarrow \mathbb{Z}/2 \longrightarrow 0$$

(5.47d)
$$0 \longrightarrow \mathbb{Z}/2 \longrightarrow \widetilde{ko}_6(X) \longrightarrow \mathbb{Z}/2 \oplus \mathbb{Z}/2 \longrightarrow 0.$$

(In fact, a priori, there are five extension problems, but Margolis' theorem splits $E_{\infty}^{0,6} \cong \mathbb{Z}/2$ off from the rest of the t - s = 6 line.)

Both (5.47a) and (5.47c) split for the same reason. For k = 4, 5, assume the sequence does not split; then, $\tilde{ko}_k(X)$ has an element x such that $2x \neq 0$ and if y is the image of 2x in the E_{∞} -page, then $h_1y \neq 0$. This fact lifts to a nonzero action by $\eta \in ko_1$ carrying 2x to some element $z \in \tilde{ko}_{k+1}(X)$ such that $z = 2\eta x$ and $z \neq 0$, but $2\eta = 0$, causing a contradiction.

Because (5.47a) splits and $(h_0 \cdot) : E_{\infty}^{1,5} \to E_{\infty}^{2,6}$ is an isomorphism, all possible extensions in (5.47b) give $\widetilde{ko}_4(X) \cong \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 2}$.

Lastly, (5.47d). Action by h_1 defines isomorphisms $E_{\infty}^{0,5} \to E_{\infty}^{1,7}$ and $E_{\infty}^{2,7} \to E_{\infty}^{3,9}$, and this lifts to imply $(\eta \cdot) : \tilde{ko}_6(X) \to \tilde{ko}_7(X)$ is injective, splitting (5.47d).

We return to Proposition 5.46. Our proof strategy is to compute $\tilde{ko}_4(X)$ a different way. First, we pass to $\tau_{0:4}ko$ -cohomology, following a strategy of Campbell [Cam17, §7.4] and Freed-Hopkins [FH19a, §5.1], by

$$(5.44a) \qquad \qquad 0 \longrightarrow \Sigma \mathbb{Z}/2 \longrightarrow N_1 \longrightarrow \mathbb{Z}/2 \longrightarrow 0$$

(5.44b)
$$0 \longrightarrow \Sigma^2 N_1 \longrightarrow \hat{\mathcal{O}} \longrightarrow \mathbb{Z}/2 \longrightarrow 0$$

together with the fact that the boundary maps in the long exact sequences commute with the $H^{*,*}(\mathcal{A}(1))$ -action.

²⁹After some practice with $\mathcal{A}(1)$ -modules, writing this minimal resolution down is straightforward, if a little tedious; we found it a helpful exercise when learning this material and the interested reader might too. Though this minimal resolution is certainly known, it is not explicitly written in many places; the resolution will not be televised.

³⁰The $H^{*,*}(\mathcal{A}(1))$ -action on $\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(N_1,\mathbb{Z}/2)$ is a little obscure from this perspective; one can show that all h_0 - and h_1 -actions that could be nonzero for degree reasons are in fact nonzero, as stated in [BB96, §3] and [WWZ20, Figure 15]. One way to see this would be to use the long exact sequences in Ext associated to the two short exact sequences

way of Lemma 5.48. We then run the Atiyah-Hirzebruch spectral sequence computing the $\tau_{0:4}ko$ -cohomology of X. As input, we need $\tilde{H}^*(X;\mathbb{Z})$, which we compute in Theorem 5.49 using a Lyndon-Hochschild-Serre spectral sequence [Lyn48, Ser50, HS53].

Lemma 5.48 (Campbell [Cam17, (7.35), (7.36)]). There is a noncanonical equivalence $I_{\mathbb{Z}}(\tau_{0:4}ko) \simeq \Sigma^{-4}\tau_{0:4}ko$. Thus, if $\tau_{0:4}\tilde{ko}_k(Y)$ is torsion, $\tau_{0:4}\tilde{ko}_k(Y) \cong \tau_{0:4}\tilde{ko}^{k-3}(Y)$.

This is a corollary of the shifted self-equivalence $I_{\mathbb{Z}}KO \simeq \Sigma^4 KO$ [And69, Theorem 4.16].³¹

By Lemma 3.24, $\widetilde{ko}_4(X) \cong \tau_{0:4}\widetilde{ko}_4(X)^{32}$ is torsion, so is isomorphic to $\tau_{0:4}\widetilde{ko}^1(X)$. We study this group with the Atiyah-Hirzebruch spectral sequence. As input, we compute $\widetilde{H}^*(X;\mathbb{Z}_{(2)})$, which the Thom isomorphism equates with $H^*(BS_4;(\mathbb{Z}_{(2)})_{w_1(V_\lambda)})$.

Theorem 5.49.

$$H^{0}(BS_{4}; (\mathbb{Z}_{(2)})_{w_{1}(V_{\lambda})}) \cong 0$$

$$H^{1}(BS_{4}; (\mathbb{Z}_{(2)})_{w_{1}(V_{\lambda})}) \cong \mathbb{Z}/2$$

$$H^{2}(BS_{4}; (\mathbb{Z}_{(2)})_{w_{1}(V_{\lambda})}) \cong 0$$

$$H^{3}(BS_{4}; (\mathbb{Z}_{(2)})_{w_{1}(V_{\lambda})}) \cong \mathbb{Z}/2 \oplus \mathbb{Z}/2$$

$$H^{4}(BS_{4}; (\mathbb{Z}_{(2)})_{w_{1}(V_{\lambda})}) \cong \mathbb{Z}/2$$

$$H^{5}(BS_{4}; (\mathbb{Z}_{(2)})_{w_{1}(V_{\lambda})}) \cong \mathbb{Z}/2 \oplus \mathbb{Z}/2.$$

Proof. Let $R := \mathbb{Z}_{(2)}[x]/(x^2 - 1)$, which is a $\mathbb{Z}[C_2]$ -module in which the nontrivial element of C_2 sends $1 \mapsto 1$ and $x \mapsto -x$. As $\mathbb{Z}[C_2]$ -modules, $R \cong \mathbb{Z}_{(2)} \oplus (\mathbb{Z}_{(2)})_{\sigma}$, so we will recover $H^*(BS_4; (\mathbb{Z}_{(2)})_{w_1(V_{\lambda})})$ from $H^*(BS_4; R)$. The Lyndon-Hochschild-Serre spectral sequence

$$(5.50) E_2^{*,*} = H^*(BC_2; H^*(BA_4; R)) \Longrightarrow H^*(BS_4; R)$$

is multiplicative; here S_4 acts on R through $sign: S_4 \to C_2$ and A_4 acts trivially. R is a $\mathbb{Z}/2$ -graded ring, where x is in odd degree, and hence R-valued cohomology is $\mathbb{Z} \times \mathbb{Z}/2$ -graded. We use $\{+, -\}$ to denote the $\mathbb{Z}/2$ -grading.

Proposition 5.51 (Čadek [Čad99, Lemma 3.1]). There is an isomorphism of $\mathbb{Z} \times \mathbb{Z}/2$ -graded rings $H^*(BC_2; R) \cong \mathbb{Z}_{(2)}[y]/(2y)$ with |y| = (1, -).

Proposition 5.52 (Bruner-Greenlees [BG03, §2.6]). There is a presentation of $H^*(BA_4; \mathbb{Z}_{(2)})$ whose only generators and relations below degree 6 are generators α and β in degrees 3 and 4, respectively, and relations $2\alpha = 2\beta = 0$.

Corollary 5.53. As $\mathbb{Z} \times \mathbb{Z}/2$ -graded rings,

(5.54)
$$H^*(BA_4; R) \cong \mathbb{Z}_{(2)}[\alpha_+, \alpha_-, \beta_+, \beta_-, \dots]/(2\alpha_{\pm}, 2\beta_{\pm}, \dots)$$

where the generators and relations not displayed are in \mathbb{Z} -degrees ≥ 6 , $|\alpha_{\pm}| = (2, \pm)$, and $|\beta_{\pm}| = (3, \pm)$.

³¹Anderson gives this proof in unpublished lecture notes; see Yosimura [Yos75, Theorem 4] for Anderson's proof. There are at least four additional proofs that $I_{\mathbb{Z}}KO \simeq \Sigma^4 KO$, due to Freed-Moore-Segal [FMS07, Proposition B.11], Heard-Stojanoska [HS14, Theorem 8.1], Ricka [Ric16, Corollary 5.8], and Hebestreit-Land-Nikolaus [HLN20, Example 2.8], all by different methods.

³²We abuse notation slightly to let $\tau_{0:4}\tilde{ko}$ denote reduced $\tau_{0:4}ko$ -cohomology, rather than $\tau_{0:4}\tilde{ko}$.

We can now display the E_2 -page. Elements with + grading are colored red, and elements with - grading are colored blue; differentials are even in this $\mathbb{Z}/2$ -grading.



The map $S_4 \rightarrow C_2$ admits a section given by $\{1, (1 \ 2)\} \subset S_4$, so the q = 0 line supports no nonzero differentials and does not participate in nontrivial extension problems. Looking just at elements graded -, we are done if we can show that $d_2(\beta_-) = \alpha_- y^2$ and $d_2(\beta_+ y) = 0$. Fortunately, Thomas [Tho74] has computed $H^*(BS_4;\mathbb{Z}_{(2)})$: since $H^4(BS_4;\mathbb{Z}_{(2)})\cong\mathbb{Z}/4\oplus\mathbb{Z}/2$, $d_2(\beta_+)=0$, so the Leibniz rule implies $d_2(\beta_+y)=0$ too. And since $H^5(BS_4;\mathbb{Z}_{(2)}) \cong \mathbb{Z}/2$, $d_2(\beta_- y) \neq 0$, so $d_2(\beta_-) \neq 0$, hence must be $\alpha_- y^2$.

Thus equipped, we tackle the Atiyah-Hirzebruch spectral sequence.

Theorem 5.56. $|\widetilde{ko}_4(X)| \ge 16$ (thus implying Proposition 5.46).

Proof. After using Lemma 5.48, we want to compute $\tau_{0:4} \tilde{ko}^1(X)$, which we attack with the Atiyah-Hirzebruch spectral sequence

(5.57)
$$E_2^{p,q} = \widetilde{H}^p(X; (\tau_{0:4}ko)^q) \Longrightarrow \tau_{0:4}\widetilde{ko}^{p+q}(X).$$

Using Proposition 5.34 and Theorem 5.49 as input, the E_2 -page is



Maunder [Mau63, Theorem 3.4] identifies the first nonzero differential in the cohomological Atiyah-Hirzebruch spectral sequence with a k-invariant; this includes all differentials shown in (5.58). Let $r: H^*(-;\mathbb{Z}) \to \mathbb{Z}$ $H^*(-;\mathbb{Z}/2)$ denote reduction mod 2 and $\beta: H^*(-;\mathbb{Z}/2) \to H^{*+1}(-;\mathbb{Z})$ be the Bockstein. Then, Bruner-Greenlees [BG10, Corollary A.5.2] determine the k-invariants we need for ko-cohomology:

- (1) The green $d_2: E_2^{p,0} \to E_2^{p+2,-1}$ is $\operatorname{Sq}^2 \circ r: \widetilde{H}^p(X; \mathbb{Z}) \to \widetilde{H}^{p+2}(X; \mathbb{Z}/2).$ (2) Each blue $d_2: E_2^{p,-1} \to E_2^{p+2,-1}$ is $\operatorname{Sq}^2: \widetilde{H}^p(X; \mathbb{Z}/2) \to \widetilde{H}^{p+2}(X; \mathbb{Z}/2).$

(3) Each purple $d_3 \colon E_3^{p,-2} \to E_3^{p+3,-4}$ is $\beta \circ \operatorname{Sq}^2 \colon \widetilde{H}^p(X; \mathbb{Z}/2) \to \widetilde{H}^{p+3}(X; \mathbb{Z}).$

We computed the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(X;\mathbb{Z}/2)$ in (5.42) (and drew it in Figure 17, left), and r and β follow from this and a few facts we just calculated for $\widetilde{H}^*(X;\mathbb{Z})$. For $k \leq 5$, we proved $2\widetilde{H}^k(X;\mathbb{Z}) = 0$, so r is injective in these degrees. Moreover, combining this with Lemma 3.28, that $r \circ \beta = \mathrm{Sq}^1$, we conclude for $k \leq 2$ and $x \in \widetilde{H}^k(X;\mathbb{Z}/2)$, $\beta \mathrm{Sq}^2(x) = 0$ iff $\mathrm{Sq}^1 \mathrm{Sq}^2(x) = 0$.

All together, these allow us to resolve almost all of the indicated differentials — a priori, we do not know $\beta \operatorname{Sq}^2(x)$ when $x \in E_2^{3,-2} \cong \widetilde{H}^3(X; \mathbb{Z}/2)$, but for all x not in the image of $d_2: E_2^{1,-1} \to E_2^{3,-2}$, $\operatorname{Sq}^2(x) = 0$, so this is fine. We find the 1-line of the E_4 -page has five $\mathbb{Z}/2$ summands, one in $E_4^{2,-1}$, two in $E_4^{3,-2}$, and two in $E_4^{5,-4}$. There could be a nonzero $d_4: E_4^{2,-1} \to E_4^{6,-5}$, but the remaining four summands are generated by permanent cycles.

5.3.2. Class D, spin-1/2 case. As V_{λ} is not pin⁻, Theorem 2.11 tells us to compute the spin bordism of $X := (BS_4)^{\text{Det}(V_{\lambda})-1}$.

Theorem 5.59. The first few spin bordism groups of X are

$$\begin{split} \widetilde{\Omega}_{0}^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_{1}^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/6\\ \widetilde{\Omega}_{2}^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/8 \oplus \mathbb{Z}/2\\ \widetilde{\Omega}_{3}^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_{4}^{\mathrm{Spin}}(X) &\cong 0, \end{split}$$

and $\widetilde{\Omega}_5^{\text{Spin}}(X)$ is torsion.

Proof. Odd-primary information is computed in the range we need by (5.40a). For 2-primary information, we use the Adams spectral sequence as usual. Recall the $\mathcal{A}(1)$ -module structure on $H^*(BS_4; \mathbb{Z}/2) \cong \mathbb{Z}/2[a, b, c]/(ac)$ from Propositions 5.34 and 5.35. Lemma 3.30 shows that inclusion of a transposition extends to a splitting

(5.60)
$$X \xrightarrow{\simeq} (B\mathbb{Z}/2)^{\sigma-1} \lor M,$$

and the map $\widetilde{H}^*(M; \mathbb{Z}/2) \to \widetilde{H}^*(X; \mathbb{Z}/2)$ is injective, with image a complementary subspace to the span of $\{Ua^n \mid n \geq 0\}$. As usual, we write down $\widetilde{H}^*(M; \mathbb{Z}/2)$ as an $\mathcal{A}(1)$ -module in low degrees, using $w_1(\text{Det}(V_{\lambda}) - 1) = a$ and $w_2(\text{Det}(V_{\lambda}) - 1) = 0$, and give the answer in Figure 18, left.



FIGURE 18. Left: The $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(M; \mathbb{Z}/2)$ in low degrees. This submodule contains all elements of degree at most 4. Right: the Ext of this module, which is the beginning of the Adams spectral sequence computing $\widetilde{ko}_*(M)$. More information in the proof of Theorem 5.59.

Let $\Sigma^2 N_3$ denote the $\mathcal{A}(1)$ -submodule generated by Ub; this module is studied by Baker [Bak18, §5], who calls it the "whiskered Joker." There is an isomorphism of $\mathcal{A}(1)$ -modules

(5.61)
$$\widetilde{H}^*(M;\mathbb{Z}/2) \cong \Sigma^2 N_3 \oplus \Sigma^3 \mathcal{A}(1) \oplus P,$$

where P contains no elements of degree less than 4. Therefore if the 4-line of the E_2 -page is empty, P does not enter into our calculations — and we will see momentarily that the 4-line is in fact empty. We know what $\Sigma^3 \mathcal{A}(1)$ summand contributes to the E_2 -page of the Adams spectral sequence. For N_3 , we leverage what we learned from N_1 in §5.3.1. Specifically, the unique nonzero $\mathcal{A}(1)$ -module map $\mathcal{A}(1) \to N_3$ has kernel isomorphic to $\Sigma^5 N_1$, so a minimal resolution for $\Sigma^5 N_1$ induces a minimal resolution for N_3 which has an additional copy of $\mathcal{A}(1)$ in topological degree 0 and filtration 0, and in which everything else is shifted up one in filtration, giving the red summands in Figure 18, right.

Thus the E_2 -page for this Adams spectral sequence is as in Figure 18, right. In this range, the spectral sequence collapses. Combine this with the pin⁻ bordism summands from [ABP69, KT90b] as usual to obtain the groups in the theorem statement, and Lemma 3.24 finishes us off by telling us $\widetilde{\Omega}_5^{\text{Spin}}(X)$ is torsion. \Box

5.3.3. Class A, spinless case. Let f_0^A denote the equivariant local system of symmetry types for class A with spinless fermions. In this case, the ansatz asks us to consider the spin^c bordism of $X := (BS_4)^{3-V_{\lambda}}$.

Theorem 5.62. The first few spin^c bordism groups of X are

$$\begin{split} \widetilde{\Omega}_0^{\text{Spin}^c}(X) &= \mathbb{Z}/2\\ \widetilde{\Omega}_1^{\text{Spin}^c}(X) &= \mathbb{Z}/3\\ \widetilde{\Omega}_2^{\text{Spin}^c}(X) &= (\mathbb{Z}/2)^{\oplus 2}\\ \widetilde{\Omega}_3^{\text{Spin}^c}(X) &= \mathbb{Z}/3\\ \widetilde{\Omega}_4^{\text{Spin}^c}(X) &= (\mathbb{Z}/2)^{\oplus 4}, \end{split}$$

and $\widetilde{\Omega}_5^{\mathrm{Spin}^c}(X)$ is torsion. Therefore $Ph_0^{S_4}(\mathbb{R}^3, f_0^A) \cong (\mathbb{Z}/2)^{\oplus 4}$.

Proof. We will use the Adams spectral sequence over $\mathcal{E}(1)$ as usual to capture the 2-primary information; for odd-primary information, see (5.40b).

We use the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(X;\mathbb{Z}/2)$ that we determined in (5.42) and drew in Figure 17 to determine the $\mathcal{E}(1)$ -module structure: as $\mathcal{E}(1)$ -modules, $\mathcal{A}(1) \cong \mathcal{E}(1) \oplus \Sigma^2 \mathcal{E}(1)$, and $N_2 \cong \mathcal{E}(1) \oplus \Sigma \mathcal{E}(1) \oplus \Sigma^2 N_1$, so as $\mathcal{E}(1)$ -modules,

(5.63)
$$\widetilde{H}^*(X;\mathbb{Z}/2) \cong \mathcal{E}(1) \oplus \Sigma^2 \mathcal{E}(1) \oplus \Sigma^2 N_1 \oplus \Sigma^4 \mathcal{E}(1) \oplus \Sigma^4 \mathcal{E}(1) \oplus \Sigma^5 \mathcal{E}(1) \oplus P$$

where P is 5-connected. We draw this in Figure 19, left. Recalling $\text{Ext}_{\mathcal{E}(1)}(N_1)$ from (4.49), the E_2 -page of



FIGURE 19. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*((BS_4)^{3-V_{\lambda}}; \mathbb{Z}/2)$ in low degrees. The pictured submodule contains all elements of degree at most 5. Right: the Ext of this module, which is the beginning of the E_2 -page of the Adams spectral sequence computing $\widetilde{ku}_*((BS_4)^{3-V_{\lambda}})$.

the Adams spectral sequence is in Figure 19, right. There can be no differentials in the range drawn for

degree reasons, and Margolis' theorem (Theorem 3.22) implies there are no nontrivial extensions, either, so we are done. \Box

5.3.4. Class A, spin-1/2 case. Theorem 2.24 says that to compute the S_4 -equivariant phase homology groups in class A with spin-1/2 fermions, given by the equivariant local system of symmetry types $f_{1/2}^A$, we should investigate the spin^c bordism of $X := (BS_4)^{\text{Det}V_{\lambda}-1}$: we know V_{λ} is not pin^c because its pullback along $BA_4 \to BS_4$ is not pin^c, as we established in Lemma 5.7.

Theorem 5.64. The first few spin^c bordism groups of X are

$$\begin{split} \widetilde{\Omega}_0^{\mathrm{Spin}^c}(X) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_1^{\mathrm{Spin}^c}(X) &\cong \mathbb{Z}/3\\ \widetilde{\Omega}_2^{\mathrm{Spin}^c}(X) &\cong \mathbb{Z}/4 \oplus \mathbb{Z}/2\\ \widetilde{\Omega}_3^{\mathrm{Spin}^c}(X) &\cong \mathbb{Z}/6\\ \widetilde{\Omega}_4^{\mathrm{Spin}^c}(X) &\cong \mathbb{Z}/8 \oplus (\mathbb{Z}/2)^{\oplus 2}. \end{split}$$

By Lemma 3.24, $\widetilde{\Omega}_5^{\mathrm{Spin}^c}(X)$ is torsion, so $Ph_0^{S_4}(\mathbb{R}^3; f_{1/2}^A) \cong \mathbb{Z}/8 \oplus (\mathbb{Z}/2)^{\oplus 2}$.

Proof. See (5.40b) for the odd-primary torsion in $\widetilde{\Omega}_*^{\text{Spin}^c}(X)$. For 2-torsion, we reuse our work from §5.3.2. First, $X \simeq (B\mathbb{Z}/2)^{\sigma-1} \lor M$, and we gave the low-degree cohomology of M as an $\mathcal{A}(1)$ -module in (5.61), and drew it in Figure 18, left. This determines the $\mathcal{E}(1)$ -module structure on it, so we can calculate spin^c bordism of M using the Adams spectral sequence. For the other summand, we have $MTSpin^c \land (B\mathbb{Z}/2)^{\sigma-1} \simeq MTPin^c$, so we direct-sum in the pin^c bordism groups computed by Bahri-Gilkey [BG87a, BG87b].

There are isomorphisms of $\mathcal{E}(1)$ -modules $\mathcal{A}(1) \cong \mathcal{E}(1) \oplus \Sigma^2 \mathcal{E}(1)$ and $N_3 \cong \mathcal{E}(1) \oplus \Sigma^2 N_1$. Therefore as an $\mathcal{E}(1)$ -module,

(5.65)
$$\widetilde{H}^*(M;\mathbb{Z}/2) \cong \Sigma^2 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^4 N_1 \oplus P,$$

where P is 4-connected. As usual for these cases, we will see that $\operatorname{Ext}(\widetilde{H}^*(M;\mathbb{Z}/2),\mathbb{Z}/2)$ has no nonzero elements with t-s=4 and s>1, so P does not affect our calculations. See Figure 20, left, for a picture of the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*(M;\mathbb{Z}/2)$. We calculated $\operatorname{Ext}(\Sigma^4 N_1)$ in (4.49), so can draw the E_2 -page of



FIGURE 20. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*(M; \mathbb{Z}/2)$ in low degrees; the pictured summands include all elements in degrees 4 and below. Here $\alpha := a^2b + b^2$. Right: the Ext of this module, which is the beginning of the E_2 -page of the Adams spectral sequence computing $\widetilde{Ku}_*(M)$.

the Adams spectral sequence in Figure 20, right. This collapses, so we add in the pin^c bordism summands and conclude.

5.4. Chiral octahedral symmetry. Let $\lambda: S_4 \to O_3$ denote the representation as symmetries of an octahedron and $V_{\lambda} \to BS_4$ denote the associated vector bundle. Recall from Proposition 5.34 the mod 2 cohomology of BS^4 .

Lemma 5.66. $w_1(V_{\lambda}) = 0$ and $w_2(V_{\lambda}) = b$.

65

Proof. Since $\operatorname{Im}(\lambda) \subset \operatorname{SO}_3$, $w_1(V_{\lambda}) = 0$. We know $w_2(V_{\lambda})$ restricts to $u \in H^2(BA_4; \mathbb{Z}/2)$ by considering tetrahedral symmetry inside octahedral symmetry and using Lemma 5.3, so $w_2(V_{\lambda})$ could be $a^2 + b$ or b. The fact that λ splits as $\sigma \oplus \mathbb{R}^2$ when restricted to a $\mathbb{Z}/2$ subgroup given by a transposition tells us $w_2(V_{\lambda})$ is b, not $a^2 + b$.

By Lemma 5.7, the pullback of V_{λ} to BA_4 is not pin^c, so V_{λ} is not pin^c, and hence V_{λ} is also not pin⁻. Lemma 5.67.

(5.68)
$$\widetilde{\Omega}_{k}^{\text{SO}}(BS_{4}) \otimes \mathbb{Z}[1/2] \cong \begin{cases} \mathbb{Z}/3, & k = 3\\ 0, & k = 0, 1, 2, 4, 5, 6. \end{cases}$$

Proof. Let ℓ be an odd prime and consider the Atiyah-Hirzebruch spectral sequence

(5.69)
$$E_{p,q}^2 = H_p(BS_4; (MTSO_\ell^{\wedge})_q) \Longrightarrow (MTSO_\ell^{\wedge})_{p+q}(BS_4) = \Omega_{p+q}^{SO}(BS_4)_\ell^{\wedge}.$$

If $\ell \neq 3$, then $\ell \nmid |S_4|$, so the \mathbb{Z}_{ℓ} -cohomology of BS_4 vanishes in positive degrees and (5.69) is trivial, contributing no ℓ -torsion to $\widetilde{\Omega}_*^{SO}(BS_4) \otimes \mathbb{Z}[1/2]$. For $\ell = 3$, use Thomas' calculation of $H^*(BS_4; \mathbb{Z})$ [Tho74] and the universal coefficient theorem to show that $H_*(BS_4; \mathbb{Z}_3)$ consists of \mathbb{Z}_3 in degree 0, $\mathbb{Z}/3$ in degree 2, and nothing else nonzero in degrees 5 and below. Therefore (5.69) collapses, giving us the desired result. \Box

5.4.1. Class D, spinless case. Let f_0^D denote the equivariant local system of symmetry types for the spinless class D case. Theorem 2.11 identifies

(5.70)
$$Ph_k^{S_4}(\mathbb{R}^3; f_0^D) \cong [MTSpin \wedge (BS_4)^{3-V_{\lambda}}, \Sigma^{k+4}I_{\mathbb{Z}}],$$

so we study the spin bordism of $X \coloneqq (BS_4)^{3-V_{\lambda}}$.

Theorem 5.71. There is an $r \ge 2$ such that the first few spin bordism groups of X are

$$\begin{aligned} \Omega_0^{\mathrm{Spin}}(X) &\cong \mathbb{Z} \\ \widetilde{\Omega}_1^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/2 \\ \widetilde{\Omega}_2^{\mathrm{Spin}}(X) &\cong 0 \\ \widetilde{\Omega}_3^{\mathrm{Spin}}(X) &\cong \mathbb{Z}/6 \oplus \mathbb{Z}/2^l \\ \widetilde{\Omega}_4^{\mathrm{Spin}}(X) &\cong \mathbb{Z}, \end{aligned}$$

and $\widetilde{\Omega}_5^{\mathrm{Spin}}(X)$ is torsion. Hence $Ph_0^{S_4}(\mathbb{R}^3; f_0^D) = 0$.

The Atiyah-Hirzebruch spectral sequence allows one to show k = 1, so $\widetilde{\Omega}_3^{\text{Spin}}(X) \cong \mathbb{Z}/2 \oplus \mathbb{Z}/2$. As usual, we will not need this, so do not prove it.

Proof. For odd-primary torsion, use the fact that $MTSpin \to MTSO$ is an isomorphism, so it suffices to understand $\widetilde{\Omega}_*^{SO}(X)$, and that $V_{\lambda} \to BS_4$ is orientable, so there is a Thom isomorphism $\Omega_k^{SO}(BS_4) \to \widetilde{\Omega}_k^{SO}(X)$, and we can read off the odd-primary torsion from Lemma 5.67.

On to the prime 2. From Propositions 5.34 and 5.35 we know the mod 2 cohomology of BS_4 and the action of the Steenrod algebra, and using Lemma 5.66 we can draw $\widetilde{H}^*(X;\mathbb{Z}/2)$ as an $\mathcal{A}(1)$ -module in low degrees, which we do in Figure 21, left.

Let N_4 denote the $\mathcal{A}(1)$ -submodule of $\widetilde{H}^*(X;\mathbb{Z}/2)$ generated by U and Ua. Then,

(5.72)
$$H^*(X;\mathbb{Z}/2) \cong \mathbb{N}_4 \oplus \Sigma^3 \mathcal{A}(1) \oplus \Sigma^4 \mathbb{N}_4 \oplus \Sigma^5 \mathcal{A}(1) \oplus \mathbb{P}_4$$

where P is 6-connected. We have not seen N_4 before, and need to calculate its Ext. Fortunately, there is a short exact sequence of $\mathcal{A}(1)$ -modules

$$(5.73) 0 \longrightarrow \Sigma J \longrightarrow N_4 \longrightarrow \hat{O} \longrightarrow 0,$$

which induces a long exact sequence in Ext. In Figure 22, we display a picture both of this extension and of the Adams chart for computing the boundary map in the long exact sequence.

We draw the E_2 -page in Figure 21, right. Because differentials must be h_0 -equivariant, they all vanish in the range pictured except possibly for those from the 4-line to the 3-line, one of which is indicated in the chart. By Lemma 3.24, $\tilde{ko}_4(X) \otimes \mathbb{Q} \cong \tilde{ko}_0(X) \otimes \mathbb{Q}$, and from Figure 21, right, the latter group is isomorphic ARUN DEBRAY



FIGURE 21. Left: the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*((BS_4)^{3-V_\lambda}; \mathbb{Z}/2)$ in low degrees. The pictured submodule contains all elements of degrees 6 and below. Right: the E_2 -page of the corresponding Adams spectral sequence computing $\widetilde{ko}_*((BS_4)^{3-V_\lambda})_2^{\wedge}$. We will see there is a differential from the 4-line to the 3-line; it is in fact the d_2 depicted, though we do not prove that.



FIGURE 22. Left: the extension (5.73) of $\mathcal{A}(1)$ -modules. Right: the long exact sequence in Ext induced from that extension.

to \mathbb{Q} . Thus $\widetilde{\Omega}_4^{\text{Spin}}(X)$ has exactly one free summand, so one of the two towers in the 4-line lives to the E_{∞} -page, and the other admits a nonzero d_r differential to the tower in degree 3. Thus, on the 3-line of the E_{r+1} -page, there is a single green $\mathbb{Z}/2$ summand in degree s = 0, together with a red tower with finitely many $\mathbb{Z}/2$ summands, giving $\mathbb{Z}/2 \oplus \mathbb{Z}/2^k$ in degree 3 as promised.³³

5.4.2. Class D, spin-1/2 case. Let $f_{1/2}^D$ be the S_4 -equivariant local system of symmetry types for the case of spin-1/2 fermions in class D. Theorem 2.11 computes the equivariant phase homology of this local system in terms of $\Omega_*^{\text{Spin}}(BS_4)$.

³³We have not determined which elements of the 4-line the differential is nonzero on. One way to determine this is to use that the generator of $H^{3,7}(\mathcal{A}(1)) \cong \mathbb{Z}/2$ carries the summands in the 0-line onto a subset of the red tower in the 4-line. Differentials are equivariant for this action, and differentials emerging from the 0-line vanish, so all differentials must vanish on the red tower too.

$$\Omega_0^{\mathrm{Spin}}(BS_4) \cong \mathbb{Z}$$

$$\Omega_1^{\mathrm{Spin}}(BS_4) \cong (\mathbb{Z}/2)^{\oplus 2}$$

$$\Omega_2^{\mathrm{Spin}}(BS_4) \cong (\mathbb{Z}/2)^{\oplus 3}$$

$$\Omega_3^{\mathrm{Spin}}(BS_4) \cong \mathbb{Z}/24 \oplus \mathbb{Z}/2^{r+1},$$

$$\Omega_4^{\mathrm{Spin}}(BS_4) \cong \mathbb{Z} \oplus \mathbb{Z}/2$$

$$\Omega_5^{\mathrm{Spin}}(BS_4) \cong 0$$

$$\Omega_6^{\mathrm{Spin}}(BS_4) \cong \mathbb{Z}/2.$$

Therefore $Ph_0^{S_4}(\mathbb{R}^3; f_{1/2}^D) \cong \mathbb{Z}/2.$

One can use the Atiyah-Hirzebruch spectral sequence to show r = 2 in Theorem 5.74; we do not need this so do not present the proof.

Proof. First, we use the Adams spectral sequence to determine the free and 2-primary parts. Since $ko_*(BS_4)$ splits as $ko_*(\text{pt}) \oplus \tilde{ko}_*(BS_4)$, we focus on $\tilde{ko}_*(BS_4)$ and add the Bott-song summands in at the end. There is a section s of the parity map $S_4 \to \mathbb{Z}/2$, which stably splits BS_4 . That is, there is a spectrum M, a map $t: M \to \Sigma^{\infty} BS_4$, and a weak equivalence

$$(5.75) (s,t): \Sigma^{\infty} B\mathbb{Z}/2 \vee M \xrightarrow{\simeq} \Sigma^{\infty} BS_4.$$

This also splits the \mathcal{A} -module structure of $\widetilde{H}^*(BS_4; \mathbb{Z}/2)$ as $\widetilde{H}^*(M; \mathbb{Z}/2) \oplus \widetilde{H}^*(B\mathbb{Z}/2; \mathbb{Z}/2)$, where $\widetilde{H}^*(B\mathbb{Z}/2; \mathbb{Z}/2)$ is embedded via the parity map. Therefore $\widetilde{H}^*(M; \mathbb{Z}/2)$ is isomorphic to a complimentary subspace of $\mathbb{Z}/2 \cdot \{a^k \mid k \geq 0\} \subset \widetilde{H}^*(BS_4; \mathbb{Z}/2)$. As this isomorphism is realized by a map of spectra, it is an isomorphism of \mathcal{A} -modules, hence $\mathcal{A}(1)$ -modules. We will run the Adams spectral sequence for $\widetilde{ko}_*(M)$, and add the $\widetilde{ko}_*(B\mathbb{Z}/2)$ summands in at the end.

The mod 2 cohomology of BS_4 is given in Proposition 5.34, and the action of the Steenrod squares in Proposition 5.35. We can therefore draw $\widetilde{H}^*(M;\mathbb{Z}/2)$ as an $\mathcal{A}(1)$ -module in low degrees, which we do in Figure 23, left. We have



FIGURE 23. Left: the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(M; \mathbb{Z}/2)$ in low degrees. The submodule pictured here contains all elements of degree at most 6. Right: the corresponding Ext, which is the E_2 -page for the Adams spectral sequence converging to the 2-primary part of $\widetilde{ko}_*(M)$.

(5.76)
$$\widetilde{H}^*(M;\mathbb{Z}/2) \cong \Sigma^2 J \oplus \Sigma^3 \partial \oplus \Sigma^4 \mathcal{A}(1) \oplus \Sigma^6 \mathcal{A}(1) \oplus P,$$

ARUN DEBRAY

where P is 6-connected. Names of $\mathcal{A}(1)$ -modules are as in previous sections; for all these modules except for P, we have already seen $\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(-,\mathbb{Z}/2)$, and P is irrelevant for degree reasons. We display the E_2 -page in Figure 23, right.

In the range pictured, h_0 -equivariance of differentials implies the only possible nontrivial differentials are from the infinite tower in degree 4 to the infinite tower in degree 3; a d_2 is pictured as an example. In fact, those towers must support a nonzero d_r for some r; by h_0 -equivariance, d_r is either zero for every element of the tower in degree 4, or nonzero for every element. Hence, if all d_r were zero for all r, then $\tilde{ko}_3(BS_4)$ would contain a free summand, contradicting Lemma 3.24. Therefore there is some $r \geq 2$ for which all d_r differentials from the tower in degree 4 to the tower in degree 3 are nontrivial (not necessarily the d_2 s pictured). On the E_{∞} -page, the tower in degree 4 vanishes, and only r + 1 summands of the degree-3 tower remain. Thus we have computed the 2-primary part of $ko_*(BS_4)$ in degrees 6 and lower:

- From $ko_*(pt)$, we have a Z summand in degrees 0 and 4 and a $\mathbb{Z}/2$ summand in degrees 1 and 2.
- From ko_{*}(BZ/2), we have Z/2 summands in degrees 1 and 2 and a Z/8 summand in degree 3 [MM76].
 From Figure 23, right, we have Z/2 in degree 2, Z/2^{r+1} in degree 3, and a Z/2 each in degrees 4 and
- $\begin{array}{c} \bullet \text{ from figure 20, fight, we have <math>22/2 \text{ in degree 2, } 22/2 \end{array} \quad \text{in degree 3, and } a 22/2 \text{ each in degrees 4 and 6.} \end{array}$

To determine the odd-primary torsion, use first that the forgetful map $\Omega_*^{\text{Spin}}(-) \to \Omega_k^{\text{SO}}(-)$ is an isomorphism on odd-primary torsion, so we just have to determine the odd-primary torsion in $\Omega_k^{\text{SO}}(BS_4)$ for $k \leq 6$, which we did in Lemma 5.67.

5.4.3. Class A. As in the case of chiral tetrahedral symmetry, V_{λ} does not admit a pin^c structure, since we saw in Lemma 5.7 that its pullback along $BA_4 \rightarrow BS_4$ also does not admit a pin^c structure. Let f_0^A , resp. $f_{1/2}^A$, denote the equivariant local systems of spectra associated to the class A spinless, resp. spin-1/2 cases. Theorem 2.24 expresses $Ph_0^{S_4}(\mathbb{R}^3; f_0^A)$ and $Ph_0^{S_4}(\mathbb{R}^3; f_{1/2}^A)$ in terms of the spin^c bordism of $(BS_4)^{3-V_{\lambda}}$ for spinless fermions and BS_4 for spin-1/2 fermions.

Theorem 5.77. There are integers $r, r' \ge 2$ such that the low-degree spin^c bordism groups of $(BS_4)^{3-V_{\lambda}}$ and BS_4 are

$$\begin{split} \widetilde{\Omega}_{0}^{\mathrm{Spin}^{c}}((BS_{4})^{3-V_{\lambda}}) &\cong \mathbb{Z} & \Omega_{0}^{\mathrm{Spin}^{c}}(BS_{4}) &\cong \mathbb{Z} \\ \widetilde{\Omega}_{1}^{\mathrm{Spin}^{c}}((BS_{4})^{3-V_{\lambda}}) &\cong \mathbb{Z}/2 & \Omega_{1}^{\mathrm{Spin}^{c}}(BS_{4}) &\cong \mathbb{Z}/2 \\ \widetilde{\Omega}_{2}^{\mathrm{Spin}^{c}}((BS_{4})^{3-V_{\lambda}}) &\cong \mathbb{Z} & \Omega_{2}^{\mathrm{Spin}^{c}}(BS_{4}) &\cong \mathbb{Z}/2 \\ \widetilde{\Omega}_{3}^{\mathrm{Spin}^{c}}((BS_{4})^{3-V_{\lambda}}) &\cong \mathbb{Z}/6 \oplus \mathbb{Z}/2^{r} & \Omega_{3}^{\mathrm{Spin}^{c}}(BS_{4}) &\cong \mathbb{Z}/12 \oplus \mathbb{Z}/2^{r'} \\ \widetilde{\Omega}_{4}^{\mathrm{Spin}^{c}}((BS_{4})^{3-V_{\lambda}}) &\cong \mathbb{Z}^{2} & \Omega_{4}^{\mathrm{Spin}^{c}}(BS_{4}) &\cong \mathbb{Z}/2 \\ \widetilde{\Omega}_{5}^{\mathrm{Spin}^{c}}((BS_{4})^{3-V_{\lambda}}) &\cong \mathbb{Z}/2^{r-1} \oplus \mathbb{Z}/6 \oplus (\mathbb{Z}/2)^{\oplus 3} & \Omega_{5}^{\mathrm{Spin}^{c}}(BS_{4}) &\cong \mathbb{Z}/2 \oplus \mathbb{Z}/2 \oplus \mathbb{Z}/2^{r'+1} \\ \widetilde{\Omega}_{6}^{\mathrm{Spin}^{c}}((BS_{4})^{3-V_{\lambda}}) &\cong \mathbb{Z}^{2} & \Omega_{6}^{\mathrm{Spin}^{c}}(BS_{4}) &\cong \mathbb{Z}/2 \oplus \mathbb{Z}/2 \oplus \mathbb{Z}/2^{r'+1} \end{split}$$

One can use the Atiyah-Hirzebruch spectral sequence to show r = r' = 2. We do not need this, so do not go into the details.

Proof. As usual, the calculation separates into odd-primary and 2-primary parts.

Lemma 5.78. The only odd-primary torsion in the spin^c bordism of $(BS_4)^{3-V_{\lambda}}$ and BS_4 in degrees 6 and below consists of two $\mathbb{Z}/3$ summands in degrees 3 and 5.

Proof. Since $|S^4| = 2^3 \cdot 3$, we only have to check 3-torsion: if $\ell \geq 5$ is prime, the maps $BS_4 \to \text{pt}$ and $(BS_4)^{3-V} \to \text{pt}$ are stable ℓ -primary equivalences by the Whitehead theorem [Ser53, Chapitre III, Théorème 3]. The forgetful map $MTSpin^c \to MSO \land (BU_1)_+$ is an odd-primary equivalence, and since $3 - V_{\lambda}$ is orientable, there is a Thom isomorphism

$$(5.79) MSO \wedge (BU_1)_+ \wedge (BS_4)^{3-V} \xrightarrow{\simeq} MSO \wedge (BU_1)_+ \wedge (BS_4)_+,$$

so in both the spinless and spin-1/2 cases, we can glean the 3-torsion from $\Omega_*^{\text{Spin}^c}(BU_1 \times BS_4)$. As the homology of BU_1 is torsion-free, the Künneth map $H_*(BU_1) \otimes H_*(BS_4) \to H_*(BU_1 \times BS_4)$ is an isomorphism of graded abelian groups. Using this together with Thomas' [Tho74] calculation of $H_*(BS_4)$, we conclude that the only odd-primary torsion in $H_*(BU_1 \times BS_4)$ in degrees below 7 is $\mathbb{Z}/3 \subset H_3(BU_1 \times BS_4)$ and $\mathbb{Z}/3 \subset H_5(BU_1 \times BS_4)$.

Now feed this to the Atiyah-Hirzebruch spectral sequence with signature

(5.80)
$$E_{p,q}^2 = H_p(BU_1 \times BS_4, \Omega_q^{SO}(\text{pt})) \Longrightarrow \Omega_{p+q}^{SO}(BU_1 \times BS_4)$$

The coefficients are sums of \mathbb{Z} and $\mathbb{Z}/2$; since we only care about 3-torsion, we can ignore the $\mathbb{Z}/2$ summands, whose differentials cannot map nontrivially to or from any 3-torsion element. The only 3-torsion on the E^2 -page in total degree less than 7 is a single $\mathbb{Z}/3$ summand in each of $E_{3,0}^2$ and $E_{5,0}^2$, coming from our calculation above of 3-torsion in homology. These 3-torsion summands cannot participate in any nonzero differentials: they do not map to each other, and cannot receive any differentials from free summands, or from the 7-line (which we have not calculated). Thus they persist to the E^{∞} -page. It is a priori possible more 3-torsion is created from free summands on the E^2 -page, which could happen if a differential maps from a free summand to another free summand. All free summands are in even total degree, though, so this does not happen, and the only 3-torsion in $\Omega_k^{SO}(BU_1 \times BS_4)$, for k < 7, is two $\mathbb{Z}/3$ summands in degrees 3 and 5. \Box

Next, we compute the 2-torsion using the Adams spectral sequence over $\mathcal{E}(1)$.

For the spinless case, recall from (5.72) (drawn in Figure 21) the calculation of $\widetilde{H}^*((BS_4)^{3-V_{\lambda}}; \mathbb{Z}/2)$ as an $\mathcal{A}(1)$ -module. There are isomorphisms of $\mathcal{E}(1)$ -modules $N_4 \cong \widetilde{O} \oplus \Sigma \mathcal{E}(1) \oplus \Sigma^3 \mathbb{Z}/2$ and $\mathcal{A}(1) \cong \mathcal{E}(1) \oplus \Sigma^2 \mathcal{E}(1)$, so as $\mathcal{E}(1)$ -modules,

 $(5.81) \qquad \widetilde{H}^*((BS_4)^{3-V_{\lambda}}; \mathbb{Z}/2) \cong \mathcal{O} \oplus \Sigma \mathcal{E}(1) \oplus \Sigma^3 \mathbb{Z}/2 \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^4 \mathcal{O} \oplus \Sigma^5 \mathcal{E}(1) \oplus \Sigma^5 \mathcal{E}(1) \oplus \Sigma^5 \mathcal{E}(1) \oplus P,$

where P is 6-connected. We draw this in Figure 24, left.



FIGURE 24. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*((BS_4)^{3-V_{\lambda}}; \mathbb{Z}/2)$ in low degrees. The pictured submodule contains all elements of degrees at most 6. Here $\alpha := a^5 + a^3b$. Right: the corresponding Ext, which is the E_2 -page of the Adams spectral sequence for $\widetilde{ku}_*((BS_4)^{3-V})$. Some nonzero v_1 -actions are hidden for clarity.

To draw the E_2 -page of the Adams spectral sequence, use the computations of $\text{Ext}(\hat{O})$ from (4.55) and $\text{Ext}_{\mathcal{E}(1)}(\mathbb{Z}/2)$ from (3.7) to obtain Figure 24, right. For clarity, we do not draw most v_1 -actions. There may be differentials in this range, though we do not determine whether they are the d_2 s pictured.

From Figure 24, right, $\widetilde{ku}_0((BS_4)^{3-V_\lambda}) \cong \mathbb{Z}$, so Lemma 3.24 implies there is a single free summand in each even degree and the odd-degree ku-groups are torsion. Therefore, one of the towers on the 4-line must admit a nontrivial d_r differential to the tower on the 3-line, and in fact, v_1 -equivariance of the differentials implies that tower on the 4-line must be the blue one coming from $\Sigma^4 \mathcal{O}$. The remaining tower must survive, so on the E_∞ -page, the 3-line has its $\mathbb{Z}/2$ summand and a $\mathbb{Z}/2^r$ summand coming from the red tower, and the 4-line has a single \mathbb{Z} summand left. The results on \widetilde{ku}_5 and \widetilde{ku}_6 follow from v_1 - and h_0 -equivariance of d_r .

On to the spin-1/2 case. We factor $ku_*(BS_4) \cong ku_*(\text{pt}) \oplus ku_*(BS_4)$. In the proof of Theorem 5.74, we split $\Sigma^{\infty}BS_4 \simeq \Sigma^{\infty}B\mathbb{Z}/2 \lor M$ and determined the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(M;\mathbb{Z}/2)$. Combining this

with Nguyen's computation [Ngu09, Theorem 2.3.1] of the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*(BS_4; \mathbb{Z}/2)$, we have that as $\mathcal{E}(1)$ -modules,

(5.82)
$$\widetilde{H}^*(M;\mathbb{Z}/2) \cong \Sigma^2 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{O} \oplus \Sigma^4 \mathbb{Z}/2 \oplus \Sigma^4 \mathcal{E}(1) \oplus \Sigma^6 \mathcal{E}(1) \oplus \Sigma^6 \mathcal{E}(1) \oplus P,$$

where P is 6-connected. We draw this in Figure 25, left.



FIGURE 25. Left: the $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*(M; \mathbb{Z}/2)$ in low degrees. The pictured submodule contains all elements of degrees at most 6. Right: the corresponding Ext, which is the E_2 -page of the Adams spectral sequence computing $\widetilde{ku}_*(M)$. Some v_1 -actions are hidden to declutter the diagram.

For each of these modules N (except P, which as usual is too high-degree to be relevant), we already calculated $\operatorname{Ext}_{\mathcal{E}(1)}^{s,t}(N,\mathbb{Z}/2)$: for \mathcal{O} , see (4.55), and for $\mathbb{Z}/2$, see (3.7). Therefore the E_2 -page for the Adams spectral sequence is as drawn in Figure 25, right. Most of the v_1 -actions are hidden to make the diagram clearer. We indicate locations of some possible differentials, but they are not necessarily d_2 s.

Lemma 3.24 implies $ku_*(BS_4)$ is torsion, so all towers present on the E_2 -page must emit or receive differentials. Thus there is some $r' \ge 2$ such that the green tower on the 3-line is killed by a $d_{r'}$ emerging from the orange tower on the 4-line; therefore on the E_{∞} -page, the 4-line contains only the $\mathbb{Z}/2$ summand in $E_{\infty}^{0,4}$, and the 3-line contains $r' \mathbb{Z}/2$ summands, the remains of the tower. For ku_5 and ku_6 , v_1 -equivariance of $d_{r'}$ determines the E_{∞} -page in the same way.

It remains to add in the summands corresponding to $ku_*(\text{pt})$ and $ku_*(B\mathbb{Z}/2)$; the former contributes a \mathbb{Z} summand in each even dimension, and the latter contributes $\mathbb{Z}/2$ in degree 1, $\mathbb{Z}/4$ in degree 3, and $\mathbb{Z}/8$ in degree 5, by work of Hashimoto [Has83, Theorem 3.1].

5.5. Full octahedral symmetry. The full group of symmetries of the octahedron, including orientationreversing ones, is isomorphic to $G \coloneqq A_4 \times \mathbb{Z}/2$. Let $\lambda \colon G \to O_3$ denote the corresponding three-dimensional real representation of G, and $V_{\lambda} \to BG$ denote the associated vector bundle. We saw in §5.4 the pullback of V_{λ} along $BS_4 \to BG$ is not pin^c, so V_{λ} is also not pin^c, and therefore is also not pin⁻.

The Künneth formula and Proposition 5.34 together imply

(5.83) $H^*(BG; \mathbb{Z}/2) \cong \mathbb{Z}/2[x, a, b, c]/(ac),$

where |x| = |a| = 1, |b| = 2, and |c| = 3.

Lemma 5.84. $w_1(V_{\lambda}) = x$ and $w_2(V_{\lambda}) = b + x^2$.

Proof. For w_1 , we know $w_1(V_{\lambda}) \neq 0$ because V_{λ} is unorientable, but because $V_{\lambda}|_{BS_4}$ is orientable, $w_1(V_{\lambda})$ cannot be a or x + a, leaving $w_1(V_{\lambda}) = x$.

For w_2 , we know the pullback of V_{λ} to BS_4 has $w_2(V|_{S_4}) = b$. If $i: B\mathbb{Z}/2 \to BG$ is induced by the inclusion of a reflection in G, then $i^*\lambda$ decomposes as a direct sum of three copies of the sign representation, so $i^*V_{\lambda} \cong 3\sigma$. Therefore $i^*w_2(V_{\lambda}) = x^2$, uniquely constraining $w_2(V_{\lambda}) = b + x^2$.

71

5.5.1. Class D, spinless case. The FCEP says we should study the spin bordism of $(BG)^{3-V_{\lambda}}$. We will argue as we did in the case of pyritohedral symmetry in §5.2, replacing $3 - V_{\lambda}$ with a virtual vector bundle whose Adams E_2 -page is isomorphic to that of $(BG)^{3-V_{\lambda}}$, but which is easier to calculate. This isomorphism did not come from a map of spectra, so cannot tell us anything about differentials or hidden extensions, but just as for pyritohedral symmetry, we will see that for entirely formal reasons, all differentials vanish and all hidden extensions split in the range we need. Using the twisted Künneth formula, $\tilde{H}^*((BG)^{3-V_{\lambda}})$ contains no odd-primary torsion, so neither does $\tilde{\Omega}_*^{\text{Spin}}((BG)^{3-V_{\lambda}})$, so using the 2-primary Adams spectral sequence suffices.

For the rest of this section, all homology and cohomology is understood to be with $\mathbb{Z}/2$ coefficients.

Lemma 5.85. Let $E \to BG$ denote the virtual vector bundle induced from the virtual representation

$$(5.86) 2 - (V_{\lambda}|_{S_4} \boxplus (-\sigma)).$$

Then, there is an isomorphism of $\mathcal{A}(1)$ -modules $\widetilde{H}^*((BG)^{3-V_{\lambda}}) \cong \widetilde{H}^*((BG)^E)$, hence an isomorphism between the E_2 -pages of the Adams spectral sequences for $ko \wedge (BG)^{3-V_{\lambda}}$ and $ko \wedge (BG)^E$.

Proof. The E_2 -pages of these Adams spectral sequences are determined by the $\mathcal{A}(1)$ -module structures on cohomology, which are in turn determined by w_1 and w_2 of the virtual bundles $3 - V_\lambda$ and E. Since $w_1(E) = x$ and $w_2(E) = u$, then for $i = 1, 2, w_i(3 - V_\lambda) = w_i(E)$.

Because E is induced from a representation which is an exterior sum, its Thom spectrum splits as

(5.87)
$$(BG)^E \simeq (BS_4)^{3-V_\lambda|_{S_4}} \wedge (B\mathbb{Z}/2)^{\sigma-1}$$

The Künneth theorem then simplifies the E_2 -page:

(5.88)
$$E_2^{s,t} = \operatorname{Ext}_{\mathcal{A}(1)}^{s,t} (\widetilde{H}^*((BS_4)^{3-V_\lambda|_{S_4}}) \otimes_{\mathbb{Z}/2} \widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1}), \mathbb{Z}/2).$$

Campbell [Cam17, Figure 6.1] computes the $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1})$, and we computed $\widetilde{H}^*((BS_4)^{3-V_{\lambda}})$ in (5.72) (drawn in Figure 21).

Proposition 5.89. There is an isomorphism of $\mathcal{A}(1)$ -modules

(5.90)
$$\widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1}) \otimes_{\mathbb{Z}/2} N_4 \cong \Sigma N_5 \oplus (V_2 \otimes_{\mathbb{Z}/2} \mathcal{A}(1)) \oplus P_2,$$

where N_5 is as in Figure 26, V_2 is a graded vector space with a homogeneous basis in degrees $\{0, 2, 3, 4\}$, and P_2 is 4-connected.

Proof. Compute directly, by hand or by computer.

Recall from (5.72) (drawn in Figure 21) the $\mathcal{A}(1)$ -module structure on $(BS_4)^{3-V_{\lambda}}$. Margolis' theorem (Theorem 3.22) splits off a $\Sigma^k H\mathbb{Z}/2$ summand from $ko \wedge (BS_4)^{3-V_{\lambda}}$ for every direct summand of $\Sigma^k \mathcal{A}(1)$ in $\widetilde{H}^*((BS_4)^{3-V_{\lambda}})$; below degree 8, this occurs for k = 3, 5. Therefore, by the same line of reasoning as in §5.2, there is a spectrum Y' such that

(5.91)
$$\widetilde{ko}_n((BG)^{3-V_{\lambda}}) \cong \pi_n(Y') \oplus \widetilde{H}_{n-3}((B\mathbb{Z}/2)^{\sigma-1}) \oplus \widetilde{H}_{n-5}((B\mathbb{Z}/2)^{\sigma-1}),$$

and as \mathcal{A} -modules,

(5.92)
$$\widetilde{H}^*(Y') \cong \mathcal{A} \otimes_{\mathcal{A}(1)} (N_4 \oplus \Sigma^4 N_4 \oplus P_3) \otimes_{\mathbb{Z}/2} \widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1})$$

where P_3 is a 4-connected $\mathcal{A}(1)$ -module. Therefore the change-of-rings formula (3.4) applies to the E_2 -page of the Adams spectral sequence for $\pi_*(Y')$, showing

(5.93)
$$E_2^{s,t}(Y') \cong \operatorname{Ext}_{\mathcal{A}(1)}^{s,t}((N_4 \oplus \Sigma^4 N_4 \oplus P_3) \otimes_{\mathbb{Z}/2} \widetilde{H}^*((B\mathbb{Z}/2)^{\sigma-1}), \mathbb{Z}/2).$$

To calculate the spin bordism groups of $(BG)^{3-V_{\lambda}}$, we will work with this spectral sequence, adding the summands corresponding to $\Sigma^{3}H\mathbb{Z}/2$ and $\Sigma^{5}H\mathbb{Z}/2$ at the end.

Theorem 5.94. The first few spin bordism groups of $(BG)^{3-V_{\lambda}}$ are

$$\widehat{\Omega}_{0}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong \mathbb{Z}/2$$

$$\widetilde{\Omega}_{1}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong \mathbb{Z}/2$$

$$\widetilde{\Omega}_{2}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong (\mathbb{Z}/2)^{\oplus 2}$$

$$\widetilde{\Omega}_{3}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong (\mathbb{Z}/2)^{\oplus 2}$$

$$\widetilde{\Omega}_{4}^{\mathrm{Spin}}((BG)^{3-V_{\lambda}}) \cong (\mathbb{Z}/2)^{\oplus 4}$$

and $\widetilde{\Omega}_5^{\text{Spin}}((BG)^{3-V_{\lambda}})$ is torsion, so the 0th $(S_4 \times \mathbb{Z}/2)$ -equivariant phase homology group for this case is isomorphic to $(\mathbb{Z}/2)^{\oplus 4}$.

Proof. Proposition 5.89 and (5.93) together imply the E_2 -page for Y' is

(5.95)
$$E_2^{s,t}(Y') \cong \operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(\Sigma N_5 \oplus V_2 \otimes_{\mathbb{Z}/2} \mathcal{A}(1) \oplus \Sigma^5 N_5 \oplus \Sigma^4 V_2 \otimes_{\mathbb{Z}/2} \mathcal{A}(1) \oplus P, \mathbb{Z}/2),$$

where P is 4-connected. We will see that the E_2 -page in $t - s \le 4$ is empty for $s \ge 2$, so there can be no differentials involving Ext(P) in the range we care about.

Our first order of business is therefore to determine $\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(N_5,\mathbb{Z}/2)$ for small s,t. There is an extension of $\mathcal{A}(1)$ -modules

$$(5.96) 0 \longrightarrow R_3 \longrightarrow N_5 \longrightarrow \Sigma R_0 \longrightarrow 0$$

which we draw in Figure 26, left, fitting $\operatorname{Ext}_{\mathcal{A}(1)}^{s,t}(N_5,\mathbb{Z}/2)$ into a long exact sequence (Figure 26, right). The $\mathcal{A}(1)$ -module R_3 and its Ext are calculated in the range we need by Freed-Hopkins [FH16a, Figure 5, case s = 3] and Beaudry-Campbell [BC18, Figures 32, 33]. In the range pictured, there are two boundary maps in Figure 26, right, which could be nonzero; the existence of a nonzero map $N_5 \to \Sigma^4 \mathbb{Z}/2$ forces the boundary map $\delta \colon \operatorname{Ext}^{0,4}(R_3) \to \operatorname{Ext}^{1,4}(\Sigma R_0)$ to vanish. We do not need to know whether the other pictured boundary map vanishes.



FIGURE 26. Left: the $\mathcal{A}(1)$ -module N_5 in the extension (5.96). Right: the corresponding long exact sequence in Ext.

Hence the E_2 -page for computing $\pi_*(Y')$ is
The 4-line is concentrated in filtration 0 and 1, and so there can be neither nonzero differentials nor nontrivial extension problems involving elements of degree 4 or less. This accounts for $\pi_*(Y')$; for the each factor of $\widetilde{H}_{*-\ell}((B\mathbb{Z}/2)^{\sigma-1})$, add a single $\mathbb{Z}/2$ summand in degrees ℓ and above.

5.5.2. Class D, spin-1/2 case. Now we ask for the symmetries to mix. Let $f_{1/2}^D$ denote the local system of symmetry types for this case. By Theorem 2.11, we consider the spin bordism of $X := (BS_4 \times B\mathbb{Z}/2)^{\text{Det}(V_\lambda)-1}$, because V_λ is not pin⁻. The isomorphism $\text{Det}V_\lambda \cong 0 \boxplus \sigma$ provides an isomorphism $X \simeq (BS_4)_+ \wedge (B\mathbb{Z}/2)^{\sigma-1}$, so (2.10b) implies the spin bordism of this spectrum computes the pin⁻ bordism of BS_4 , which could be independently interesting.

Theorem 5.98. The first few spin bordism groups of X are

$$\begin{split} &\widetilde{\Omega}_0^{\mathrm{Spin}}(X) \cong \mathbb{Z}/2 \\ &\widetilde{\Omega}_1^{\mathrm{Spin}}(X) \cong (\mathbb{Z}/2)^{\oplus 2} \\ &\widetilde{\Omega}_2^{\mathrm{Spin}}(X) \cong \mathbb{Z}/8 \oplus \mathbb{Z}/4 \oplus \mathbb{Z}/2 \\ &\widetilde{\Omega}_3^{\mathrm{Spin}}(X) \cong (\mathbb{Z}/2)^{\oplus 4} \\ &\widetilde{\Omega}_4^{\mathrm{Spin}}(X) \cong (\mathbb{Z}/2)^{\oplus 2}. \end{split}$$

Since $\widetilde{\Omega}_5^{\text{Spin}}(X)$ is torsion by Lemma 3.24, $Ph_0^{S_4 \times \mathbb{Z}/2}(\mathbb{R}^3; f_{1/2}^D) \cong \mathbb{Z}/2.$

Proof. As usual, Lemma 3.30 spits X as a sum of $(B\mathbb{Z}/2)^{\sigma-1}$ and another spectrum M, where $\widetilde{H}^*(M;\mathbb{Z}/2)$ is complementary in $\widetilde{H}^*(X;\mathbb{Z}/2)$ to the space spanned by $\{Uw_1(\lambda)^k\}$. The $(B\mathbb{Z}/2)^{\sigma-1}$ summand gives us pin⁻ bordism, and we focus on M.

We have $w_1(\text{Det}(V_{\lambda}) - 1) = w_1(V_{\lambda}) = x$ and $w_2(\text{Det}V_{\lambda} - 1) = 0$; this and the \mathcal{A} -module structure on $BS_4 \times B\mathbb{Z}/2$ calculated in (5.83) determine the $\mathcal{A}(1)$ -module structure on M. We obtain an isomorphism of $\mathcal{A}(1)$ -modules

(5.99)
$$\widetilde{H}^*(M;\mathbb{Z}/2) \cong \Sigma R_5 \oplus \Sigma^2 R_3 \oplus \Sigma^3 \mathcal{A}(1) \oplus \Sigma^3 \mathcal{A}(1) \oplus \Sigma^3 \mathcal{A}(1) \oplus \Sigma^3 \mathcal{A}(1) \oplus \Sigma^4 \mathcal{A}(1) \oplus \Sigma^4 \mathcal{A}(1) \oplus P,$$

where P is 4-connected. We will see momentarily that for $t - s \leq 4$, $E_2^{s,t}$ is empty for $s \geq 2$; this and the 4-connectedness of P imply its contribution to the E_2 -page cannot affect the spectral sequence in degrees $t - s \leq 4$, which is all we need. We draw these summands, except for P, in Figure 27.



FIGURE 27. The $\mathcal{A}(1)$ -module structure on $\widetilde{H}^*(M; \mathbb{Z}/2)$ in low degrees. The pictured summand contains all classes in degrees 4 and below. Here $\alpha := b^2 x^2 + a^2 b^2 + c^2$.

Freed-Hopkins [FH16a, Figure 5, cases $s = \pm 3$] and Beaudry-Campbell [BC18, Figures 32, 33, 37] calculate $\text{Ext}(R_5)$ and $\text{Ext}(R_3)$ in the degrees we need, and we can draw the E_2 -page of the Adams spectral sequence:



This collapses, and it and the pin⁻ bordism groups from the $(B\mathbb{Z}/2)^{\sigma-1}$ summand, which are computed in [ABP69, KT90b], sum together to the groups in the theorem.

5.5.3. Class A, spinless case. Let f_0^A denote the local system of symmetry types for this case. We want to calculate $\widetilde{\Omega}_*^{\text{Spin}^c}((BG)^{3-V_{\lambda}})$. Using the twisted Künneth formula, $\widetilde{H}^*((BG)^{3-V_{\lambda}}); \mathbb{Z}/2)$ is 2-torsion, and therefore $\widetilde{\Omega}_*^{\text{Spin}^c}((BG)^{3-V_{\lambda}})$ is too, so it suffices to use the 2-primary Adams spectral sequence.

Theorem 5.101. The first few spin^c bordism groups of $(BG)^{3-V_{\lambda}}$ are:

$$\begin{split} \widetilde{\Omega}_{0}^{\mathrm{Spin}^{c}}((BG)^{3-V_{\lambda}}) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_{1}^{\mathrm{Spin}^{c}}((BG)^{3-V_{\lambda}}) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_{2}^{\mathrm{Spin}^{c}}((BG)^{3-V_{\lambda}}) &\cong (\mathbb{Z}/2)^{\oplus 3}\\ \widetilde{\Omega}_{3}^{\mathrm{Spin}^{c}}((BG)^{3-V_{\lambda}}) &\cong (\mathbb{Z}/2)^{\oplus 3}\\ \widetilde{\Omega}_{4}^{\mathrm{Spin}^{c}}((BG)^{3-V_{\lambda}}) &\cong \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 4}. \end{split}$$

and $\widetilde{\Omega}_5^{\operatorname{Spin}^c}((BG)^{3-V_{\lambda}})$ is torsion. Hence $Ph_0^{S_4 \times \mathbb{Z}/2}(\mathbb{R}^3; f_0^A) \cong \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 4}$.

Proof. There is an isomorphism of $\mathcal{E}(1)$ -modules

(5.102)
$$N_5 \cong \mathcal{E}(1) \oplus \Sigma R_0 \oplus \Sigma^2 R_0,$$

hence another isomorphism of $\mathcal{E}(1)$ -modules

(5.103)
$$\widetilde{H}^*((BG)^{3-V_{\lambda}}) \cong (V_c \otimes_{\mathbb{Z}/2} \mathcal{A}(1)) \oplus \Sigma^2 R_0 \oplus \Sigma^3 R_0 \oplus P_c,$$

where P_c is 4-connected and V_c is a graded vector space with a homogeneous basis of elements in degrees $\{0, 1, 2, 2, 3, 3, 4, 4, 4, 4\}$. Therefore the E_2 -page of the Adams spectral sequence is



Below degree 5, there are no nonzero differentials, because there is nothing in Adams filtration 2 or higher. And degree considerations rule out hidden extensions, so we are done. \Box

5.5.4. Class A, spin-1/2 case. Because V_{λ} is not pin^c, Theorem 2.24 tells us to compute the spin^c bordism groups of $X := (BS_4 \times B\mathbb{Z}/2)^{\text{Det}(V_{\lambda})-1}$.

Theorem 5.105. The first few spin^c bordism groups of X are

$$\begin{split} \Omega_0^{\mathrm{Spin}^-}(X) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_1^{\mathrm{Spin}^c}(X) &\cong \mathbb{Z}/2\\ \widetilde{\Omega}_2^{\mathrm{Spin}^c}(X) &\cong \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 2}\\ \widetilde{\Omega}_3^{\mathrm{Spin}^c}(X) &\cong (\mathbb{Z}/2)^{\oplus 4}\\ \widetilde{\Omega}_4^{\mathrm{Spin}^c}(X) &\cong \mathbb{Z}/8 \oplus \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 4} \end{split}$$

As $\widetilde{\Omega}_5^{\text{Spin}^c}(X)$ is torsion, the 0th $(S_4 \times \mathbb{Z}/2)$ -equivariant phase homology group for this case is isomorphic to $\mathbb{Z}/8 \oplus \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 4}$.

Proof. By Lemma 3.30, X splits as $(B\mathbb{Z}/2)^{\sigma-1} \vee M$, where $\widetilde{H}^*(M; \mathbb{Z}/2)$ is isomorphic to a complementary subspace to the subspace $\mathbb{Z}/2 \cdot \{Ux^k\}$ inside $\widetilde{H}^*(X; \mathbb{Z}/2)$. As usual, the $(B\mathbb{Z}/2)^{\sigma-1}$ summand contributes pin^c bordism groups to the final answer, so we focus on M. The $\mathcal{A}(1)$ -module structure we computed in (5.99) and drew in Figure 27 tells us the $\mathcal{E}(1)$ -structure; here, we use that $R_5 \cong \mathcal{E}(1) \oplus \Sigma R_0$ and $R_3 \cong \mathcal{E}(1) \oplus \Sigma^2 R_0$ as $\mathcal{E}(1)$ -modules. Therefore, there is an $\mathcal{E}(1)$ -module isomorphism

 $(5.106) \quad \widetilde{H}^*(M;\mathbb{Z}/2) \cong \Sigma \mathcal{E}(1) \oplus \Sigma^2 R_0 \oplus \Sigma^2 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^3 \mathcal{E}(1) \oplus \Sigma^4 R_0 \oplus \Sigma^4 \mathcal{E}(1) \oplus \Sigma^4 \mathcal{E}(1) \oplus P,$

where P is 4-connected. Therefore to infer anything about $\widetilde{\Omega}_4^{\text{Spin}^c}(M)$ from this spectral sequence, we must argue that P does not affect it; this will follow when we see the t - s = 4 line of the E_2 -page is empty in Adams filtration 2 and above, so there can be no nonzero differentials from the 5-line to the 4-line. We draw (5.106) in Figure 28.



FIGURE 28. The $\mathcal{E}(1)$ -module structure on $\widetilde{H}^*(M; \mathbb{Z}/2)$ in low degrees. Here $\alpha := abx + b^2 + cx$. This submodule contains all elements in degrees 4 and below.

Recalling $Ext(R_0)$ from Proposition 4.48, the E_2 -page of the Adams spectral sequence for $\widetilde{ku}_*(M)$ is



In this range, the spectral sequence collapses, so we read off $\widetilde{\Omega}^{\text{Spin}^c}_*(M)$ and combine it with pin^c bordism as computed in [BG87a, BG87b] to conclude.

5.6. Chiral icosahedral symmetry. Let $\lambda: A_5 \to SO_3$ denote the representation given by chiral icosahedral symmetry, and as usual let $V_{\lambda} \to BA_5$ denote the associated vector bundle.

Remark 5.108. Unlike the previous symmetry groups we studied, icosahedral symmetry is incompatible with translations, and there are no space groups whose underlying point group is either the chiral icosahedral group or the full icosahedral group. This means one should not expect to realize any phases equivariant for these symmetry groups as a lattice Hamiltonian system on a periodic lattice on \mathbb{R}^3 . This does not rule out the possibility of interesting phases with an icosahedral symmetry: there are examples of phases studied via lattice Hamiltonian realizations on lattices in great generality, such as the toric code model in [Fre19, §2.3], the GDS model in [FH16b, Deb20, FHHT20], and the phases on aperiodic lattices studied by Huang-Wu-Liu [HWL20]. In a similar vein, it may be possible for a Hamiltonian on an aperiodic lattice with icosahedral symmetry to model a nontrivial crystalline SPT. See [VLP+19] for an example of how such an implementation might look.

For icosahedral symmetry, the hard work is behind us. Let $\lambda: A_5 \to O_3$ denote the representation as the orientation-preserving symmetries of the icosahedron. The restriction to $A_4 \subset A_5$ corresponds to symmetries that preserve a concentric tetrahedron. Let $V_{\lambda} \to BA_5$ be the associated bundle to λ .

Lemma 5.109. The inclusion $\varphi: A_4 \hookrightarrow A_5$ induces an equivalence on mod 2 cohomology. Hence φ induces 2-primary equivalences $\Sigma^{\infty}(BA_4)_+ \to \Sigma^{\infty}(BA_5)_+$ and $(BA_4)^{3-\varphi^*(V_{\lambda})} \to (BA_5)^{3-V_{\lambda}}$.

Proof. The first part is Lemma 3.27: here $[A_5 : A_4] = 5$, $P = \mathbb{Z}/2 \times \mathbb{Z}/2$, and for both A_4 and A_5 , $N(P)/P \cong \mathbb{Z}/3$.

For the second part, the Thom isomorphism theorem tells us $\varphi' : (BA_4)^{3-\varphi^*(V_\lambda)} \to (BA_5)^{3-V_\lambda}$ induces an isomorphism on mod 2 cohomology. The desired 2-primary equivalences then follow from the mod 2 Whitehead theorem [Ser53, Chapitre III, Théorème 3].

We can therefore reuse the calculations we made at the prime 2 in §5.1 to obtain the 2-primary parts of $\widetilde{\Omega}_k^{\text{Spin}}((BA_5)^{3-V_{\lambda}})$ and $\Omega_k^{\text{Spin}}(BA_5)$; the odd-primary pieces are different, but not hard.

Proposition 5.110. The only odd-primary torsion in $H_k(BA_5)$ for k < 7 is contained in $H_3(BA_5) \cong \mathbb{Z}/30$.

Proof sketch. One can compute this using GAP; we also indicate how to do it by hand. Since $|A_5| = 60 = 2^2 \cdot 3 \cdot 5$, there is no *p*-primary torsion for p > 5, so it suffices to determine $H^k(BA_5; \mathbb{Z}/3)$ and $H^k(BA_5; \mathbb{Z}/5)$ in low degrees. This can be done using the theorem of Adem-Milgram [AM04, Theorem II.6.8] mentioned above, since the Sylow 3- and 5-subgroups of A_5 are abelian.

Corollary 5.111. In $\widetilde{\Omega}_k^{\text{Spin}}((BA_5)^{3-V_{\lambda}})$ and $\Omega_k^{\text{Spin}}(BA_5)$, the only odd-primary torsion for k < 7 is a $\mathbb{Z}/15$ in degree 3.

Proof. As usual, we use the fact that $\Omega^{\text{Spin}}_* \to \Omega^{\text{SO}}_*$ is an isomorphism on odd-primary torsion, together with the Thom isomorphism $\widetilde{\Omega}^{\text{SO}}_*((BA_5)^{3-V_{\lambda}}) \cong \Omega^{\text{SO}}_*(BA_5)$, to reduce to showing the claim for $\Omega^{\text{SO}}_k(BA_5)$. For this, use the Atiyah-Hirzebruch spectral sequence

(5.112)
$$E_{p,q}^2 = H_p(BA_5; \Omega_q^{SO}(\text{pt})) \Longrightarrow \Omega_{p+q}^{SO}(BA_5)$$

On the E^2 -page, the only odd-primary torsion in total degree below 7 is $\mathbb{Z}/15 \subset E_{3,0}^2 = H_3(BA_5)$. In all differentials involving $E_{3,0}^r$, the other group is zero, so this odd-primary torsion lives to the E^{∞} -page.

We also must check that the free summands in total degree below 7 do not receive differentials that produce more odd-primary torsion. There are only two such free summands, in $E_{0,0}^2$ and $E_{0,4}^2$, and they can only receive differentials from 2-torsion abelian groups, so that does not happen.

Now we need to combine this with the 2-primary summands. For $(BA_5)^{3-V_{\lambda}}$, we need $\Omega_*^{\text{Spin}}((BA_4)^{3-\varphi^*V_{\lambda}})$, which we computed in Theorem 5.4. For BA_5 , we need $\Omega_*^{\text{Spin}}(BA_4)$; in the degrees we need, this is isomorphic to $ko_*(BA_4)$, which Bruner-Greenlees compute in [BG10, §7.7.E].

Theorem 5.113. The low-degree spin bordism groups of $(BA_5)^{3-V}$ and BA_5 are

$$\begin{split} \widetilde{\Omega}_{0}^{\mathrm{Spin}}((BA_{5})^{3-V_{\lambda}}) &\cong \mathbb{Z} & \Omega_{0}^{\mathrm{Spin}}(BA_{5}) \cong \mathbb{Z} \\ \widetilde{\Omega}_{1}^{\mathrm{Spin}}((BA_{5})^{3-V_{\lambda}}) &\cong 0 & \Omega_{1}^{\mathrm{Spin}}(BA_{5}) \cong \mathbb{Z}/2 \\ \widetilde{\Omega}_{2}^{\mathrm{Spin}}((BA_{5})^{3-V_{\lambda}}) &\cong 0 & \Omega_{2}^{\mathrm{Spin}}(BA_{5}) \cong \mathbb{Z}/2 \oplus \mathbb{Z}/2 \\ \widetilde{\Omega}_{3}^{\mathrm{Spin}}((BA_{5})^{3-V_{\lambda}}) &\cong \mathbb{Z}/30 & \Omega_{3}^{\mathrm{Spin}}(BA_{5}) \cong \mathbb{Z}/60 \\ \widetilde{\Omega}_{4}^{\mathrm{Spin}}((BA_{5})^{3-V_{\lambda}}) &\cong \mathbb{Z} & \Omega_{4}^{\mathrm{Spin}}(BA_{5}) \cong \mathbb{Z} \\ \widetilde{\Omega}_{5}^{\mathrm{Spin}}((BA_{5})^{3-V_{\lambda}}) &\cong \mathbb{Z}/2 \oplus \mathbb{Z}/2 & \Omega_{5}^{\mathrm{Spin}}(BA_{5}) \cong 0 \\ \widetilde{\Omega}_{5}^{\mathrm{Spin}}((BA_{5})^{3-V_{\lambda}}) &\cong \mathbb{Z}/2 & \Omega_{5}^{\mathrm{Spin}}(BA_{5}) \cong \mathbb{Z}/2. \end{split}$$

Hence the 0^{th} A₅-equivariant phase homology groups vanish for both spinless and spin-1/2 fermions.

Finally, class A. Since V_{λ} is not pin^c, because its restriction to A_4 is not (Lemma 5.7), we care about $(BA_5)^{\text{Det}(V_{\lambda})-1} \cong (BA_5)_+$ in the spin-1/2 case, because V_{λ} is orientable. Let f_0^A , resp. $f_{1/2}^A$, denote the equivariant local systems of symmetry types for the class A spinless, resp. spin-1/2 cases.

Theorem 5.114. The low-degree spin^c bordism groups of $(BA_5)^{3-V_{\lambda}}$ and BA_5 are

$$\begin{split} \widetilde{\Omega}_{0}^{\mathrm{Spin}^{c}}((BA_{5})^{3-V_{\lambda}}) &\cong \mathbb{Z} & \Omega_{0}^{\mathrm{Spin}^{c}}(BA_{5}) &\cong \mathbb{Z} \\ \widetilde{\Omega}_{1}^{\mathrm{Spin}^{c}}((BA_{5})^{3-V_{\lambda}}) &\cong 0 & \Omega_{1}^{\mathrm{Spin}^{c}}(BA_{5}) &\cong 0 \\ \widetilde{\Omega}_{2}^{\mathrm{Spin}^{c}}((BA_{5})^{3-V_{\lambda}}) &\cong \mathbb{Z} & \Omega_{2}^{\mathrm{Spin}^{c}}(BA_{5}) &\cong \mathbb{Z} \oplus \mathbb{Z}/2 \\ \widetilde{\Omega}_{3}^{\mathrm{Spin}^{c}}((BA_{5})^{3-V_{\lambda}}) &\cong \mathbb{Z}/30 & \Omega_{3}^{\mathrm{Spin}^{c}}(BA_{5}) &\cong \mathbb{Z}/30 \\ \widetilde{\Omega}_{4}^{\mathrm{Spin}^{c}}((BA_{5})^{3-V_{\lambda}}) &\cong \mathbb{Z}^{2} & \Omega_{4}^{\mathrm{Spin}^{c}}(BA_{5}) &\cong \mathbb{Z}^{2}, \end{split}$$

and in both cases, $\Omega_5^{\text{Spin}^c}$ is torsion. Hence both $Ph_0^{A_5}(\mathbb{R}^3; f_0^A)$ and $Ph_0^{A_5}(\mathbb{R}^3; f_{1/2}^A)$ vanish.

Proof. The calculation separates into 2-primary and odd-primary computations; by Lemma 5.109, the 2-primary pieces are exactly as in Theorem 5.8.

The calculation of the odd-primary parts follows the same line of reasoning as the proof of Lemma 5.78: as usual, use the odd-primary equivalence $MTSpin^c \to MTSO \land (BU_1)_+$. We know from Proposition 5.110 that the only odd-primary torsion in $H_k(BA_5)$ for $k \leq 6$ is $\mathbb{Z}/15 \subset H_3$; feeding that to the Künneth formula, the only odd-primary torsion in $H_k(BU_1 \times BA_5)$ is two $\mathbb{Z}/15$ summands in H_3 and H_5 . Then the Atiyah-Hirzebruch argument is identical to the argument in Lemma 5.78.

5.7. Full icosahedral symmetry. If one includes orientation-reversing symmetries of the icosahedron, the symmetry group enlarges to $A_5 \times \mathbb{Z}/2$, with the $\mathbb{Z}/2$ generated by an inversion. This symmetry group is also incompatible with translations, so Remark 5.108 applies. This calculation also quickly reduces to something we already know: restricting the representation to $A_4 \times \mathbb{Z}/2$ yields the pyritohedral symmetry representation we studied in §5.2.

Theorem 5.115. Let ρ be a virtual $A_5 \times \mathbb{Z}/2$ -representation with rank zero, and let $V_{\rho} \to BG$ denote the associated virtual vector bundle. Suppose that $w_1(V_{\rho}) = x$, where x denotes the generator of $H^1(B\mathbb{Z}/2; \mathbb{Z}/2) \subset H^1(B(A_5 \times \mathbb{Z}/2); \mathbb{Z}/2)$. Then inclusion of the pyritohedral symmetry subgroup $\varphi \colon A_4 \times \mathbb{Z}/2 \hookrightarrow A_5 \times \mathbb{Z}/2$ induces a homotopy equivalence $B(A_4 \times \mathbb{Z}/2)^{V_{\rho}} \xrightarrow{\simeq} B(A_5 \times \mathbb{Z}/2)^{V_{\rho}}$.

Proof. By the Whitehead theorem, it suffices to establish that φ induces an isomorphism $\widetilde{H}^*(B(A_5 \times \mathbb{Z}/2)^{V_{\rho}}; k) \to \widetilde{H}^*(B(A_4 \times \mathbb{Z}/2)^{V_{\rho}}; k)$ for $k = \mathbb{Q}$ and $k = \mathbb{Z}/p$ for all primes p.

Lemma 5.109 and the Künneth theorem imply that $\varphi^* \colon H^*(B(A_5 \times \mathbb{Z}/2); \mathbb{Z}/2) \to H^*(B(A_4 \times \mathbb{Z}/2); \mathbb{Z}/2)$ is an isomorphism. Together with the Thom isomorphism theorem, this takes care of the case $k = \mathbb{Z}/2$.

Let G be either of $A_4 \times \mathbb{Z}/2$ or $A_5 \times \mathbb{Z}/2$; the map $B\varphi \colon B(A_4 \times \mathbb{Z}/2) \to B(A_5 \times \mathbb{Z}/2)$ allows us to think of V_{ρ} as over BG for either G, and make sense of the statement $w_1(V_{\rho}) = x$. The Thom isomorphism implies $\widetilde{H}^*((BG)^{V_{\rho}};\mathbb{Z}) \cong H^*(BG;\mathbb{Z}_x)$, and since \mathbb{Z}_x arises as a pullback local system along $BG \to B\mathbb{Z}/2$, the twisted Künneth formula proves $\widetilde{H}^*(BG;\mathbb{Z})$ is 2-torsion. The universal coefficient theorem then implies that when we take coefficients in $k = \mathbb{Q}$ or $k = \mathbb{Z}/p$ for p odd, $\widetilde{H}^*(B(A_4 \times \mathbb{Z}/2)^{V_\rho}; k)$ and $H^*(B(A_5 \times \mathbb{Z}/2)^{V_\rho}; k)$ vanish, so the map between them is vacuously an isomorphism. \square

Let $\lambda: A_5 \times \mathbb{Z}/2 \to O_3$ denote the representation as the group of symmetries of an icosahedron and $V_{\lambda} \to B(A_5 \times \mathbb{Z}/2)$ denote the associated vector bundle. Then $w_1(V_{\lambda}) = x$. Let f_0^D and $f_{1/2}^D$ denote the spinless, resp. spin-1/2 class D equivariant local systems of symmetry types, and f_0^A and $f_{1/2}^{A'}$ denote their analogues in class A.

Corollary 5.116. φ induces homotopy equivalences

 $B(A_4 \times \mathbb{Z}/2))^{3-V_{\lambda}} \xrightarrow{\cong} (B(A_5 \times \mathbb{Z}/2))^{3-V_{\lambda}}$ (5.117a)

(5.117b)
$$(B(A_4 \times \mathbb{Z}/2))^{\operatorname{Det}(V_{\lambda})-1} \xrightarrow{\cong} (B(A_5 \times \mathbb{Z}/2))^{\operatorname{Det}(V_{\lambda})-1}.$$

Therefore

- (1) Proposition 5.12 implies that $Ph_0^{A_5 \times \mathbb{Z}/2}(\mathbb{R}^3; f_0^D) \cong (\mathbb{Z}/2)^{\oplus 3}$; (2) Theorem 5.27 implies that $Ph_0^{A_5 \times \mathbb{Z}/2}(\mathbb{R}^3; f_{1/2}^D) \cong \mathbb{Z}/2$;
- (3) Theorem 5.29 implies that $Ph_0^{A_5 \times \mathbb{Z}/2}(\mathbb{R}^3; f_0^A) \cong \mathbb{Z}/4 \oplus (\mathbb{Z}/2)^{\oplus 3}$; and (4) Theorem 5.32 implies that $Ph_0^{A_5 \times \mathbb{Z}/2}(\mathbb{R}^3; f_{1/2}^A) \cong \mathbb{Z}/8 \oplus (\mathbb{Z}/2)^{\oplus 3}$.

6. GLIDE SYMMETRY PROTECTED PHASES

Though we have focused on point group symmetries thus far, Freed-Hopkins' ansatz [FH19a] also applies to crystallographic groups. In this section, we apply their ansatz to the group of glide symmetries; invertible phases equivariant for this symmetry have been studied by Lu-Shi-Lu [LSL17] and Xiong-Alexandradinata [XA18], and our results agree with theirs. In particular, Lu-Si-Lu make a conjecture classifying certain glide-symmetric phases in all symmetry types, and we prove that their conjecture follows from Freed-Hopkins' ansatz.

The group of glide symmetries acting on \mathbb{R}^d , $d \geq 2$, is the free group on the single generator

(6.1)
$$(x_1, x_2, \dots, x_d) \mapsto (x_1 + 1, -x_2, x_3, \dots, x_d).$$

In previous sections, when the symmetry type is H = Spin, Spin^{c} , Pin^{\pm} , etc., the symmetry type can mix with the group action on spacetime, corresponding physically to spinless or spin-1/2 fermions. Here, this cannot happen: if μ_2 denotes the kernel of the map $\operatorname{Spin}_n \to \operatorname{SO}_n$ or $\operatorname{Pin}_n^{\pm} \to \operatorname{O}_n$, all extensions

$$(6.2) 0 \longrightarrow \mu_2 \longrightarrow \widetilde{G} \longrightarrow \mathbb{Z} \longrightarrow 0$$

split, so given one of these symmetry types, there is a unique equivariant symmetry type for this Z-action with respect to mixing with fermion parity, corresponding to the trivial local system $E \to \mathbb{R}^d$ with value $E := \operatorname{Map}(MTH, \Sigma^2 I_{\mathbb{Z}}).$

Definition 6.3. Recall from Remark 1.28 that we defined a "forgetful map" $\varphi \colon Ph_*^{\mathbb{Z}}(\mathbb{R}^d;\underline{E}) \to Ph_*(\mathbb{R}^d;\underline{E})$ The intrinsically \mathbb{Z} -equivariant phase homology, denoted $\widehat{Ph}^{\mathbb{Z}}_{\mathbb{Z}}(\mathbb{R}^d; E)$, is the kernel of this map.

This corresponds under Freed-Hopkins' ansatz to what Lu-Shi-Lu call a *glide SPT*: an invertible phase equivariant for a \mathbb{Z} glide symmetry which is trivializable when one forgets the symmetry.

Let $TP_d(H)$ denote the abelian group of SPT phases in (spatial) dimension d; Freed-Hopkins' ansatz [FH16a] classifying these phases in terms of invertible field theories predicts $TP_d(H) \cong E_{-d}$.

Lu-Shi-Lu [LSL17] studied groups of glide SPTs and conjectured a formula classifying them in terms of the classification of ordinary SPTs. We prove the corresponding statement on phase homology groups.

Theorem 6.4. For a given symmetry type $\rho_n \colon H_n \to O_n$, there is a natural isomorphism $\widehat{Ph}_0^{\mathbb{Z}}(\mathbb{R}^d;\underline{E}) \cong$ $E_{-(d-1)} \otimes \mathbb{Z}/2.$

Passing this through the ansatz, this predicts that the group of glide SPTs is naturally isomorphic to $TP_{d-1}(H) \otimes \mathbb{Z}/2$, which is Lu-Shi-Lu's original conjecture [LSL17, Conjecture 1]. Xiong-Alexandradinata [XA18] also obtain this result using physics-based arguments.

Proof of Theorem 6.4. We calculate the 0th \mathbb{Z} -equivariant Borel-Moore *E*-homology of \mathbb{R}^d . As the \mathbb{Z} -action is free, this is the 0th (nonequivariant) Borel-Moore *E*-homology of the fundamental domain $X := \mathbb{R}^d/\mathbb{Z}$. Since the one-point compactification \overline{X} of X is a finite CW complex, this Borel-Moore homology is isomorphic to $\widetilde{E}_0(\overline{X})$.

If $\sigma \to S^1$ denotes the Möbius bundle, then X is diffeomorphic to the total space of $\sigma \oplus \mathbb{R}^{d-2} \to S^1$, so \overline{X} is the Thom space $(S^1)^{\sigma+d-2}$. The identification $(S^1)^{\sigma} \cong \mathbb{RP}^2$ induces $\overline{X} \cong \Sigma^{d-2} \mathbb{RP}^2$, and therefore

(6.5)
$$Ph_*^{\mathbb{Z}}(\mathbb{R}^d;\underline{E}) \cong \widetilde{E}_0(\Sigma^{d-2}\mathbb{R}\mathbb{P}^2) \cong \widetilde{E}_{2-d}(\mathbb{R}\mathbb{P}^2).$$

Lemma 6.6. Let $p: S^2 \to \mathbb{RP}^2$ be the double cover map and $s: \widetilde{E}_k(S^1) \to \widetilde{E}_{k+1}(S^2)$ be the suspension isomorphism. The composition $p_* \circ \delta \circ s: \widetilde{E}_{-1}(S^2) \to \widetilde{E}_{-1}(S^2)$ is multiplication by 2.

Proof. This follows because the suspension is the cofiber of the cofiber; then one explicitly checks what happens on mapping cylinders. \Box

Lemma 6.7. Under these isomorphisms, the forgetful map $Ph_0^{\mathbb{Z}}(\mathbb{R}^d; \underline{E}) \to Ph_0(\mathbb{R}^d; \underline{E})$ is identified with δ .

Proof. Because \mathbb{Z} acts freely on \mathbb{R}^d , $E_{0,BM}^{\mathbb{Z}}(\mathbb{R}^d)$ is identified with \widetilde{E}_0 of the one-point compactification of \mathbb{R}^d/\mathbb{Z} , which we saw above is homeomorphic to $\Sigma^{d-2}\mathbb{RP}^2$. The codomain of the forgetful map is $E_{0,BM}(\mathbb{R}^d) \cong \widetilde{E}_0(\Sigma^{d-2}S^2)$, so we have identified δ with a map $\widetilde{E}_0(\Sigma^{d-2}\mathbb{RP}^2) \to \widetilde{E}_0(\Sigma^{d-2}S^2)$. But tracing through the construction in Remark 1.28, this map comes from applying \widetilde{E}_0 to an actual map $\Sigma^{d-2}\mathbb{RP}^2 \to \Sigma^{d-2}S^2$. Next, precompose with $\Sigma^{d+2}p \colon \Sigma^{d-2}S^2 \to \Sigma^{d-2}\mathbb{RP}^2$ and check that this map has degree 2, agreeing with

Next, precompose with $\Sigma^{d+2}p: \Sigma^{d-2}S^2 \to \Sigma^{d-2}\mathbb{RP}^2$ and check that this map has degree 2, agreeing with Lemma 6.6. This suffices to identify the maps because $p^*: [\mathbb{RP}^2, S^2] \to [S^2, S^2]$ is injective.

 \mathbb{RP}^2 is homeomorphic to the cofiber of a degree-2 map $S^1 \to S^1$. Hence there is a long exact sequence in reduced *E*-homology

(6.8)
$$\cdots \longrightarrow \widetilde{E}_{2-d}(S^1) \xrightarrow{m} \widetilde{E}_{2-d}(S^1) \xrightarrow{r} \widetilde{E}_{2-d}(\mathbb{RP}^2) \xrightarrow{\delta} \widetilde{E}_{1-d}(S^1) \longrightarrow \cdots$$

where *m* is multiplication by 2. Exactness implies $\ker(\delta) = \operatorname{Im}(r) = \operatorname{coker}(m)$. Using the suspension isomorphism, $\widetilde{E}_k(S^1) \cong \widetilde{E}_{k-1}$, and therefore $\operatorname{coker}(m) \cong E_{-(d-1)} \otimes \mathbb{Z}/2$, and 6.7 identifies δ with the forgetful map from equivariant to nonequivariant phase homology for \mathbb{R}^d . In particular, $\widehat{Ph}_0^{\mathbb{Z}}(\mathbb{R}^d; \underline{E}) \cong \ker(\delta)$, which we have naturally identified with $E_{-(d-1)} \otimes \mathbb{Z}/2$.

Remark 6.9. Using the long exact sequence (6.8), we observe that $Ph_0^{\mathbb{Z}}(\mathbb{R}^d; \underline{E})$ has exponent 4. This is because for any long exact sequence of abelian groups

$$(6.10) \qquad \qquad \cdots \longrightarrow A \xrightarrow{\cdot 2} A \xrightarrow{f} B \xrightarrow{g} C \xrightarrow{\cdot 2} C \longrightarrow \cdots$$

in which A and C are finitely generated, $\text{Im}(f) \cong A/2$, hence has exponent 2, and ker(g) is isomorphic to the subgroup of order-2 elements of C, which also has exponent 2. Since B is an extension of ker(g) by Im(f), B has exponent 4.

Passing this observation through Freed-Hopkins' ansatz, this recovers an observation of Xiong-Alexandradinata [XA18]: that *any* phase equivariant with respect to glide symmetry, whether a glide SPT or not, has order dividing 4.

Example 6.11. In Altland-Zirnbauer class AII, corresponding to the symmetry type $pin^{\tilde{c}+}$, the ansatz predicts a unique nontrivial glide SPT in dimension 2 + 1, coming from the classification

$$[MTPin^{\tilde{c}+}, \Sigma^4 I_{\mathbb{Z}}] \otimes \mathbb{Z}/2 \cong \mathbb{Z}/2$$

(the calculation of $[MTPin^{\tilde{c}+}, \Sigma^4 I_{\mathbb{Z}}]$ is due to Freed-Hopkins [FH16a, §9.3]). Physicists are particularly interested in this nontrivial glide SPT phase, which is predicted to have unusual surface states called "hourglass fermions" [WACB16], and which has been studied experimentally [MYL⁺17].

7. Conclusion and outlook

We conclude by indicating a few directions of potential further research.

7.1. From free fermions to interacting phases. Free fermion phases are a rich source of examples of invertible phases in the physics literature, at least for symmetry types spin, pin^{\pm} , $spin^{c}$, etc. The classification of free fermion systems uses *K*-theory: see Kitaev [Kit09] for the original proposal and Freed-Moore [FM13] for a comprehensive classification. However, for a given dimension and symmetry type, the map from free fermion systems to invertible phases of matter can in general have both kernel (as first observed by Fidkowski-Kitaev [FK10, FK11] and Turner-Pollmann-Berg [TPB11]) and cokernel (as first observed by Wang-Potter-Senthil [WPS14] and Wang-Senthil [WS14]). Researchers are also interested in the free-to-interacting map for phases with spatial symmetries, and this map has been studied from a physics point of view for crystalline phases in several works, including [YR13, IF15, MFM15, LTH16, SS17, RL18, Zou18, LVK19, RS20, ZYQG20, ACR⁺21].

Freed-Hopkins [FH16a, §9.2, §9.3] mathematically model the map from free to interacting systems using the Atiyah-Bott-Shapiro map $MTSpin \rightarrow KO$ [ABS64], but they do not consider spatial symmetries. In view of the large bodies of research on free fermions with spatial symmetries and invertible phases with spatial symmetries, it would be nice to understand the map between them in the presence of spatial symmetry from the low-energy field theory perspective, and to make contact with the work of Adem, Antolín Camarena, Semenoff, and Sheinbaum [AACSS16], Sheinbaum and Antolín Camarena [SC20], and Cornfeld-Carmeli [CC21] studying free fermion phases with spatial symmetries using methods from homotopy theory. This is something we hope to tackle in future work.

7.2. Other symmetry types. We investigated two of the ten Altland-Zirnbauer classes, and it would be interesting to know whether a version of the FCEP holds for other classes. One starting point could be class C, corresponding to a spin^h structure [FH16a, (9.25)];³⁴ the calculations in §2.8 could be applied to Spin_n^h to obtain a fermionic crystalline equivalence principle for class C and hopefully phase homology calculations predicting the existence of additional crystalline SPT phases.

Several teams of researchers have studied or classified interacting fermionic crystalline SPTs for other Altland-Zirnbauer types, including [YR13, YX14, CHMR15, LTH16, WF17, CW18, RL18, SXG18, MSH19, ZXXS20, ZYQG20]. It would be good to compare their computations with the predictions made by an FCEP in other symmetry types.

Another interesting potential connection with preexisting work is the case of class A phases with a spatial reflection interacting with the internal U₁ symmetry. Depending on how the symmetries mix, Shiozaki-Shapourian-Gomi-Ryu [SSGR18, §V.C, §V.E] and Thorngren-Else [TE18, §VII.B] obtain classifications in terms of pin^{$\tilde{c}\pm$} bordism, and we would be interested in knowing whether that can also be obtained from our ansatz. Similarly, can one begin with class C phases and a reflection acting on the internal SU₂ symmetry and obtain a classification in terms of pin^{$h\pm$} bordism?

7.3. Crystallographic groups. Though we discussed glide symmetries in §6, we have barely touched upon the rich world of crystallographic groups. Free-fermion phases equivariant for these groups have been studied, e.g. in [SMJZ13, KdBvW⁺17, SSG18, OSS19], but much less is known about the interacting case, even though the our ansatz applies to it. There are some classifications by other methods for various classes of crystallographic groups; for example, Ouyang-Wang-Gu-Qi [OWGQ20] study wallpaper group symmetries, and Sheinbaum-Antolín Camarena [SC20] provide a general framework and a few examples. There is also work by Wang-Alexandradinata-Cava-Bernevig [WACB16] and Guo-Ohmori-Putrov-Wan-Wang [GOP⁺20] studying interacting phases for specific crystallographic groups that are not point groups.

7.4. Lattice realizations. Modeling topological phases as lattice Hamiltonian systems can make any crystallographic symmetries acting on space very explicit, using a lattice and Hamiltonian invariant under the symmetry of interest. Our predictions of point group SPTs should correspond to actual lattice models of phases. We listed several specific predicted phases of interest in §3.1, and these would make for good starting points for lattice realizations.

³⁴Spin^h is the symmetry type Spin \times_{μ_2} SU₂ \rightarrow O. Freed-Hopkins [FH16a, Proposition 9.16] call this symmetry type G^0 ; it is sometimes also called spin-SU₂, e.g. in [WWW19]. Likewise, the symmetry types pin^{h±} we refer to later in this section are defined to be Pin[±] \times_{μ_2} SU₂, and are called G^{\pm} by Freed-Hopkins [FH16a, Proposition 9.16].

References

- [AACSS16] Alejandro Adem, Omar Antolín Camarena, Gordon W. Semenoff, and Daniel Sheinbaum. Topology of Fermi surfaces and anomaly inflows. J. High Energy Phys., (11):083, front matter + 22, 2016. https://arxiv.org/abs/ 1509.01635. 4, 80
- [AB16] Mohammad Kazi Dakel Al-Boshmki. The connective K-theory of elementary abelian p-groups for odd primes. PhD thesis, The University of Sheffield, 2016. https://etheses.whiterose.ac.uk/16783/1/Al-Boshmki-thesis.pdf. 18
- [ABG10] Matthew Ando, Andrew J. Blumberg, and David Gepner. Twists of K-theory and TMF. In Superstrings, geometry, topology, and C*-algebras, volume 81 of Proc. Sympos. Pure Math., pages 27–63. Amer. Math. Soc., Providence, RI, 2010. https://arxiv.org/abs/1002.3004. 6
- [ABG⁺14a] Matthew Ando, Andrew J. Blumberg, David Gepner, Michael J. Hopkins, and Charles Rezk. An ∞-categorical approach to *R*-line bundles, *R*-module Thom spectra, and twisted *R*-homology. J. Topol., 7(3):869–893, 2014. https://arxiv.org/abs/1403.4325.4, 6, 11
- [ABG⁺14b] Matthew Ando, Andrew J. Blumberg, David Gepner, Michael J. Hopkins, and Charles Rezk. Units of ring spectra, orientations and Thom spectra via rigid infinite loop space theory. J. Topol., 7(4):1077–1117, 2014. https://arxiv.org/abs/1403.4320.4, 6, 11
- [ABG18] Matthew Ando, Andrew J. Blumberg, and David Gepner. Parametrized spectra, multiplicative Thom spectra and the twisted Umkehr map. *Geom. Topol.*, 22(7):3761–3825, 2018. https://arxiv.org/abs/1112.2203. 6
- [ABP67] D. W. Anderson, E. H. Brown, Jr., and F. P. Peterson. The structure of the Spin cobordism ring. Ann. of Math. (2), 86:271–298, 1967. 17, 36, 41
- [ABP69] D. W. Anderson, E. H. Brown, Jr., and F. P. Peterson. Pin cobordism and related topics. *Comment. Math. Helv.*, 44:462–468, 1969. 24, 25, 34, 36, 54, 63, 74
- [ABS64] M.F. Atiyah, R. Bott, and A. Shapiro. Clifford modules. Topology, 3(Supplement 1):3–38, 1964. 17, 41, 80
- [ACR⁺21] Ömer M. Aksoy, Jyong-Hao Chen, Shinsei Ryu, Akira Furusaki, and Christopher Mudry. Stability of a topological superconductor in two-dimensional space protected by time-reversal and reflection symmetries against contact interactions. 2021. https://arxiv.org/abs/2101.11609. 80
- [Ada58] J. F. Adams. On the structure and applications of the Steenrod algebra. Comment. Math. Helv., 32:180–214, 1958. 17
- [Ada61] J. F. Adams. On Chern characters and the structure of the unitary group. Proc. Cambridge Philos. Soc., 57:189–199, 1961. 18
- [AH61] M. F. Atiyah and F. Hirzebruch. Vector bundles and homogeneous spaces. In Proc. Sympos. Pure Math., Vol. III, pages 7–38. American Mathematical Society, Providence, R.I., 1961. 18, 22
- [AM04] Alejandro Adem and R. James Milgram. Cohomology of finite groups, volume 309 of Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Springer-Verlag, Berlin, second edition, 2004. 22, 49, 56, 76
- [And69] D.W. Anderson. Universal coefficient theorems for K-theory. 1969. https://faculty.tcu.edu/gfriedman/notes/ Anderson-UCT.pdf. 7, 42, 60
- [AP76] J. F. Adams and S. B. Priddy. Uniqueness of BSO. Math. Proc. Cambridge Philos. Soc., 80(3):475–509, 1976. 32, 38
- [Ati61] M. F. Atiyah. Thom complexes. Proceedings of the London Mathematical Society, s3-11(1):291–310, 1961. 6
- [AZ97] Alexander Altland and Martin R. Zirnbauer. Nonstandard symmetry classes in mesoscopic normal-superconducting hybrid structures. *Phys. Rev. B*, 55:1142–1161, Jan 1997. https://arxiv.org/abs/cond-mat/9602137. 9
- [Bak18] Andrew Baker. Iterated doubles of the joker and their realisability. *Homology Homotopy Appl.*, 20(2):341–360, 2018. https://arxiv.org/abs/1710.02974. 62
- [BB96] Dilip Bayen and Robert R. Bruner. Real connective K-theory and the quaternion group. Trans. Amer. Math. Soc., 348(6):2201–2216, 1996. 59
- [BC18] Agnès Beaudry and Jonathan A. Campbell. A guide for computing stable homotopy groups. In Topology and quantum theory in interaction, volume 718 of Contemp. Math., pages 89–136. Amer. Math. Soc., Providence, RI, 2018. https://arxiv.org/abs/1801.07530. 3, 4, 17, 18, 19, 24, 30, 32, 35, 37, 43, 50, 53, 54, 72, 74
- [BG87a] Anthony Bahri and Peter Gilkey. The eta invariant, Pin^c bordism, and equivariant Spin^c bordism for cyclic 2-groups. Pacific J. Math., 128(1):1–24, 1987. 14, 24, 25, 36, 40, 44, 47, 56, 64, 75
- [BG87b] Anthony Bahri and Peter Gilkey. Pin^c cobordism and equivariant Spin^c cobordism of cyclic 2-groups. Proceedings of the American Mathematical Society, 99(2):380–382, 1987. 14, 24, 25, 26, 28, 36, 40, 44, 47, 56, 64, 75
- [BG97] Boris Botvinnik and Peter Gilkey. The Gromov-Lawson-Rosenberg conjecture: the twisted case. *Houston J. Math.*, 23(1):143–160, 1997. 27
- [BG03] R. R. Bruner and J. P. C. Greenlees. The connective K-theory of finite groups. Mem. Amer. Math. Soc., 165(785):viii+127, 2003. 17, 18, 47, 51, 60
- [BG10] Robert R. Bruner and J. P. C. Greenlees. Connective real K-theory of finite groups, volume 169 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2010. 17, 26, 27, 44, 50, 61, 76
- [BGS97] Boris Botvinnik, Peter Gilkey, and Stephan Stolz. The Gromov-Lawson-Rosenberg conjecture for groups with periodic cohomology. J. Differential Geom., 46(3):374–405, 1997. 27
- [BLT20] Lakshya Bhardwaj, Yasunori Lee, and Yuji Tachikawa. SL(2, Z) action on QFTs with Z₂ symmetry and the Brown-Kervaire invariants. Journal of High Energy Physics, 2020(141), 2020. https://arxiv.org/abs/2009.10099. 19

BM60]	A. Borel and J. C. Moore. Homology theory for locally compact spaces. Michigan Math. J., 7:137–159, 1960. 6
BM19	Vincent Braunack-Mayer. Combinatorial parametrised spectra. 2019. https://arxiv.org/abs/1907.08496. 6
BMSS15]	Robert Bruner, Khairia Mira, Laura Stanley, and Victor Snaith. Ossa's theorem via the Kunneth formula.
Br1100]	Robert B Bruner Osso's theorem and Adams covers Proc. Amer. Math. Soc. 127(8):2443-2447 1000 http://
Diuss	//www.rrb.wavne.edu/papers/osac2.pdf. 47
[BWHV17]	Nick Bultinck, Dominic J. Williamson, Jutho Haegeman, and Frank Verstraete. Fermionic matrix product states
	and one-dimensional topological phases. <i>Phys. Rev. B</i> , 95:075108, Feb 2017. https://arxiv.org/abs/1610.07849. 24
[Čad99]	Martin Čadek. The cohomology of $BO(n)$ with twisted integer coefficients. J. Math. Kyoto Univ., $39(2)$:277–286, 1999. 60
[Cam17]	Jonathan A. Campbell. Homotopy theoretic classification of symmetry protected phases. 2017. https://arxiv. org/abs/1708.04264. 1, 18, 19, 25, 27, 32, 33, 43, 46, 59, 60, 71
Car65]	H. Cardenas. El álgebra de cohomología del grupo simetrico de grado p^2 . Bol. Soc. Mat. Mex., 10:1–30, 1965. 56
[CC21]	Eyal Cornfeld and Shachar Carmeli. Tenfold topology of crystals: Unified classification of crystalline topological insulators and superconductors. <i>Phys. Rev. Research</i> , 3:013052, Jan 2021. https://arxiv.org/abs/2009.04486.
CHMR15]	Gil Young Cho, Chang-Tse Hsieh, Takahiro Morimoto, and Shinsei Ryu. Topological phases protected by reflection
	symmetry and cross-cap states. Phys. Rev. B, 91:195142, May 2015. https://arxiv.org/abs/1501.07285. 80
COSY20]	Clay Córdova, Kantaro Ohmori, Shu-Heng Shao, and Fei Yan. Decorated \mathbb{Z}_2 symmetry defects and their time-reversal anomalies. <i>Phys. Rev. D</i> , 102:045019, Aug 2020. https://arxiv.org/abs/1910.14046. 25
CW18]	Meng Cheng and Chenjie Wang. Rotation symmetry-protected topological phases of fermions. 2018. https://www.anglebe/1910.10200.0.2.10.12.02.06.08.80
Dav74]	Donald Davis Generalized homology and the generalized vector field problem <i>Quart I Math Orford Ser (2)</i>
Davij	25:169–193, 1974. 18
DDHM]	Arun Debray, Markus Dierigl, Jonathan Heckman, and Miguel Montero. In preparation. 19
Deb20]	Arun Debray. The low-energy TQFT of the generalized double semion model. Comm. Math. Phys., 375(2):1079–
DGG21]	1115, 2020. https://arxiv.org/abs/1811.03583. 20, 76 Diego Delmastro, Davide Gajotto, and Jaume Gomis, Global anomalies on the Hilbert space, 2021, https://
00021]	//arxiv.org/abs/2101.02218. 19
[DGL20]	Joe Davighi, Ben Gripaios, and Nakarin Lohitsiri. Global anomalies in the standard model(s) and beyond. J. High Energy Phys., (7):232, 50, 2020. https://arxiv.org/abs/1910.11277. 19
[DH20]	Markus Dierigl and Jonathan J. Heckman. On the swampland cobordism conjecture and non-abelian duality groups. 2020. https://arxiv.org/abs/2012.00013. 19
[DL20a]	Joe Davighi and Nakarin Lohitsiri. The algebra of anomaly interplay. 2020. https://arxiv.org/abs/2011.10102. 18, 25, 27, 32, 46
DL20b]	Joe Davighi and Nakarin Lohitsiri. Anomaly interplay in U(2) gauge theories. J. High Energy Phys., (5):098, 20, 2020. https://arxiv.org/abs/2001.07731. 18
[DL20c]	Joe Davighi and Nakarin Lohitsiri. Omega vs. pi, and 6d anomaly cancellation. 2020. https://arxiv.org/abs/2012.11693. 18, 19
[DM81]	Donald M. Davis and Mark Mahowald. v_1 - and v_2 -periodicity in stable homotopy theory. Amer. J. Math., $103(4):615-659, 1981.$ 36
Dys62]	Freeman J. Dyson. The threefold way. Algebraic structure of symmetry groups and ensembles in quantum mechanics. J. Mathematical Phys., 3:1199–1215, 1962. 9
Edw91] ET19]	Steven R. Edwards. On the spin bordism of $B(E_8 \times E_8)$. Illinois J. Math., 35(4):683–689, 1991. 19 Dominic V. Else and Ryan Thorngren. Crystalline topological phases as defect networks. <i>Phys. Rev. B</i> , 99:115116.
	Mar 2019. https://arxiv.org/abs/1810.10539. 12, 19
ETS20]	Dominic V. Else, Ryan Thorngren, and T. Senthil. Non-Fermi liquids as ersatz Fermi liquids: general constraints on compressible metals. 2020. https://arxiv.org/abs/2007.07896. 19
[FH16a]	Daniel S. Freed and Michael J. Hopkins. Reflection positivity and invertible topological phases. 2016. https: //arxiv.org/abs/1604.06527. 1, 2, 3, 4, 6, 7, 9, 10, 12, 13, 14, 18, 20, 23, 28, 33, 54, 72, 74, 78, 79, 80
FH16b]	Michael H. Freedman and Matthew B. Hastings. Double semions in arbitrary dimension. Communications in Mathematical Physics, 347(2):389–419, 2016. https://arxiv.org/abs/1507.05676. 76
FH19a]	Daniel S. Freed and Michael J. Hopkins. Invertible phases of matter with spatial symmetry. 2019. https://arxiv.org/abs/1901_06/19_1_2_4_5_6_7_10_12_13_19_23_25_27_28_42_59_78
[FH19b]	Daniel S. Freed and Michael J. Hopkins. M-theory anomaly cancellation. 2019. https://arxiv.org/abs/1908. 09916. 18, 20
FHHT20]	Lukasz Fidkowski, Jeongwan Haah, Matthew B. Hastings, and Nathanan Tantivasadakarn. Disentangling the generalized double semion model. <i>Comm. Math. Phys.</i> , 380(3):1151–1171, 2020. https://arxiv.org/abs/1906.
[FK10]	U4188. 1, (0 Lukasz Fidkowski and Alexei Kitaev. Effects of interactions on the topological classification of free fermion systems.
[FK11]	Lukasz Fidkowski and Alexei Kitaev. Topological phases of fermions in one dimension. <i>Phys. Rev. B</i> , 83(7):075103, Feb 2011. https://arxiv.org/abs/1008.4138. 80

- 83
- [FM13] Daniel S. Freed and Gregory W. Moore. Twisted equivariant matter. Annales Henri Poincaré, 14(8):1927–2023, Dec 2013. https://arxiv.org/abs/1208.5055. 9, 80
- [FMS07] Daniel S. Freed, Gregory W. Moore, and Graeme Segal. The uncertainty of fluxes. Communications in Mathematical Physics, 271(1):247–274, Apr 2007. https://arxiv.org/abs/hep-th/0605198. 42, 60
- [FQ93] Daniel S. Freed and Frank Quinn. Chern-Simons theory with finite gauge group. Comm. Math. Phys., 156(3):435–472, 1993. https://arxiv.org/abs/hep-th/9111004. 20
- [Fral1] John Francis. Integrals on spin manifolds and the K-theory of $K(\mathbb{Z}, 4)$. 2011. https://sites.math.northwestern. edu/~jnkf/writ/bspin2011.pdf. 18
- [Fre19] Daniel S. Freed. Lectures on Field theory and topology, volume 133 of CBMS Regional Conference Series in Mathematics. American Mathematical Society, Providence, RI, 2019. Published for the Conference Board of the Mathematical Sciences. 2, 21, 28, 76
- [GEM19] Iñaki García-Etxebarria and Miguel Montero. Dai-Freed anomalies in particle physics. Journal of High Energy Physics, 2019(8):3, Aug 2019. https://arxiv.org/abs/1808.00009. 19, 25, 26, 27, 44
- [Gia73a] V. Giambalvo. Cobordism of line bundles with a relation. *Illinois J. Math.*, 17:442–449, 1973. 25
- [Gia73b] V. Giambalvo. Pin and Pin' cobordism. Proc. Amer. Math. Soc., 39:395-401, 1973. 24, 25, 27, 30
- [GJF19]
 Davide Gaiotto and Theo Johnson-Freyd. Symmetry protected topological phases and generalized cohomology. Journal of High Energy Physics, 2019, 2019. https://arxiv.org/abs/1712.07950. 41
- [GM20] Roman Geiko and Gregory W. Moore. Dyson's classification and real division superalgebras. 2020. https: //arxiv.org/abs/2010.01675.9
- [GMM68] S. Gitler, M. Mahowald, and R. James Milgram. The nonimmersion problem for RPⁿ and higher-order cohomology operations. Proc. Nat. Acad. Sci. U.S.A., 60:432–437, 1968. 30
- [GOP+20] Meng Guo, Kantaro Ohmori, Pavel Putrov, Zheyan Wan, and Juven Wang. Fermionic finite-group gauge theories and interacting symmetric/crystalline orders via cobordisms. Comm. Math. Phys., 376(2):1073-1154, 2020. https://arxiv.org/abs/1812.11959. 1, 12, 17, 18, 19, 24, 25, 26, 28, 34, 39, 80
- [GPW18] Meng Guo, Pavel Putrov, and Juven Wang. Time reversal, SU(N) Yang-Mills and cobordisms: Interacting topological superconductors/insulators and quantum spin liquids in 3+1d. Annals of Physics, 394:244-293, 2018. https://arxiv.org/abs/1711.11587.18
- [Guo18] Meng Guo. Some Calculations of Cobordism Groups and Their Applications in Physics. PhD thesis, Harvard University, 2018. https://dash.harvard.edu/bitstream/handle/1/40050106/GUO-DISSERTATION-2018. pdf?sequence=4. 18
- [Han93] David Handel. On products in the cohomology of the dihedral groups. *Tohoku Math. J. (2)*, 45(1):13–42, 1993. 28, 29, 31, 35, 43, 44
- [Has83] Shin Hashimoto. On the connective K-homology groups of the classifying spaces BZ/p^r . Publ. Res. Inst. Math. Sci., 19(2):765–771, 1983. 47, 51, 70
- [HCR16] Chang-Tse Hsieh, Gil Young Cho, and Shinsei Ryu. Global anomalies on the surface of fermionic symmetryprotected topological phases in (3+1) dimensions. *Phys. Rev. B*, 93:075135, Feb 2016. https://arxiv.org/abs/ 1503.01411. 24
- [HF17] Sungjoon Hong and Liang Fu. Topological order and symmetry anomaly on the surface of topological crystalline insulators. 2017. https://arxiv.org/abs/1707.02594. 24
- [HH18] Sheng-Jie Huang and Michael Hermele. Surface field theories of point group symmetry protected topological phases. Phys. Rev. B, 97:075145, Feb 2018. https://arxiv.org/abs/1711.03649. 19
- [HH20] Sheng-Jie Huang and Yi-Ting Hsu. Faithful derivation of symmetry indicators: A case study for topological superconductors with time-reversal and inversion symmetries. 2020. https://arxiv.org/abs/2010.05947. 19
- [HKT20] Itamar Hason, Zohar Komargodski, and Ryan Thorngren. Anomaly Matching in the Symmetry Broken Phase: Domain Walls, CPT, and the Smith Isomorphism. SciPost Phys., 8:62, 2020. https://arxiv.org/abs/1910.14039. 19, 25
- [HLN20] Fabian Hebestreit, Markus Land, and Thomas Nikolaus. On the homotopy type of L-spectra of the integers. Journal of Topology, 14(1):183–214, 2020. https://arxiv.org/abs/2004.06889. 42, 60
- [Hor20] Yusuke Horinouchi. Solvable lattice model for (2+1)D bosonic topological insulator. 2020. https://arxiv.org/ abs/2002.01639. 19
- [HS53] G. Hochschild and J.-P. Serre. Cohomology of group extensions. Trans. Amer. Math. Soc., 74:110–134, 1953. 49, 60
- [HS14] Drew Heard and Vesna Stojanoska. K-theory, reality, and duality. Journal of K-theory, 14(3):526–555, 2014. https://arxiv.org/abs/1401.2581. 42, 60
- [HSHH17] Sheng-Jie Huang, Hao Song, Yi-Ping Huang, and Michael Hermele. Building crystalline topological phases from lower-dimensional states. Phys. Rev. B, 96:205106, Nov 2017. https://arxiv.org/abs/1705.09243. 19
- [Hsi18] Chang-Tse Hsieh. Discrete gauge anomalies revisited. 2018. https://arxiv.org/abs/1808.02881. 19, 25
 [HTY20] Chang-Tse Hsieh, Yuji Tachikawa, and Kazuya Yonekura. Anomaly inflow and *p*-form gauge theories. 2020. https://arxiv.org/abs/2003.11550. 19
- [HWL20] Huaqing Huang, Yong-Shi Wu, and Feng Liu. Aperiodic topological crystalline insulators. Phys. Rev. B, 101:041103, Jan 2020. https://my.eng.utah.edu/~fliu/pdfs/PhysRevB.101.041103.pdf. 76
- [IF15] Hiroki Isobe and Liang Fu. Theory of interacting topological crystalline insulators. Phys. Rev. B, 92:081304, Aug 2015. https://arxiv.org/abs/1502.06962. 24, 80

Takumi	Ichikawa	and	Yuji	Tachikav	va.

[IT20] Takumi Ichikawa and Yuji Tachikawa. The super Frobenius-Schur indicator and finite group gauge theories on pin⁻ surfaces. 2020. https://arxiv.org/abs/2002.10642. 9

ARUN DEBRAY

- [JF20a]Theo Johnson-Freyd. On the classification of topological orders. 2020. https://arxiv.org/abs/2003.06663. 19[JF20b]Theo Johnson-Freyd. Topological Mathieu moonshine. 2020. https://arxiv.org/abs/2006.02922. 19
- [Joy02] A. Joyal. Quasi-categories and Kan complexes. volume 175, pages 207–222. 2002. Special volume celebrating the
- 70th birthday of Professor Max Kelly. 6 [JR17] Shenghan Jiang and Ying Ran. Anyon condensation and a generic tensor-network construction for symmetry-
- protected topological phases. Phys. Rev. B, 95:125107, Mar 2017. https://arxiv.org/abs/1611.07652. 2, 12
- [JW97] David Copeland Johnson and W. Stephen Wilson. On a theorem of Ossa. Proc. Amer. Math. Soc., 125(12):3753– 3755, 1997. https://math.jhu.edu/~wsw/papers2/math/31-ossa-J8W-1997.pdf. 47
- [KdBvW⁺17] Jorrit Kruthoff, Jan de Boer, Jasper van Wezel, Charles L. Kane, and Robert-Jan Slager. Topological classification of crystalline insulators through band structure combinatorics. *Phys. Rev. X*, 7:041069, Dec 2017. https://arxiv. org/abs/1612.02007. 80
- [Kil88] T. P. Killingback. Global anomalies, string theory and spacetime topology. Classical Quantum Gravity, 5(9):1169– 1185, 1988. With an appendix by W. Rose. 18, 19
- [Kim18] Minkyu Kim. A generalization of Dijkgraaf-Witten theory. 2018. https://arxiv.org/abs/1810.03117. 20
- [Kin] Simon King. Modular cohomology rings of finite groups. https://users.fmi.uni-jena.de/~king/cohomology/. 49
- [Kit09] Alexei Kitaev. Periodic table for topological insulators and superconductors. AIP Conference Proceedings, 1134(1):22–30, 2009. https://arxiv.org/abs/0901.2686. 9, 80
- [Kit13] Alexei Kitaev. On the classification of short-range entangled states. Conference talk at the Simons Center. http://scgp.stonybrook.edu/archives/7874., 2013. 1
- [Kit15] Alexei Kitaev. Homotopy-theoretic approach to SPT phases in action: Z_16 classification of three-dimensional superconductors. Conference talk at the Institute for Pure and Applied Mathematics. http://www.ipam.ucla.edu/abstract/?tid=12389, 2015. 1
- [KLST21] Ryohei Kobayashi, Yasunori Lee, Ken Shiozaki, and Yuya Tanizaki. Topological terms of (2+1)d flag-manifold sigma models. 2021. https://arxiv.org/abs/2103.05035. 19
- [KPMT20] Justin Kaidi, Julio Parra-Martinez, and Yuji Tachikawa. Topological superconductors on superstring worldsheets. SciPost Phys., 9:10, 2020. With a mathematical appendix by Arun Debray. https://arxiv.org/abs/1911.11780.
 18, 19
- [KT90a] R. C. Kirby and L. R. Taylor. A calculation of Pin⁺ bordism groups. Commentarii Mathematici Helvetici, 65(1):434–447, Dec 1990. 14, 24, 25, 30
- [KT90b] R. C. Kirby and L. R. Taylor. Pin structures on low-dimensional manifolds, volume 151 of London Math. Soc. Lecture Note Ser., pages 177–242. Cambridge Univ. Press, 1990. 24, 25, 30, 34, 36, 54, 63, 74
- [KT17] Anton Kapustin and Ryan Thorngren. Fermionic SPT phases in higher dimensions and bosonization. Journal of High Energy Physics, 2017(10):80, Oct 2017. https://arxiv.org/abs/1701.08264. 19, 41
- [KTTW15] Anton Kapustin, Ryan Thorngren, Alex Turzillo, and Zitao Wang. Fermionic symmetry protected topological phases and cobordisms. J. High Energy Phys., (12):052, front matter+20pp, 2015. https://arxiv.org/abs/1406.7329.
 23
- [KZ16] R. Kennedy and M. R. Zirnbauer. Bott periodicity for Z₂ symmetric ground states of gapped free-fermion systems. Comm. Math. Phys., 342(3):909-963, 2016. https://arxiv.org/abs/1409.2537.9
- [LOT20] Yasunori Lee, Kantaro Ohmori, and Yuji Tachikawa. Revisiting Wess-Zumino-Witten terms. 2020. https: //arxiv.org/abs/2009.00033. 18, 19
- [LSL17] Fuyan Lu, Bowen Shi, and Yuan-Ming Lu. Classification and surface anomaly of glide symmetry protected topological phases in three dimensions. New Journal of Physics, 19(7):073002, jul 2017. https://arxiv.org/abs/ 1701.00784. 4, 6, 78
- [LT20] Yasunori Lee and Yuji Tachikawa. Some comments on 6d global gauge anomalies. 2020. https://arxiv.org/abs/ 2012.11622. 18, 19
- [LTH16] Matthew F. Lapa, Jeffrey C. Y. Teo, and Taylor L. Hughes. Interaction-enabled topological crystalline phases. *Phys. Rev. B*, 93:115131, Mar 2016. https://arxiv.org/abs/1409.1234. 80
- [Lur09] Jacob Lurie. *Higher topos theory*, volume 170 of *Annals of Mathematics Studies*. Princeton University Press, Princeton, NJ, 2009. https://people.math.harvard.edu/~lurie/papers/highertopoi.pdf. 6
- [LVK19]Shang Liu, Ashvin Vishwanath, and Eslam Khalaf. Shift insulators: Rotation-protected two-dimensional topological
crystalline insulators. Phys. Rev. X, 9:031003, Jul 2019. https://arxiv.org/abs/1809.01636. 28, 80
- [Lyn48] Roger C. Lyndon. The cohomology theory of group extensions. Duke Math. J., 15:271–292, 1948. 49, 60
- [Mah82] Mark Mahowald. The image of J in the EHP sequence. Ann. of Math. (2), 116(1):65–112, 1982. Correction in Annals of Mathematics, 120:399–400, 1984. 27, 44
- [Mal11] Arjun Malhotra. The Gromov-Lawson-Rosenberg conjecture for some finite groups. PhD thesis, The University of Sheffield, 2011. https://arxiv.org/abs/1305.0455. 31
- [Mar74] H. R. Margolis. Eilenberg-Mac Lane spectra. Proc. Amer. Math. Soc., 43:409–415, 1974. 22
- [Mau63] C. R. F. Maunder. The spectral sequence of an extraordinary cohomology theory. *Proc. Cambridge Philos. Soc.*, 59:567–574, 1963. 61
- [May77] J. Peter May. E_{∞} ring spaces and E_{∞} ring spectra, volume 577 of Lecture Notes in Mathematics. Springer-Verlag, Berlin-New York, 1977. With contributions by Frank Quinn, Nigel Ray, and Jørgen Tornehave. https: //www.math.uchicago.edu/-may/BOOKS/e_infty.pdf. 6, 11

- $[Met15] Max A. Metlitski. S-duality of u(1) gauge theory with <math>\theta = \pi$ on non-orientable manifolds: Applications to topological insulators and superconductors. 2015. https://arxiv.org/abs/1510.05663. 33
- [MFM15] Takahiro Morimoto, Akira Furusaki, and Christopher Mudry. Breakdown of the topological classification Z for gapped phases of noninteracting fermions by quartic interactions. *Phys. Rev. B*, 92:125104, Sep 2015. https://arxiv.org/abs/1505.06341. 80
- [Mil95] John Milnor. On the Steenrod homology theory. In Novikov conjectures, index theorems and rigidity, Vol. 1 (Oberwolfach, 1993), volume 226 of London Math. Soc. Lecture Note Ser., pages 79–96. Cambridge Univ. Press, Cambridge, 1995. 6
- [MM76] M. Mahowald and R. James Milgram. Operations which detect Sq^4 in connective K-theory and their applications. Quart. J. Math. Oxford Ser. (2), 27(108):415-432, 1976. 25, 27, 44, 68
- [MM81] J. P. May and R. J. Milgram. The Bockstein and the Adams spectral sequences. Proc. Amer. Math. Soc., 83(1):128–130, 1981. 32, 39, 41, 46, 48
- [MM19] Samuel Monnier and Gregory W. Moore. Remarks on the Green-Schwarz terms of six-dimensional supergravity theories. Comm. Math. Phys., 372(3):963-1025, 2019. https://arxiv.org/abs/1808.01334. 19
- [Mon15] Samuel Monnier. Global gravitational anomaly cancellation for five-branes. Advances in Theoretical and Mathematical Physics, 19(3):701-724, 2015. https://arxiv.org/abs/1310.2250. 19
- [Mon17] Samuel Monnier. The anomaly field theories of six-dimensional (2,0) superconformal theories. 2017. https://arxiv.org/abs/1706.01903. 19
- [MS06] J. P. May and J. Sigurdsson. Parametrized homotopy theory, volume 132 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2006. 6
- [MSH19] Ian Mondragon-Shem and Taylor L. Hughes. Robust topological invariants of topological crystalline phases in the presence of impurities. 2019. https://arxiv.org/abs/1906.11847. 80
- [MT01] Ib Madsen and Ulrike Tillmann. The stable mapping class group and $Q(\mathbb{CP}^{\infty}_{+})$. Invent. Math., 145(3):509–544, 2001. 7
- [MV21] Miguel Montero and Cumrun Vafa. Cobordism conjecture, anomalies, and the string lampost principle. Journal of High Energy Physics, 2021(63), 2021. https://arxiv.org/abs/2008.11729. 25
- [MW07] Ib Madsen and Michael Weiss. The stable moduli space of Riemann surfaces: Mumford's conjecture. Ann. of Math. (2), 165(3):843-941, 2007. https://arxiv.org/abs/math/0212321.7
- [MW20] Bin-Bin Mao and Chenjie Wang. Mirror anomaly in fermionic topological orders. Phys. Rev. Research, 2:023339, Jun 2020. https://arxiv.org/abs/2002.07714. 12, 24
- [MYL⁺17] Junzhang Ma, Changjiang Yi, Baiqing Lv, ZhiJun Wang, Simin Nie, Le Wang, Lingyuan Kong, Yaobo Huang, Pierre Richard, Peng Zhang, Koichiro Yaji, Kenta Kuroda, Shik Shin, Hongming Weng, Bogdan Andrei Bernevig, Youguo Shi, Tian Qian, and Hong Ding. Experimental evidence of hourglass fermion in the candidate nonsymmorphic topological insulator KHgSb. Science Advances, 3(5), 2017. https://arxiv.org/abs/1605.06824. 1, 79
- [Ngu09] Son Phuc Nguyen. Calculations towards the complex connective K-theory of QS(0). PhD thesis, Wayne State University, 2009. 18, 56, 70
- [Oss89] E. Ossa. Connective K-theory of elementary abelian groups. In Transformation groups (Osaka, 1987), volume 1375 of Lecture Notes in Math., pages 269–275. Springer, Berlin, 1989. 47
- [OSS19] Nobuyuki Okuma, Masatoshi Sato, and Ken Shiozaki. Topological classification under nonmagnetic and magnetic point group symmetry: Application of real-space Atiyah-Hirzebruch spectral sequence to higher-order topology. *Phys. Rev. B*, 99:085127, Feb 2019. https://arxiv.org/abs/1810.12601. 19, 80
- [OWGQ20] Yunqing Ouyang, Qing-Rui Wang, Zheng-Cheng Gu, and Yang Qi. Computing classification of interacting fermionic symmetry-protected topological phases using topological invariants. 2020. https://arxiv.org/abs/2005.06572. 80
- [Ped17] Riccardo Pedrotti. Stable classification of certain families of four-manifolds. Master's thesis, Max Planck Institute for Mathematics, 2017. 43, 44
- [Pet68] F. P. Peterson. Lectures on Cobordism Theory. Lectures in Mathematics. Kinokuniya Book Store Co., Ltd., 1968. 14
- [Pow14] Geoffrey M. L. Powell. On connective K-theory of elementary abelian 2-groups and local duality. Homology Homotopy Appl., 16(1):215–243, 2014. 47
- [PW20] Abhishodh Prakash and Juven Wang. Unwinding fermionic SPT phases: Supersymmetry extension. 2020. https://arxiv.org/abs/2011.13921.
- [Ran10] Ying Ran. Weak indices and dislocations in general topological band structures. 2010. https://arxiv.org/abs/ 1006.5454. 1
- [Ric16] Nicolas Ricka. Equivariant Anderson duality and Mackey functor duality. Glasg. Math. J., 58(3):649-676, 2016. https://arxiv.org/abs/1408.1581. 42, 60
- [RL18] Alex Rasmussen and Yuan-Ming Lu. Intrinsically interacting topological crystalline insulators and superconductors. 2018. https://arxiv.org/abs/1810.12317. 80
- [Roh52] V. A. Rohlin. New results in the theory of four-dimensional manifolds. Doklady Akad. Nauk SSSR (N.S.), 84:221–224, 1952. 21
- [RS17] George Raptis and Wolfgang Steimle. Parametrized cobordism categories and the Dwyer-Weiss-Williams index theorem. J. Topol., 10(3):700-719, 2017. https://arxiv.org/abs/1606.07925.6

	states and its relation to topological superconductivity from an exact lattice \mathbb{Z}_2 charge-flux attachment. 2020.
	https://arxiv.org/abs/2012.12280.80
[RSFL10]	Shinsei Ryu, Andreas P Schnyder, Akira Furusaki, and Andreas W W Ludwig. Topological insulators and
	superconductors: tenfold way and dimensional hierarchy. New J. Phys, 12:065010, 2010. https://arxiv.org/abs/
	0912.2157. 9
[SC20]	Daniel Sheinbaum and Omar Antolín Camarena. Crystallographic interacting topological phases and equvariant
	cohomology: To assume or not to assume. 2020. https://arxiv.org/abs/2007.06595. 4, 80
[SdBKP18]	Luuk Stehouwer, Jan de Boer, Jorrit Kruthoff, and Hessel Posthuma. Classification of crystalline topological
	insulators through K-theory. 2018. https://arxiv.org/abs/1811.02592. 19
[Ser50]	Jean-Pierre Serre. Cohomologie des extensions de groupes. C. R. Acad. Sci. Paris, 231:643–646, 1950. 49, 60
[Ser53]	Jean-Pierre Serre. Groupes d'homotopie et classes de groupes abéliens. Annals of Mathematics, 58(2):258–294, 1953. 52, 55, 68, 76
[SFQ20]	Zhida Song, Chen Fang, and Yang Qi. Real-space recipes for general topological crystalline states. 11(4197), 2020.
	https://arxiv.org/abs/1810.11013.19
[SHFH17]	Hao Song, Sheng-Jie Huang, Liang Fu, and Michael Hermele. Topological phases protected by point group
	symmetry. Phys. Rev. X, 7:011020, Feb 2017. https://arxiv.org/abs/1604.08151. 19, 23, 24, 26
[Shi19]	Ken Shiozaki. Variants of the symmetry-based indicator. 2019. https://arxiv.org/abs/1907.13632. 19
$[SHQ^+19]$	Zhida Song, Sheng-Jie Huang, Yang Qi, Chen Fang, and Michael Hermele. Topological states from topological
	crystals. Science Advances, 5(12), 2019. https://arxiv.org/abs/1810.02330. 19
[Sie13]	Christian Siegemeyer. On the Gromov-Lawson-Rosenberg Conjecture for Finite Abelian 2-Groups of Rank 2. PhD
	thesis, Westfälische Wilhelms-Universität Münster, 2013. https://d-nb.info/1031885560/34. 27, 44
[SMJZ13]	Robert-Jan Slager, Andrej Mesaros, Vladimir Juričić, and Jan Zaanen. The space group classification of topological
	band-insulators. Nature Physics, 9:98–102, 2013. https://arxiv.org/abs/1209.2610. 1, 80
[Sna13]	Victor P Snaith. Topological methods in Galois representation theory. Dover Books on Mathematics. Dover,
	Mineola, NY, 2013. 28
[SS17]	Xue-Yang Song and Andreas P. Schnyder. Interaction effects on the classification of crystalline topological
	insulators and superconductors. Phys. Rev. B, 95:195108, May 2017. https://arxiv.org/abs/1609.07469.80
[SSG18]	Ken Shiozaki, Masatoshi Sato, and Kiyonori Gomi. Atiyah-Hirzebruch spectral sequence in band topology: General
	formalism and topological invariants for 230 space groups. 2018. https://arxiv.org/abs/1802.06694. 19, 80
[SSGR18]	Ken Shiozaki, Hassan Shapourian, Kiyonori Gomi, and Shinsei Ryu. Many-body topological invariants for fermionic
	short-range entangled topological phases protected by antiunitary symmetries. Phys. Rev. B, 98:035151, Jul 2018.
	https://arxiv.org/abs/1710.01886.24,80
[SSR17a]	Hassan Shapourian, Ken Shiozaki, and Shinsei Ryu. Many-body topological invariants for fermionic symmetry-
	protected topological phases. Phys. Rev. Lett., 118:216402, May 2017. https://arxiv.org/abs/1607.03896. 24,
	26
[SSR17b]	Ken Shiozaki, Hassan Shapourian, and Shinsei Ryu. Many-body topological invariants in fermionic symmetry-
	protected topological phases: Cases of point group symmetries. Phys. Rev. B, 95:205139, May 2017. https://
	//arxiv.org/abs/1609.05970. 23, 24, 26, 28
[Sta63]	James Stasheff. A classification theorem for fibre spaces. <i>Topology</i> , 2:239–246, 1963. 11
[Ste18]	Luuk Stehouwer. K-theory classifications for symmetry-protected topological phases of free fermions. Master's
	thesis, Korteweg-de Vries Institute for Mathematics, 2018. 19
[Sto63]	Robert E. Stong. Determination of $H^*(BO(k, \dots, \infty), Z_2)$ and $H^*(BU(k, \dots, \infty), Z_2)$. Trans. Amer. Math. Soc., 107:526–544, 1963. 18
[Sto86]	R. E. Stong. Appendix: calculation of $\Omega_{11}^{\text{Spin}}(K(Z,4))$. In Workshop on unified string theories (Santa Barbara,
	Calif., 1985), pages 430–437. World Sci. Publishing, Singapore, 1986. 18
[STY18]	Nathan Seiberg, Yuji Tachikawa, and Kazuya Yonekura. Anomalies of duality groups and extended conformal
	manifolds. PTEP. Prog. Theor. Exp. Phys., (7):073B04, 27, 2018. https://arxiv.org/abs/1803.07366. 19
[Sul20]	Yuri J. F. Sulyma. A slice refinement of Bökstedt periodicity. 2020. https://arxiv.org/abs/2007.13817. 7
[SW16]	Nathan Seiberg and Edward Witten. Gapped boundary phases of topological insulators via weak coupling. Progress
	of Theoretical and Experimental Physics, 2016(12), 11 2016. https://arxiv.org/abs/1602.04251. 33
[Swa60]	Richard G. Swan. The <i>p</i> -period of a finite group. <i>Illinois J. Math.</i> , 4:341–346, 1960. 22
[SXG18]	Ken Shiozaki, Charles Zhaoxi Xiong, and Kiyonori Gomi. Generalized homology and Atiyah-Hirzebruch spectral
	sequence in crystalline symmetry protected topological phenomena. 2018. https://arxiv.org/abs/1810.00801.
	19, 26, 28, 80
[SXH20]	Hao Song, Charles Zhaoxi Xiong, and Sheng-Jie Huang. Bosonic crystalline symmetry protected topological phases
	beyond the group cohomology proposal. Phys. Rev. B, 101:165129, Apr 2020. https://arxiv.org/abs/1811.06558.
	19
[TE18]	Ryan Thorngren and Dominic V. Else. Gauging spatial symmetries and the classification of topological crystalline
[77. 10.0]	phases. Phys. Rev. X, 8:011040, Mar 2018. https://arxiv.org/abs/1612.00846. 2, 3, 12, 13, 19, 23, 80

Peng Rao and Inti Sodemann. Theory of weak symmetry breaking of translations in \mathbb{Z}_2 topologically ordered

86

[RS20]

[Tei92]Peter Teichner. Topological 4-manifolds with finite fundamental group. PhD thesis, University of Mainz, 1992.
https://math.berkeley.edu/~teichner/Papers/phd.pdf. 28

[Tho54] René Thom. Quelques propriétés globales des variétés differentiables. PhD thesis, University of Paris, 1954. 6, 21
 [Tho74] C. B. Thomas. The integral cohomology ring of S₄. Mathematika, 21:228-232, 1974. 61, 65, 68

[Tho 20]	Ryan Thorngren. Anomalies and bosonization. Comm. Math. Phys., 378(3):1775-1816, 2020. https://arxiv.org/
[TPB11]	Ari M. Turner, Frank Pollmann, and Erez Berg. Topological phases of one-dimensional fermions: An entanglement point of view. <i>Phys. Rev. B</i> , 83(7):075102, Feb 2011, https://arxiv.org/abs/1008.4346, 80
[TW20]	Ryan Thorngren and Yifan Wang. Anomalous symmetries end at the boundary. 2020. https://arxiv.org/abs/ 2012.15861. 19
[TY19]	Yuji Tachikawa and Kazuya Yonekura. Why are fractional charges of orientifolds compatible with Dirac quantiza- tion? SciPost Phys., 7:58, 2019, https://arxiv.org/abs/1805.02772, 19, 25
[VLP+19]	Dániel Varjas, Alexander Lau, Kim Pöyhönen, Anton R. Akhmerov, Dmitry I. Pikulin, and Ion Cosma Fulga. Topological phases without crystalline counterparts. <i>Phys. Rev. Lett.</i> , 123:196401, Nov 2019. https://arxiv.org/ abs/1904.07242.76
[WACB16]	Zhijun Wang, A. Alexandradinata, R. J. Cava, and B. Andrei Bernevig. Hourglass fermions. <i>Nature</i> , 532:189–194, 2016. https://arxiv.org/abs/1602.05585.1, 79, 80
[Wan20]	Juven Wang. Anomaly and cobordism constraints beyond the Standard Model: Topological force. 2020. https://arxiv.org/abs/2006.16996. 25
[WF17]	Haruki Watanabe and Liang Fu. Topological crystalline magnets: Symmetry-protected topological phases of fermions. <i>Phys. Rev. B</i> , 95:081107, Feb 2017, https://arxiv.org/abs/1611.08650, 80
[WG18]	Qing-Rui Wang and Zheng-Cheng Gu. Towards a complete classification of symmetry-protected topological phases for interacting fermions in three dimensions and a general group supercohomology theory. <i>Phys. Rev. X</i> , 8:011055, Mar 2018, https://arxiv.org/abs/1703_10937_41
[WG20]	Qing-Rui Wang and Zheng-Cheng Gu. Construction and classification of symmetry-protected topological phases in interacting fermion systems. <i>Phys. Rev. X</i> , 10:031055, Sep 2020, https://arxiv.org/abs/1811.00536, 41
[Whi42]	George W. Whitehead. On the homotopy groups of spheres and rotation groups. Ann. of Math. (2), 43:634–640, 1942. 11
[WPS14]	Chong Wang, Andrew C. Potter, and T. Senthil. Classification of interacting electronic topological insulators in three dimensions. <i>Science</i> , 343(6171):629–631, 2014. https://arxiv.org/abs/1306.3238, 80
[WS14]	Chong Wang and T. Senthil. Interacting fermionic topological insulators/superconductors in three dimensions. <i>Phys. Rev. B</i> , 89:195124, May 2014, https://arxiv.org/abs/1401.1142, 9, 80
[WW19a]	Zheyan Wan and Juven Wang. Adjoint QCD_4 , deconfined critical phenomena, symmetry-enriched topological quantum field theory, and higher symmetry extension. <i>Phys. Rev. D</i> , 99:065013, Mar 2019. https://arxiv.org/
[WW19b]	Zheyan Wan and Juven Wang. Higher anomalies, higher symmetries, and cobordisms I: classification of higher- symmetry-protected topological states and their boundary fermionic/bosonic anomalies via a generalized cobordism theory. Annals of Mathematical Sciences and Applications, 4:107–311, 2019. https://arxiv.org/abs/1812.11967.
[WW20a]	Zheyan Wan and Juven Wang. Beyond standard models and grand unifications: anomalies, topological terms, and dynamical constraints via cobordisms. J. High Energy Phys., (7):062, 90, 2020. https://arxiv.org/abs/1910.
[WW20b]	Zheyan Wan and Juven Wang. Higher anomalies, higher symmetries, and cobordisms III: QCD matter phases anew. Nuclear Physics B, page 115016, 2020. https://arxiv.org/abs/1912.13514. 18, 19
[WW20c]	Juven Wang and Xiao-Gang Wen. Nonperturbative definition of the standard models. <i>Phys. Rev. Research</i> , 2:023356, Jun 2020. https://arxiv.org/abs/1809.11171. 18
[WWW18]	Juven Wang, Xiao-Gang Wen, and Edward Witten. Symmetric gapped interfaces of SPT and SET states: Systematic constructions. <i>Phys. Rev. X</i> , 8:031048, Aug 2018. https://arxiv.org/abs/1705.06728. 1
[WWW19]	Juven Wang, Xiao-Gang Wen, and Edward Witten. A new SU(2) anomaly. J. Math. Phys., 60(5):052301, 23, 2019. https://arxiv.org/abs/1810.00844. 80
[WWZ19]	Zheyan Wan, Juven Wang, and Yunqin Zheng. Quantum 4d Yang-Mills theory and time-reversal symmetric 5d higher-gauge topological field theory. <i>Phys. Rev. D</i> , 100:085012, Oct 2019. https://arxiv.org/abs/1904.00994. 18
[WWZ20]	Zheyan Wan, Juven Wang, and Yunqin Zheng. Higher anomalies, higher symmetries, and cobordisms ii: Lorentz symmetry extension and enriched bosonic/fermionic quantum gauge theory. Ann. Math. Sci. Appl., 5(2):171–257, 2020. https://arxiv.org/abs/1912.13504.18.33.58.59
[XA18]	Charles Zhaoxi Xiong and A. Alexandradinata. Organizing symmetry-protected topological phases by layering and symmetry reduction: A minimalist perspective. <i>Phys. Rev. B</i> , 97:115153, Mar 2018. https://arxiv.org/
[Xio18]	abs/1709.06998. 4, 78, 79 Charles Zhaoxi Xiong. Minimalist approach to the classification of symmetry protected topological phases. J.
[Yos75]	<i>Phys. A: Math</i> , 51(44), Oct 2018. https://arxiv.org/abs/1701.00004. 19 Zen-ichi Yosimura. Universal coefficient sequences for cohomology theories of CW-spectra. <i>Osaka J. Math.</i> ,
[You20]	12(2):305–323, 1975. 7, 60 Matthew B. Young. Orientation twisted homotopy field theories and twisted unoriented Dijkgraaf–Witten theory.
[YR13]	Comm. Math. Phys., 374(3):1645–1691, 2020. https://arxiv.org/abs/1810.04612. 20 Hong Yao and Shinsei Ryu. Interaction effect on topological classification of superconductors in two dimensions.
[Yu20]	<i>Phys. Rev. B</i> , 88:064007, Aug 2013. https://arxiv.org/abs/1202.5805.80 Matthew Yu. Symmetries and anomalies of (1+1)d theories: 2-groups and symmetry fractionalization. 2020. https://arxiv.org/abs/2010.01136.19

[YX14]	Yi-Zhuang You and Cenke Xu. Symmetry-protected topological states of interacting fermions and bosons. <i>Phys.</i>
(m)	nev. D, 90.243120, Dec 2014. https://arxiv.org/abs/1409.0166. 20, 60
[Zir96]	Martin R. Zirnbauer. Riemannian symmetric superspaces and their origin in random-matrix theory. J. Math.
	Phys., 37(10):4986-5018, 1996. https://arxiv.org/abs/math-ph/9808012. 9
[Zou18]	Liujun Zou. Bulk characterization of topological crystalline insulators: Stability under interactions and relations
	to symmetry enriched U(1) quantum spin liquids. Phys. Rev. B, 97:045130, Jan 2018. https://arxiv.org/abs/
	1711.03090. 80
$[ZWY^+20]$	Jian-Hao Zhang, Qing-Rui Wang, Shuo Yang, Yang Qi, and Zheng-Cheng Gu. Construction and classification of point-group symmetry-protected topological phases in two-dimensional interacting fermionic systems. <i>Phys. Rev.</i>
	B. 101:100501. Mar 2020. https://arxiv.org/abs/1909.05519. 2, 3, 12, 13, 28, 41
[ZXXS20]	Jinyu Zou, Qing Xie, Gang Xu, and Zhida Song. New types of topological superconductors under local magnetic symmetries. <i>National Science Review</i> , 07 2020. https://arxiv.org/abs/1911.03665. 80
[ZYQG20]	Jian-Hao Zhang, Shuo Yang, Yang Qi, and Zheng-Cheng Gu, Real-space construction of crystalline topological
	superconductors and insulators in 2D interacting fermionic systems 2020 https://arxiv.org/abs/2012 15657
	a a to so
	2, 3, 12, 80

DEPARTMENT OF MATHEMATICS, UNIVERSITY OF TEXAS, AUSTIN, TEXAS 78712 Email address: a.debray@math.utexas.edu

88